# X-ray solution scattering (SAXS) combined with crystallography and computation : defining accurate macromolecular structures, conformations and assemblies in solution

Christopher D. Putnam<sup>1</sup><sup>†</sup>, Michal Hammel<sup>2</sup><sup>†</sup>, Greg L. Hura<sup>3\*</sup> and John A. Tainer<sup>2,4\*</sup>

<sup>1</sup> Ludwig Institute for Cancer Research, La Jolla, USA

<sup>2</sup> Life Science Division, Lawrence Berkeley National Laboratory, Berkeley, CA, USA

<sup>3</sup> Physical Biosciences Division, Lawrence Berkeley National Laboratory, Berkeley, CA, USA

<sup>4</sup> Department of Molecular Biology MB4 and The Skaggs Institute for Chemical Biology,

The Scripps Research Institute, La Jolla, CA, USA

Abstract. Crystallography supplies unparalleled detail on structural information critical for mechanistic analyses; however, it is restricted to describing low energy conformations of macromolecules within crystal lattices. Small angle X-ray scattering (SAXS) offers complementary information about macromolecular folding, unfolding, aggregation, extended conformations, flexibly linked domains, shape, conformation, and assembly state in solution, albeit at the lower resolution range of about 50 Å to 10 Å resolution, but without the size limitations inherent in NMR and electron microscopy studies. Together these techniques can allow multi-scale modeling to create complete and accurate images of macromolecules for modeling allosteric mechanisms, supramolecular complexes, and dynamic molecular machines acting in diverse processes ranging from eukaryotic DNA replication, recombination and repair to microbial membrane secretion and assembly systems. This review addresses both theoretical and practical concepts, concerns and considerations for using these techniques in conjunction with computational methods to productively combine solution scattering data with highresolution structures. Detailed aspects of SAXS experimental results are considered with a focus on data interpretation tools suitable to model protein and nucleic acid macromolecular structures, including membrane protein, RNA, DNA, and protein-nucleic acid complexes. The methods discussed provide the basis to examine molecular interactions in solution and to study macromolecular flexibility and conformational changes that have become increasingly relevant for accurate understanding, simulation, and prediction of mechanisms in structural cell biology and nanotechnology.

#### I. Introduction 193

#### 2. Comparison of crystallography and SAXS techniques 195

- 2.1 Interactions of X-rays with matter 196
- 2.2 X-ray crystallography 197
  - 2.2.1 Crystal lattices unit cells symmetry 197

\* Correspondence may be addressed to either author:

J. A. Tainer: Tel.: 1 858 784 8119; Fax: 1 858 784 2277; Email: jat@scripps.edu

G. L. Hura: Tel.: 1 510 486 5378; Fax: 1 510 486 5298; Email: GLHura@lbl.gov

<sup>†</sup> Both authors contributed equally to this paper.

- 2.2.2 Diffraction from crystals and Laue conditions 197
- 2.2.3 Intensities and atomic arrangements 198
- 2.2.4 The Patterson function 199
- 2.2.5 Phase determination 200
- 2.2.6 Structure determination 204
- 2.2.7 Structure refinement 205
- 2.2.8 Flexibility and disorder in crystals 206
- 2.3 SAXS 207
  - 2.3.1 Measuring SAXS data 207
  - 2.3.2 Scattering from macromolecules 210
  - 2.3.3 Pair-distribution function 212
  - 2.3.4 Information content in scattering curves 213
  - 2.3.5 Molecular weight and multimerization state in solution 214
  - 2.3.6 Contrast variation 216
- 2.4 SAXS characterizations that aid crystallography 216
  - 2.4.1 Validating sample quality and assessing crystallographic targets 216
  - 2.4.2 Using low-resolution SAXS envelopes for phasing 218
  - 2.4.3 Structural database of SAXS 218

# 3. Computational techniques for modeling macromolecular flexibility 219

- 3.1 Time scales of macromolecular motion 220
- 3.2 Motion from experimentally determined structures 220
- 3.3 Normal mode analysis 221
- 3.4 MD simulations 221
- 3.5 Monte Carlo simulations 222

## 4. Using crystallography and SAXS to model structures 222

- 4.1 Comparing SAXS profiles and assessing agreement 223
- 4.2 Direct comparison of crystallographic structures with SAXS data 224
  - 4.2.1 Calculation of scattering curves from atomic models 225
  - 4.2.2 Rigid-body modeling with SAXS data 226
  - 4.2.3 Computational docking combined with SAXS data 228
  - 4.2.4 Differences in crystallographic and solution structures 230
  - 4.2.5 Determining biological assemblies in crystal structures 231
- 4.3 Modeling atomic assemblies using *ab-initio* SAXS structures 235
  - 4.3.1 Calculation of *ab-initio* SAXS envelopes 235
  - 4.3.2 Rigid-body docking into low-resolution envelopes 240
  - 4.3.3 Flexible docking into low-resolution envelopes 242
- 4.4 Flexibility and conformational disorder measured by SAXS 243
  - 4.4.1 Multiple well-defined conformations 243
  - 4.4.2 Conformational disorder explored by SAXS 247
  - 4.4.3 Flexibility, oligomerization, or aggregation? 251

# 5. Strategy and tactics for SAXS experiments 253

- 5.1 Data collection 253
- 5.2 Data evaluation 259
- 5.3 Data analysis 261
- 5.4 Structure modeling and interpretation 261
- 5.5 Criteria for evaluation of SAXS results 263
- 5.6 Accuracy and resolution of SAXS experiments, measurements, and models 264

#### 6. Prospects and conclusions 267

- 6.1 General biological and biophysical implications 267
- 6.2 Needs and prospects 268
- 6.3 Expectations and predictions 270

#### 7. Acknowledgments 272

8. References 272

# I. Introduction

Genomic information plays only an indirect role in organizing the spatial and temporal order in cells and organisms. Cellular functions – the decisions to grow and divide, to die by programmed cell death, or to stay static – ultimately lie with macromolecules encoded by DNA. Both proteins and RNA directly control the cell through the reactions they perform, the conformations they adopt, and the interactions that they make in solution. A modern, mechanistic understanding of cells, therefore, requires detailed knowledge of the three-dimensional configuration of the atoms involved in these processes.

Macromolecules are inherently near-sighted. Stable macromolecular interfaces involve forces that typically are only effective in short ranges that can be measured in Ångstroms, and these interfaces typically fit together like pieces of a jigsaw puzzle that exclude bulk solvent and leave very few gaps at the shared surface. Conformational changes, driven by small-molecule binding, allostery, or complex formation can be propagated through long distances. But even these changes are only the sum of short-range interactions between atoms. Ultimately, even the integration of these macromolecules and macromolecular complexes into pathways requires an appropriate milieu in which the macromolecules can act and be acted on. Macromolecules can be thought of as 'cogs in the machine' in which pathways and networks are the result of the availability of substrates.

Cellular coordination, in which macromolecules serve as bit players, functions as a gestalt, where the whole is greater than the sum of its parts. Macromolecules are controlled through their creation and their destruction and through reversible modifications, such as phosphorylation, methylation, and ubiquitination. Functional modification and even control of synthesis and degradation are mechanisms requiring the formation of dynamic interfaces and conformational states that control the macromolecule either directly, through activation or inactivation of the macromolecule of interest, or indirectly through pathways that affect the macromolecule. At its core, each of these levels of control is expressed through the shapes of specific macromolecules. Control of shape is control of information. Thus it seems highly appropriate that the English word 'information' is derived from the Latin *informare*, meaning to 'form', 'shape', or 'organize'.

We believe that the major challenge for structural biology in the next decade will be in providing a mechanistic understanding of the macromolecular and supramolecular complexes and their conformational changes that underlie cell biology and in using these structures to provide new opportunities in the medical, biotechnological, and pharmaceutical fields. Addressing these challenges is fundamentally important from a scientific standpoint, yet tremendously difficult from a practical one. Our experience has suggested that many of the most important problems will involve macromolecular complexes whose structures and complexes will not be easily solved by any single biophysical technique. Thus advances will require the use

and development of methods to bridge atomic resolution structures determined by X-ray crystallography and NMR with lower resolution information about large complexes and conformational states that are too flexible, too large, or too difficult to stabilize as homogeneous samples for these techniques. Recent substantial investments in NMR spectrometers and synchrotron facilities combined with impressive advances in electron microscopy (EM) and cryo-tomography of very large complexes has led to important advances in understanding important macromolecular complexes (Dubochet *et al.* 1988; Lucic *et al.* 2005; Craig *et al.* 2006; Scheres *et al.* 2007). We predict, however, that the combination of X-ray crystallography and small-angle X-ray scattering (SAXS) is well poised to become an important technique for generating structures in solution with a resolution range from roughly 50 Å to 10 Å. Both techniques are becoming increasingly accessible to a broad range of investigators. Sample preparation for SAXS analysis is particularly accessible to a variety of laboratories which otherwise may have thus far never used structural techniques.

As a solution technique, SAXS offers the potential for obtaining some information with every sample, requires modest sample preparation and material relative to crystallography, and is a natural technique for understanding systems possessing substantial flexibility. SAXS can characterize shape and conformation in solution for quite small to very large macromolecular systems, spanning the ranges limiting NMR and EM methods. The combination of current thirdgeneration synchrotron sources and sophisticated computational techniques has substantially increased the utility of SAXS. Experiments can be performed much more rapidly than either EM or crystallographic experiments. In addition, information derived from SAXS data can be useful both prior to and after high-resolution structures are solved. The information content in scattering curves is substantially less than that in crystallography, which is an inherent limitation of this technique. However, SAXS data can be used to determine the low-resolution structures of macromolecules without any additional experimental information. Moreover, SAXS is not only likely to be more powerful in conjunction with atomic resolution structures to provide more accurate and complete models of protein, RNA, and DNA structures, conformations, interactions, and assemblies in solution. The accessible experimental resolution can thus be made appropriate to the biological question being asked. SAXS measurements can directly define the global shape and conformation in solution, whereas the combination of SAXS with computation plus high-resolution component structures provides more detailed three-dimensional information. We therefore expect that SAXS will return any investments made into development of experimental resources or additional computational techniques.

This review aims to provide a general framework for making informed decisions about experimental design, data processing, and data interpretation to combine SAXS with atomicresolution structures from crystallography through computational methods. For the purpose of bringing everyone to the same level, Section 2 provides a comparative assessment of X-ray diffraction and scattering techniques. Section 3 considers computational techniques for modeling macromolecular flexibility, which are important for understanding most of the methods used for fitting and deforming atomic structures in the context of low-resolution information. Section 4 focuses on the principal means to directly compare SAXS data and crystal structures, employ SAXS experiments to derive *ab-initio* SAXS models, and appropriately consider flexibility and disorder in SAXS experiments. Section 5 details specific experimental strategies and tactics and provides a basis to assess the value of different interpretations of SAXS data for a given experiment. Section 6 outlines our views on the prospects for further developments and applications of SAXS to define experimentally validated macromolecular shapes and conformations in



**Fig. 1.** X-ray interactions with sample for SAXS and crystallography. (*a*) Both SAXS and X-ray crystallography involve placing a sample (orange) into a highly collimated X-ray beam (red) and measuring the scattered X-rays. The angle of any scattered position with the direct beam is  $2\theta$ . (*b*) Scattering from a solution of yeast PCNA with a maximum resolution of 23.9 Å. (*c*) Diffraction from a nickel superoxide dismutase crystal at 2.0 Å resolution. The equivalent position of the highest resolution of the SAXS experiment is indicated (red circle). The blue circle indicates the highest resolution achievable ( $q = 0.6 \text{ Å}^{-1}$ ) for SAXS data collection at SIBYLS. Both images collected at beamline 12.3.1 (SIBYLS) at the Lawrence Berkeley National Laboratories. Diffraction image courtesy David Barondeau, Department of Chemistry, Texas A&M University.

solution and provide more complete information than can be typically obtained by either technique alone. At the same time we note concerns and areas where we believe SAXS in particular will benefit from a directed research effort. An overall goal of this review is to provide the framework for improved collaborative efforts involving SAXS with other techniques with the belief that problem-driven developments will help push substantial improvements in SAXS technologies and software with obvious and important relevance to the understanding, simulation, and prediction of macromolecular interactions and conformations in solution.

## 2. Comparison of crystallography and SAXS techniques

SAXS and X-ray crystallography are fundamentally similar techniques and can share most of the hardware required to generate, prepare, and detect X-rays. In most experiments, a collimated, monochromatic beam of X-rays irradiates a sample, and the intensities of the scattered (SAXS) or diffracted (crystallography) X-rays are measured by an X-ray detector (Fig. 1*a*).

A fundamental difference between solution scattering and X-ray crystallography lies in the relative organization of target molecules during data collection. In solution scattering, the signal from all orientations of the target molecules, relative to one another and the experimental apparatus, are averaged together. Solution scattering is continuous and radially symmetric (isotropic) (Fig. 1*b*, *c*). In contrast, in X-ray crystallography the molecules are highly organized within a crystal lattice. Diffraction from a crystal lattice gives rise to discrete diffraction maxima that are caused by the convolution of the crystal lattice onto the continuous transform due to the atomic positions and provides enormously greater signal. Moreover, the lack of radial symmetry in

crystallography retains information about specific orientations in the molecule and requires that crystals be rotated during data collection (Dauter, 1997). Crystallography provides substantially more information content than SAXS, allowing atomic resolution structures to be determined; however, the requirement of packing in the crystal lattice can lead to molecules whose conformations are inappropriately fixed by non-biologically relevant interactions (Section 4.2.4).

The theoretical underpinnings for both of these techniques are well understood and have been the subject of recent reviews (Koch *et al.* 2003) and excellent books (Blundell & Johnson, 1976; Giacovazzo *et al.* 1992; Drenth, 1994). Our goal here is therefore to not to exhaustively address each technique, but to introduce and draw parallels between them. We expect that crystallographers will benefit primarily from the introduction to SAXS and that SAXS specialists will benefit most from the introduction to macromolecular crystallography. We highlight areas of overlap with the expectation that some appreciation of both techniques will be important for using these paired X-ray techniques for the growing number of multi-resolution structuredetermination problems.

# 2.1 Interactions of X-rays with matter

Both SAXS and X-ray crystallography exploit coherent (Thomson) X-ray scattering. In coherent scattering, electrons oscillating under the influence of the electric field of the X-ray beam act as secondary sources, emitting X-rays with the same wavelength as the incident beam, but  $180^{\circ}$  out of phase. The scattering measured at an angle of  $2\theta$  relative to the direct beam is proportional to  $(1 + \cos^2 2\theta)$ , reaching a maximum when the scattering is parallel to the incident X-ray beam  $(2\theta = 0^{\circ})$  and falling off at large  $2\theta$  angles. Atomic scattering factors have been accurately calculated for all of the elements and are influenced by the number of electrons for the atom and the orbitals the electrons occupy. In general, the intensity of coherently scattered X-rays decreases with increasing X-ray energies (decreasing X-ray wavelength). This decrease is discontinuous at energies near atomic orbital-binding energies unique to each atom. The atomic-scattering factors at these energies are described by additional terms accounting for this behavior. Use of this 'anomalous scattering' has become an important method for solving protein crystal structures (Section 2.2.5).

The theoretical limit for the resolution, the minimum distance  $(d_{\min})$  at which two objects can be distinguished, is based on the wave properties of the X-rays:

$$d_{\min} = \lambda/2,$$

where  $\lambda$  is the X-ray wavelength. In practice, the wavelengths typically chosen at synchrotron radiation sources (0.8–1.5 Å) are selected to limit damage to the crystal or to take advantage of anomalous scattering. The theoretical  $d_{\min}$  values of these wavelengths are typically much smaller than can be measured (typically 3 Å to 1 Å) from crystals of macromolecules due to internal disorder or size of the crystal, and are significantly smaller that those that can be meaningfully recorded in SAXS experiments (typically 50 Å to 10 Å, depending on the sample).

Neutrons can also be used in both crystallographic (neutron diffraction) and solution (smallangle neutron scattering; SANS) experiments (Gutberlet *et al.* 2001). Neutrons differ, however, in the fact that they interact with atomic nuclei and thus generate substantially fewer radicals than X-rays during the experiment. Radical formation is known to reduce redox-active sites, such as disulfides and metal centers, and increase sensitivity of biological samples to X-rays (Burmeister, 2000; Ravelli & McSweeney, 2000). Unfortunately, the recent increases in X-ray source intensities has not been mirrored by the fission reactors or spallation sources that currently generate thermal neutrons (Taylor *et al.* 2007). Signal-to-noise problems with neutron diffraction and scattering experiments are significant challenges to obtaining high-quality data sets. Such neutron experiments involve more specialized efforts and have been well discussed in detail elsewhere (Gutberlet *et al.* 2001). Herein we consider X-ray-based experiments, which have broader general utility and applicability to macromolecular systems.

## 2.2 X-ray crystallography

## 2.2.1 Crystal lattices - unit cells - symmetry

X-ray crystallography requires the generation of crystals, and macromolecular crystals are typically grown under conditions where molecules are reversibly driven out of solution (Weber, 1997). Macromolecular samples require that these conditions are gentle and do not cause unfolding or disassociation of complexes. Typically, crystal lattice forces are much weaker than macromolecular folding energies.

Crystals are ordered arrays of atoms related by pure translation (a transformation with only a change in position but not orientation or rotation) in one dimension (fibers), two dimensions (sheets), and three dimensions (lattices). Although fiber diffraction of samples such as DNA can be studied with X-rays, the most common crystals studied by macromolecular crystallography are three-dimensional. The smallest repeating unit of the crystal that is related only by translations is called the unit cell. For three-dimensional crystals, the shape and size of the unit cell is defined by the length of three axes (*a*, *b*, and *c*) and three angles between these axes ( $\alpha$ ,  $\beta$ , and  $\gamma$ ).

Frequently, internal symmetry exists within the unit cell when the unit cell contains multiple molecules. If these symmetries apply to the entire lattice, they are crystallographic symmetries and constrain the parameters of the unit cell. The smallest portion of structural information required to reconstruct the entire lattice through crystallographic symmetries and lattice translations is termed the asymmetric unit. In contrast, symmetries that do not apply to the lattice are non-crystallographic.

In many cases the biologically relevant complex possesses symmetry. These biological symmetries may be observed by some combination of both the crystallographic and noncrystallographic symmetry operators, if the appropriate assembly is present in the crystal structure. For crystallographic analysis, non-crystallographic symmetries can be tremendously useful for map improvement (Section 2.2.6), as well as generation of constraints for the refinement of atomic models, such as have been implemented in the program CNS (Brunger *et al.* 1998). Analogously, particle symmetry provides important constraints on model-based and *ab-initio* reconstruction of three-dimensional shapes from one-dimensional SAXS data (Sections 4.2 and 4.3). Application of these symmetry constraints, like use of the non-crystallographic symmetries in crystallography, substantially increases the accuracy of the final models.

#### 2.2.2 Diffraction from crystals and Laue conditions

X-rays diffracted from crystals can be mathematically treated as if they were being reflected from a plane of angle  $\theta$  to the incident X-rays, and hence the diffraction maxima measured during

the crystallographic experiment are frequently termed 'reflections'. By this definition and due to the geometry of the diffraction event, the diffracted X-rays make an angle of  $2\theta$  with incident beam, and thus  $2\theta$  is the experimentally measured angle between the direct beam position and the diffraction maximum on the X-ray detector (Fig. 1*a*).

As electromagnetic waves, X-rays possess both a wavelength and phase. In the crystal, X-rays are diffracted from multiple parallel planes simultaneously, and this leads to a path difference through which these X-rays travel. When the path difference corresponds to an integral number of wavelengths of the incident X-ray, then the diffracted X-rays undergo constructive (in-phase) interference and can be detected experimentally as diffraction maxima. If not, then the X-rays interfere destructively (out-of-phase) and are not observed. This requirement can be expressed mathematically using the Laue conditions:

$$\mathbf{a} * \mathbf{S} = b$$
,

$$\mathbf{b} * \mathbf{S} = k$$
,

 $\mathbf{c} * \mathbf{S} = l$ ,

where **a**, **b**, and **c** are vectors corresponding to the orientation of the unit cell edges, **S** is the vector corresponding to the path difference between incident and scattered X-rays, and *b*, *k*, and *l* must be integers to ensure constructive interference. The *b*, *k*, and *l* values are the Miller indices and are used to identify each reflection. Thus, the planes that scatter X-rays are determined by the wavelength of the incident X-rays (or wavelengths in the case of Laue multi-wavelength diffraction experiments), the unit cell parameters, and the orientation of the crystal. The Laue conditions give rise to a regularized lattice of diffraction spots (Fig. 1*c*). Importantly, the unit cell size, shape, and orientation, but not the positions of atoms in the unit cell, control which reflections are in the diffraction condition and where these diffracted reflections occur on the detector. This allows crystallographers to collect, index, and process diffraction data prior to knowing the atomic structure.

Bragg's law provides a measure of the distance between the theoretical planes giving rise to X-ray scattering:

$$2d\sin\theta = n\lambda$$
.

By analogy to light scattering through slits, in the case of a theoretical perfect crystal of infinite size the first-order spectrum will occur when n = 1, a second-order spectrum will occur at n = 2, and so forth. The maximum resolution (smallest spacing between planes) measured from the crystal can provide insights into its suitability for data collection and structure determination. In contrast, evaluating samples in SAXS for suitability for structural reconstruction is more difficult because every macromolecular solution will scatter X-rays. In this sense, determining if a SAXS curve is suitable for further analysis (Section 5) is more reminiscent of determining if a crystal is merohedrally or pseudo-merohedrally twined (Yeats, 1997), rather than whether or not it can diffract X-rays.

# 2.2.3 Intensities and atomic arrangements

In contrast to the positions of the reflections, the intensities of the diffracted X-rays are dictated by the atomic arrangements in the unit cell. The positions and types of atoms within the unit cell control both the amplitude and phase. Mathematically,

$$F(b, k, l) = \sum_{j} f_j \exp[2\pi i(bx_j + ky_j + lz_j)],$$

where F(b, k, l) is the structure factor, b, k, l are the Miller indices of the structure factor,  $f_j$  is the resolution-dependent atomic scattering factor, and  $x_j$ ,  $y_j$ , and  $z_j$  are the fractional positions of the *j*th atom in the unit cell. Unfortunately, data collection only allows measurement of the intensities, I(b, k, l), which are the square of the amplitude of F(b, k, l), but not the relative phase information necessary to calculate the electronic distribution in the unit cell. Measurement of the relative phase of X-rays striking the detector at any two diffraction spots has not been possible. This problem is the 'phase problem' of protein crystallography that must be solved in order for structures to be determined.

In general, measured intensities are on a relative scale, not an absolute one. From a theoretical standpoint, I(0, 0, 0) is the square of the sum of the number of electrons in the unit cell and is directly comparable to the important SAXS result where I(0) is proportional to the square of the number of electrons in the scattering particle (Section 2.3.2). Data in crystallography are generally put on only a quasi-absolute scale using the Wilson plot (Wilson, 1942) using data between  $3 \cdot 0$  Å and  $1 \cdot 5$  Å resolution. Placing the data on a true absolute scale is, however, quite difficult. In addition to the ordered atoms, the non-ordered bulk solvent must be included in the calculation, which can be demonstrated by the importance of modeling bulk solvent in refining X-ray structures against low-resolution reflections (Urzhumtsev & Podjarny, 1995). In contrast, the contribution of bulk solvent is explicitly subtracted out during SAXS data processing and typically is involved in the modeling process as a scale factor between calculated and observed intensities. This subtraction is the basis of contrast variation techniques where matching the average electron density of bulk solvent to specific components of a scattering complex causes their contributions to be eliminated from the processed SAXS data (Section 2.3.6).

The crystal lattice has two effects. First, the orientation of the molecules allows the diffraction data to retain information about atomic positions in three-dimensional space, which is lost in SAXS data collected on molecules in solution that are orientationally averaged. Second, the scattering from the atoms in the unit cell is convoluted with the scattering from the lattice so that the crystal diffraction is sampled only at discrete positions defined by unit cell, which also increases signal-to-noise. The measured X-ray intensities in crystallography are the square of the summed amplitudes from the atoms in the unit cell. In contrast, SAXS intensities are the sum of the squared amplitudes from each scattering event. SAXS data are also continuous and vastly over-sampled in comparison to the independent data content as derived from Shannon's theorem. Even with higher signal-to-noise, crystallographic data are substantially undersampled and cannot take advantage of 'super-resolution' techniques that rely upon oversampling to determine the phases of measured reflections (Koch *et al.* 2003). Hence, techniques to solve the phase problem use additional information (Section 2.2.5).

## 2.2.4 The Patterson function

The autocorrelation function for the electron density, which is essentially a three-dimensional map of all of the atom–atom vectors in the crystal, can be written as:

$$P(u, v, w) = \int \rho(x, y, z) \times \rho(x + u, y + v, z + w) \mathrm{d}x \mathrm{d}y \mathrm{d}z,$$

where u, v, w correspond to some difference vector between the position x, y, and z and x+u, y+v, and z+w. Thus N peaks in the electron density map (atoms) will give rise to  $N^2$  peaks in the Patterson function. It has been shown that the above formulation is equivalent to the expression:

$$P(u, v, w) = 1/V \sum_{b} \sum_{k} \sum_{l} |F(b, k, l)|^{2} \cos[2\pi(hu + kv + hw)],$$

where F(b, k, l) is the observed amplitude and V is the volume of the unit cell. The importance of the second expression is that this function can be calculated in the absence of any phasing information, and this function is important for several macromolecular phasing techniques (Section 2.2.5).

When the Patterson function is calculated using the measured amplitudes F(h, k, l), the largest peaks in the autocorrelation are the positions of direct translations between molecules in the crystal lattice; this Patterson function is typically called the 'self-Patterson' (Fig. 2). The self-Patterson function in crystallography is an autocorrelation function and is related to the autocorrelation function, P(r), calculated from the SAXS intensities (Section 2.3.3). Unlike the SAXS P(r) function, the crystallographic Patterson function is calculated from molecules that are restricted in their rotations and thus retains three-dimensional information about the inter-atomic vectors (Fig. 2). Additionally, the Patterson function includes all vectors between atoms for all molecules within the crystal. The P(r) function, on the other hand, is a histogram of distances that are orientationally averaged and correspond only to the scattering particle.

# 2.2.5 Phase determination

To build a map of the electron density in the unit cell by adding together the diffracted X-ray waves, it is necessary to determine phases for each of the reflections whose intensity is measured. The phases of reflections are not constrained mathematically unless they possess special symmetry relationships. The phase for each reflection needs to be determined and one of three techniques is used: experimental methods, direct methods, and molecular replacement.

Experimental phasing techniques systematically perturb the intensities in ways that can be used to extract information about the relative phases of the measured reflections. Isomorphous replacement (IR) depends upon the introduction of 'heavy atoms', e.g. atoms with a large number of electrons such as mercury or uranium, into crystals without perturbing the overall lattice (Ke, 1997). To take advantage of the heavy atoms for phasing information, the positions of these atoms must first be identified either through analysis of Patterson maps calculated from differences in intensity or through direct methods (as described below). For determining heavy atom positions, the differences in the amplitudes between the heavy atom derivative and the native crystal are used. From the  $N^2$  peaks in the *u*, *v*, *w* space of the Patterson function and the symmetry of the crystal, the *N* peaks in *x*, *y*, z space can be calculated. Importantly, as the number of sites where the heavy atom binds increase, so do the number of Patterson peaks. Thus Patterson-based methods for solving heavy atom positions can quickly become challenging to solve. Moreover, as the size of the unit cell increases, the height of each Patterson peak becomes relatively weaker; for particularly large unit cells, heavy atom clusters, such as Ta<sub>6</sub>Br<sub>12</sub><sup>2+</sup>, are used instead of single heavy atoms (Knablein *et al.* 1997).

The other major experimental phasing technique relies upon introducing atoms with anomalous scattering or dispersion into the crystal. Anomalous scattering occurs when X-ray energies



**Fig. 2.** Comparison of the Patterson autocorrelation function in X-ray crystallography and the pairdistribution autocorrelation function in SAXS. A theoretical two-dimensional molecule of four atoms is placed in an arbitrary two-dimensional crystal in solution. The Patterson function contains cross peaks for every interatomic distance in the crystal and these cross-peaks in the *u*,*v*-plane, are indicated by circles and retain directional information about their positions in the crystal. The cross-peaks between symmetry mates are not shown in the expanded view due to the size of the unit cell. The pair-distribution function, on the other hand, resolves distances but not directions within each scattering unit. Thus, all equivalent distances in the four-atom molecule add together.

are near electronic (typically) or nuclear excitations and are typically described as:

$$f = f_0 - f' + if'',$$

where f is the scattering factor of the atom,  $f_0$  is the wavelength-independent component of the scattering, and f' and f'' are the real and imaginary parts of the atomic scattering. At any wavelength, f' is constant, so that f' differences can only be measured by comparing data at different wavelengths. The imaginary part, the photoelectric absorption f'' (Fig. 3a), however, leads to a breakdown in Friedel's law so that the amplitude of the Friedel pairs [F(b, k, l)] and F(-b, -k, -l) are not the same. Phasing experiments that use multiple wavelengths, and can



Fig. 3. The structure of *E. coli* YgbM determined by selenomethionine MAD. (*a*) Comparison of theoretical and measured X-ray fluorescence at the selenium edge for the crystal. (*b*) Anomalous difference Patterson map identifying the selenomethionine Met11-Met11 and Met105-Met105 cross-peaks generated by crystallographic symmetry operators (at the Harker section) calculated using differences between Friedel pairs at the selenium fluorescence inflection point (panel *a*). (*c*) Anomalous difference density contoured at  $5\sigma$  above the mean superimposed with the final refined structure. (*d*)  $2F_{\rm o} - F_{\rm c}$  experimental electron density after MAD phasing and density modification contoured at  $1\sigma$  (green),  $3\sigma$  (lime), and  $5\sigma$  (yellow). (*e*) Final refined structure and  $2F_{\rm o} - F_{\rm c}$  map calculated using final refined phases at  $1\sigma$  (blue),  $3\sigma$  (orchid), and  $5\sigma$  (red). (*f*) Overall structure of YgbM, a zinc-containing TIM-barrel, is shown as a cartoon.

take advantage of both f' and f'' differences (multi-wavelength anomalous dispersion, MAD) or at a single wavelength that can only use f'' differences (single-wavelength anomalous dispersion, SAD) can be performed. Both SAD and MAD experiments use wavelengths at the atomic transitions to maximize the information. Further, experiments using f'' differences must have carefully measured Friedel pairs. As for MIR, the positions of anomalous scatters can also be determined by Patterson-based techniques (Fig. 3b; Hendrickson & Ogata, 1997) and used to determine phases (Fig. 3c, d). Because phase information can be readily combined, it is not unusual for the anomalous signal from heavy atom derivatives (MIRAS) to be used and potentially combined with phasing information from other sources such as MAD or SAD experiments or even partial structures (Blow & Crick, 1959; Sim, 1959). Anomalous dispersion techniques have become a method of choice for solving crystal structures due to the tunability of synchrotron radiation (Helliwell, 1997), the ability to cryo-cool and collect datasets from single crystals (Hope, 1990; Garman & Schneider, 1997), and the techniques to introduce anomalous scatters such as selenomethionine in proteins and bromouridine in DNA molecules (Doublie, 1997). Anomalous scattering has also been used in conjunction with SAXS (ASAXS) (Stuhrmann, 1981; Miake-Lye et al. 1983); however, the orientational averaging of SAXS data eliminates the f'' component, so that anomalous differences can only be measured between wavelengths. In theory, ASAXS has a number of potential applications such as monitoring the distance between two anomalous scatters; however, the signal is small and most applications have involved simple biological systems such as ion solvation of DNA (Andresen *et al.* 2004), and anomalous scattering is not yet as important in SAXS as it is in macromolecular crystal-lography.

Direct methods, on the other hand, have been primarily used in macromolecular crystallography as an alternative to Patterson-based methods for solving heavy-atom or anomalousscattering substructures (Weeks *et al.* 2002). Direct methods take advantages of relationships between phases between multiple reflections (Giacovazzo *et al.* 1992) and require that data are complete, accurate, and are of high enough resolution so that individual scatterers can be resolved (resolutions of better than  $1 \cdot 2$  Å). In macromolecular crystallography, direct methods can be readily applied to substructure determination, as atoms in these substructures typically are much more than  $1 \cdot 2$  Å apart and fit the 'atomaticity' requirements even at moderate resolutions. Direct methods have been able to determine large anomalous substructures (Weeks *et al.* 2002), despite the fact that the use of the differences between intensities rather than intensities introduces some noise into the substructure determination. The determination of these substructures will likely continue to be one of the major roles for this technique in macromolecular crystallography.

Unlike the other techniques described above, molecular replacement attempts to computationally position an atomic model using experimental intensities. From the positioned molecule or molecules, phases can then be calculated for the calculation of electron density maps. The atomic model must be 'similar' to the structure of the crystallized molecule, where the degree of similarity depends on the number of molecules to be found and any potential conformational changes that can occur. Molecular replacement is a six-dimensional search; however, proper rotation of a model will allow a calculated Patterson function containing all interatomic vectors to correlate well with one calculated from the experimental data. This allows the problem to be broken down into a three-dimensional rotational search, followed by a three-dimensional translational search. In the rotation search, typically only vectors within a radius similar to the longest intramolecular distance are considered; however, the close intermolecular vectors between atoms in the atomic packing and non-crystallographically related molecules with other orientations result in 'noise' in this search. Increasing the number of molecules in the asymmetric unit typically makes the molecular replacement problem more difficult. Similarly, the translation function can also be calculated by comparing the Patterson function from the experimental data with the Patterson functions calculated from rotated molecules to which different translations have been applied. Although molecular replacement can be performed by calculating and overlaying explicitly calculated Patterson functions, faster algorithms that do the equivalent searches are used in practice (e.g. Navaza, 2001).

Molecular replacement solutions introduce the possibility of model-biased phases, which generate maps that do not show the differences between the atomic model used to solve the structure and the electron density that gives rise to the scattering. Importantly, model bias tends to increase as homology with the atomic model decreases, but can be detected through the use of omit maps, which are calculated with portions of the model omitted in the phase calculation. For true solutions, the electron density for the omitted regions will still be observed. For phase-biased results that are entirely dependent upon the model, no electron density will be observed in these omitted regions. As the number of solved protein structures increases, the ability to use molecular replacement to rapidly screen through all reasonable or all possible molecular replacement targets will increasingly become a reasonable strategy to solve new structures. To this

end, the ability to identify overall structural similarities through comparison of experimental and calculated SAXS (see Section 2.4.3) could greatly reduce the number of atomic models to be screened and thereby improve the efficiency and success of molecular replacement methods for crystallography.

#### 2.2.6 Structure determination

Given an initial set of phases, either from experimental or computational sources, electron density maps of the unit cell can be calculated. Frequently initial phase information has substantial errors; however, the goal is to generate a map of sufficiently good quality so that an atomic model can be built. Multiple density modification techniques can be used to improve phase information. The two most important ones are solvent flattening or flipping and noncrystallographic symmetry averaging (Vellieux & Read, 1997; Zhang *et al.* 1997). These density modification techniques mainly operate by directly modifying the electron density maps and back calculating new phases. Solvent flattening and solvent flipping operate on the assumption that the bulk solvent regions in crystals should have uniform density and that both positive and negative deviations should either be flattened to this average density or flipped in magnitude. Non-crystallographic symmetry averaging, on the other hand, averages the density between noncrystallographically related molecules. Use of these phase modification techniques not only allows for the improvement of initial phases, but also allows for phase extension in cases where experimental phasing information is at lower resolution than the native dataset (Fig. 3*d*).

From the initial interpretable maps, an atomic structure is usually fit through rounds of atom placement followed by automated refinement (Kleywegt & Jones, 1997). Although experimental maps after phase modification techniques can be of excellent quality, it is possible to place atoms into maps that retain substantial errors in the phases. Thus it is quite common, although not required, to use maps calculated from phases from the updated model itself (Fig. 3e). In the case of atomic partial models or excellent experimental phases, the phases calculated from the model itself can also be combined with experimental phasing information. The crystallographers normally evaluate the experimental electron density map, calculated with the Fourier coefficients  $2F_0 - F_c$ , and the difference electron density map, calculated with the coefficients  $F_{\rm o} - F_{\rm c}$ , where  $F_{\rm o}(b, k, l)$  are observed amplitudes and  $F_{\rm c}(b, k, l)$  are calculated amplitudes. Placement of residues into maps can now be automated, such as by the program suite ARP/wARP (Perrakis et al. 1999) and RESOLVE (Terwilliger, 2003), which can allow for rapid building of macromolecular structures by cycling between an automated density modeling and model refinement programs, given initial phases with sufficient quality. In these cases, the crystallographer supervises the process and corrects regions that are wrong or are trapped by refinement into local minima.

The overall agreement of the model of the asymmetric unit with the experimental data is measured by the 'R-factor':

$$R = \frac{\sum |F_{o}(b, k, l) - F_{c}(b, k, l)|}{\sum F_{o}(b, k, l)},$$

where  $F_{o}(b, k, l)$  are observed amplitudes and  $F_{c}(b, k, l)$  are amplitudes calculated from the model. This number allows the crystallographer to monitor the effects of making manual modifications to the structure as well as providing a numerical target for minimization by automated refinement packages.

One important advance in helping detect the problem of overfitting is the  $R_{\rm free}$  parameter, which calculates an *R*-factor for the current model using a set of reflections, typically several thousand, that are withheld from the refinement calculation (Brunger, 1992). However,  $R_{\rm free}$  is a global parameter and while it can help determine overfitting, it is not sensitive enough to evaluate the validity of small changes to the crystallographic model. Moreover, choosing the number of reflections and which reflections to include in an  $R_{\rm free}$  set can be difficult, particularly when substantial non-crystallographic symmetry exists. Although  $R_{\rm free}$  is imperfect in some ways, it is a universally accepted measure of quality which is useful for both crystallographers and external reviewers. In contrast to X-ray crystallography, an appropriate analog to the *R*-factor is under debate (Section 4.1), and there currently is no suitable SAXS analog to  $R_{\rm free}$  (Section 6).

## 2.2.7 Structure refinement

Crystallographers are aware that molecules in crystals are not completely rigid and are composed of atoms held together by electrons in specific orbitals. However, crystallographers are constrained in their ability to fit these features by the amount of unique data observed in the experiment. Thus, crystallographers are rarely able to fit all of the features of the molecules that are present. Since macromolecular crystals have fairly consistent density ranges (Matthews, 1968), 'rules of thumb' of how the molecules can be modeled based on the data-to-parameter ratio can be given as a function of the highest resolution data measured from the crystal.

Traditionally, a crystallographic model is constructed of primarily one conformation. Additional alternate conformations are typically added at high resolutions (<2.0 Å) when clear evidence for their existence can be observed in difference electron density maps. Atomic positions in the model are most frequently described using three positional parameters, x, y, and zand some number of parameters to describe the displacement of the atom from an equilibrium position. An additional parameter describing the 'occupancy' of a particular atom is normally only refined for structures with alternate conformations or partially bound ligands. For most structures at moderate to high resolutions (3.0 Å to 1.3 Å), a single parameter is used to describe the Gaussian motion of each atom about their equilibrium positions. This isotropic atomic displacement factor (ADF), alternately called the B-factor, the temperature factor, or the Debye-Waller factor, assumes that all atoms can be treated as an isotropic Gaussian distribution of atomic positions centered at an equilibrium position. However, this model fails to capture atomic displacement directed along a single direction (anisotropic displacements) or if the disorder is due to the superposition of multiple static conformations. Thus, in this resolution range, each atom is typically described by four parameters. Introducing geometric restraints, such as bond lengths, bond angles, torsion angles, chiral volumes, and planar restraints, plays an important role in constraining atomic positions to chemically reasonable positions (Engh & Huber, 1991). These constraints are required to maintain a ratio of data and constraints that prevents over-refinement and is applied to both molecular dynamics (MD)-based refinement, such as implemented in CNS (Brunger et al. 1998), as well as generalized least-squares-based refinement, such as implemented in SHELX (Sheldrick & Schneider, 1997).

At lower resolutions (below 3 Å), individual isotropic ADFs for all non-hydrogen atoms can introduce too many fittable parameters, and typically a single isotropic ADF is then refined for groups of atoms (such as side-chains, residues, or even whole domains). In contrast, very high-resolution structures (1·3 Å or better), use anisotropic ADFs that describe probability ellipsoids with six parameters (Willis & Pryor, 1975). And at subatomic resolutions (0·7 Å or better), the

treatment of atoms as spheres of electrons begins to become inappropriate as valence electrons in the protein backbone atoms and unpaired electrons on oxygen atoms become visible in difference electron density maps (Jelsch *et al.* 2000; Ko *et al.* 2003). At these resolutions, additional modeling can be used to fit the experimental data using 'multipolar models' for fitting the non-spherical valence shell electrons, 'dummy atoms' to account for valence bond electrons with additionally Gaussian scatterers, and quantum mechanics modeling methods (reviewed in Petrova & Podjarny, 2004). In each of these cases, the non-spherical treatment of electrons corresponding to the atoms introduces additional parameters that require these extraordinarily high resolutions to be fit.

Importantly, the decision on how to model the ADFs is not dictated by whether or not anisotropic motions (or valence electrons) are present in the crystal, but whether or not any particular model introduces too many parameters. For example, more economical parameterizations of non-isotropic motion have been recently applied, recognizing that much of the anisotropic motion of atoms can be correlated to domain motions within crystals. These schemes simultaneously model the motion of groups of atoms by translation-libration-screw (TLS) models (Schomaker & Trueblood, 1968) or normal mode models (Kidera & Go, 1990) and have been used to explain motion and help refine crystal structures at moderate resolutions (Howlin *et al.* 1993; Winn *et al.* 2001, 2003). The decisions on how to properly model the structure given the information content of the data is as important in the generation of SAXS models as it is in X-ray crystallography. For SAXS, the application of use of external constraints, such as symmetry or atomic structures of individual domains, can be very important to ensure the reproducible reconstruction of solution structures (Section 4), and these constraints are analogous to the use of geometric constraints during crystallographic refinement derived from chemistry or noncrystallographic symmetry.

#### 2.2.8 Flexibility and disorder in crystals

The crystallographic Debye-Waller or B-factor has been used as a surrogate for flexibility and local disorder in crystals. A number of alternative ways to model these features in crystal structures have emerged more recently. These schemes seek to better fit the disorder to improve R-factors as well as to better understand disorder in the crystallized molecules. For example, the use of multiple models provides a possible means to analyze disorder within crystal structures (Furnham et al. 2006). Two types of 'crystallographic ensembles' can be envisioned. In the first, the different structures represent independent refinements against the raw data and do not 'see' each other. These are most equivalent to the independently calculated models generated during NMR refinement. From a number of test cases, it has been suggested that multiple distinct isotropic models can fit the experimental data equally well and thereby suggests that classic measures of model accuracy fail to capture inaccuracies and ambiguities in single model refinements (dePristo et al. 2004). In the second sort of crystallographic ensemble, multiple structures can be simultaneously calculated against the raw data. This refinement would be most appropriate for structures that have the types of disorder that tend to limit the resolution of diffraction. Unfortunately, these ensembles also introduce the real possibility of introducing far more parameters than can be justified by the raw data. Attempts to ensure that individual refinements are restrained have been performed by refining only single models with isotropic ADFs at a time while monitoring R<sub>free</sub> to monitor overfitting (Rejto & Freer, 1996). However, in at least one case, TLS refinement performed better than multiple model refinement (Wilson & Brunger, 2000).

In addition to alternative modeling techniques to fit potential information about disorder in the X-ray diffraction data, theories for the interpretation of diffuse scatter, which is normally ignored in X-ray diffraction experiments, have emerged (Faure *et al.* 1994; Mizuguchi *et al.* 1994). This diffuse scatter arises from transient and static imperfections in the crystal lattice and causes scattered X-ray intensities to be observed at positions other than the Bragg peaks. Since these motions occur in crystals trapped in the lattice, the diffuse scatter is not radially averaged as it is in SAXS. Fitting of this diffuse scatter by techniques like normal mode analysis (NMA; Section 3.3) has suggested that they involve correlated motions of domains of the proteins. Importantly, this diffuse scatter does not include distortions affecting distances between unit cells that give rise to streaks due to lattice distortions. Although these experiments have been largely restricted to model systems (Wall *et al.* 1997a,b; Meinhold & Smith, 2007), they hold potentially valuable information regarding biologically relevant motions.

These methods for understanding flexibility of molecules in the context of a crystal lattice can directly complement more direct measurements of flexibility derived from solution experiments including NMR, SAXS, and fluorescence studies. However, care must be taken as the crystal lattice can directly influence what conformations can be observed and what range of motions are possible.

# 2.3 SAXS

#### 2.3.1 Measuring SAXS data

Unlike X-ray crystallography, SAXS is inherently a contrast method where the scattering signal is derived from the difference in the average electron density,  $\Delta \rho(\mathbf{r})$ , of solute molecules of interest,  $\rho(\mathbf{r})$ , and bulk solvent  $\rho_{\rm S}$  (~0.33 e<sup>-</sup>/Å<sup>3</sup> for pure water):

## $\Delta \rho(\mathbf{r}) = \rho(\mathbf{r}) - \rho s.$

Proteins, for example, have an average electron density of  $\sim 0.44 \text{ e}^{-}/\text{Å}^{3}$ . Larger  $\Delta \rho(\mathbf{r})$  values give rise to larger signals (Table 1), which is important to maximize scattering from dilute solutions as well as for contrast variation techniques (Section 2.3.6). This result also makes SAXS particularly attractive for determining RNA and DNA structures, which have higher contrasts than proteins. In practice, data is collected on a buffer blank and on a sample. Subtraction of observed scattering yields the signal from the scattering due to the macromolecule. Subtracting scattering of the blank from the sample must be done as precisely as possible to accurately measure differences of over three orders of magnitude (Section 5.1).

The scattering curve resulting from the subtraction of the buffer from the sample, I(q), is radially symmetric (isotropic) due to the randomly oriented distribution of particles in solution (Fig. 4). I(q) is a function of the momentum transfer  $q = (4\pi \sin \theta)/\lambda$ , where  $2\theta$  is the scattering angle, as in X-ray crystallography, and  $\lambda$  is the wavelength of the incident X-ray beam. In various treatments, the symbols s and b can be used for q. Confusingly other treatments define  $S = (2 \sin \theta)/\lambda$ , so that  $q = 2\pi S$ , and others define  $\theta$ , rather than  $2\theta$ , as the scattering angle. Each of these definitions is equivalent; the convention being followed must be defined. Here we will consistently use q as defined above with  $2\theta$  as the scattering angle. The units of q are the inverse of units used in the wavelength, typically Å<sup>-1</sup> or nm<sup>-1</sup>, and the value is a measure of the directional momentum change that the photons undergo. By comparison with Bragg's law in X-ray crystallography,  $q = 2\pi/d$ , where 1/d is the reciprocal resolution. Regardless of the

Parameter	Formula	Range of data used and variable definitions	Comments		
Radius of gyration $(R_G)$ : Guinier approximation	$\ln [I(q)] = \ln [I(0)] - \frac{q^2 R_G^2}{3}$	$qR_{\rm G} < 1.3$ globular, $qR_{\rm G} < 0.8$ elongated. I(0): Intensity of the scattering	Most common method of estimating $R_G$ Measured via the slope of the plot $\ln[I(q)]$ vs. $q^2$		
Radius of gyration $(R_G)$ : Debye approximation	$I(q) = \frac{2I(0)}{q^4 R_G^4} (q^2 R_G^2 - 1 + e^{-q^2 R_G^2})$	profile extrapolated to $q=0$ $qR_{\rm G} < 1.4$ for elongated macromolecules	Particularly useful for elongated proteins where the Guinier approximation is valid over narrower range		
Radius of gyration $(R_G)$ : defined by $P(r)$	$R_{\rm G}^2 = \int_0^{D_{\rm max}} r^2 P(r) \mathrm{d}r \Big/ \int_0^{D_{\rm max}} P(r) \mathrm{d}r$	Entire q-range. $D_{max}$ : Maximum dimension of particle	Good consistency check for $R_{\rm G}, D_{\rm max}$ , and $P(r)$		
Pair distribution function $(P(r))$	$P(r) = \frac{r}{2\pi^2} \int_0^\infty I(q)q \sin(qr) dq$	Entire q-range	Indirect Fourier transform methods have been developed for calculating $P(r)$		
Maximum dimension ( $D_{\rm max}$ )	$D_{\text{max}}$ is the value of $r$ at $P(r) = 0$ for large $r$	Requires data $q \leq \pi/D_{\text{max}}$	Assignment of $D_{\text{max}}$ may be complicated by flexibility or multimerization		
Particle volume (V): defined by Porod Invariant	$V = 2\pi^2 I_{\exp}^2(0) \left/ \left( \int_0^\infty I(q) q^2 dq \right) \right.$	Entire q-range. $I_{exp}(0)$ is the experimental intensity at $q=0$ and does not require an absolute scale	The integral portion of this equation is known as the Porod invariant. Accuracy varies for shape and size; however absolute scale and concentration		
I(0): Intensity at $q=0$ which is also proportional to mass and volume	$I(0) = 4\pi \left( \int_0^{D_{\max}} P(r) \mathrm{d}r \right)$	Entire <i>q</i> -range	information are unnecessary Calculation of <i>M</i> and <i>V</i> using this version of <i>I</i> (0) is less susceptible to aggregation and inter-particle correlations than extrapolation of low <i>q</i> data		
Mass (M)	$M = \frac{I(0)\mu^2}{N_A (1 - (\rho_S / \rho_p))^2}$	$\mu$ : Average mass per number of electrons. $\rho_{S}$ : Solvent electron density $\rho_{p}$ : Particle electron density $N_{A}$ : Avagadro's number	<i>I</i> (0) must be on an absolute scale and normalized by mass/volume and not molar concentration		
Formulas for elongated or flexibly linked linear macromolecules					
Radius of gyration of	$\ln [qI(q)] = \ln [qI(0)] - \frac{q^2 R_{\rm XC}^2}{2}$	Intermediate q values	The slope of the linear portion of a plot of $\ln[qI(q)]$		

 Table 1. Common parameters defined by SAXS for monodisperse and homogeneous scatterers

-

Radius of gyration of	$\ln [qI(q)] = \ln [qI(0)] - \frac{T - XC}{2}$	Intermediate q values	The slope of the linear portion of a plot of ln[ <i>qI</i> ( <i>q</i> )]
cross-section $(R_{\rm XC})$	2		vs. $q^2$ is $R_{\rm XC}^2$ ; however, $R_{\rm XC}^2$ goes to 0 as q goes to 0
			in regimes where scattering is dominated by $R_{\rm G}$
Length (L)	$L = (12(R_{\rm G}^2 - R_{\rm XC}^2))^{\frac{1}{2}}$	See $R_{\rm G}$ and $R_{\rm XC}$	The co-axial length rather than the hypotenuse $(D_{\text{max}})$



**Fig. 4.** Experimental SAXS curves and parameters measured for the *Pyrococus furiosis* PF1282 rubredoxin (magenta), the 'designed' scaffoldin protein S4 (red) (Hammel *et al.* 2005), the 'designed' minicellulosome containing three catalytic subunits (green), and the DNA-dependent protein kinase (blue). (*a*)  $D_{\text{max}}$  of the scattering particle is a simple function of molecular weight for perfect spheres (spheres), but not for proteins that adopt different shapes (diamonds). Envelopes correspond to *ab-initio* models calculated from experimental curves using GASBOR. (*b*) The experimental scattering curves for each protein show that the intensity of scattering falls more slowly for rubredoxin ( $R_G$  11 Å; magenta) than the minicellulosome ( $R_G$  82 Å; green). (*c*) The linear region of the Guinier plot, from which  $R_G$  and I(0) can be derived, is function of the  $R_G$ . (*d*) Each protein has both a substantially different  $D_{\text{max}}$  as well as pair-distribution function, reflecting the different atomic arrangements.

incident wavelength, a plot of I(q) vs. q should be identical for the same sample, except at wavelengths where anomalous scattering of atoms within the sample occurs.

Unlike X-ray crystallography, where diffraction provides a clear measure of quality, it can be more difficult to confirm that a measured scattering curve is appropriate for further analysis. In general this is an unsolved problem; however, some empirical guidelines do exist for assessing data quality (Sections 5.2–5.5). Many issues are primarily understood anecdotally, and a directed

effort on the best methods to assess sample quality will benefit from a growing group of researchers adopting SAXS methodologies. We encourage researchers to describe problems as well as solutions in the literature.

#### 2.3.2 Scattering from macromolecules

The theoretical basis for solution scattering has been the subject of an excellent review (Koch *et al.* 2003). Here we briefly consider the most common situation for structure reconstruction (Section 4), in which samples are homogeneous, monodisperse, and lacking long-range interactions in solution. Many of the most commonly used relationships relevant to this case are tabulated in Table 1. More complicated or recalcitrant samples require additional experimental and theoretical treatment (Section 5).

The scattering curve of a homogeneous sample can be derived from the electron distribution of the particle [the pair-distribution function, P(r), Section 2.3.3]:

$$I(q) = 4\pi \int_0^{D_{\text{max}}} P(r) \frac{\sin(qr)}{qr} dr$$

where  $D_{\text{max}}$  is the maximum distance present in the scattering particle.

From a practical standpoint, the lowest resolution portion of the SAXS curve is dictated by a single size parameter (Fig. 4). This size parameter, the radius of gyration ( $R_G$ ), is the square root of the average squared distance of each scatterer from the particle center (Table 1). For example, a sphere of radius r with uniform electron density, for example, has a  $R_G = (3/5)^{\frac{1}{2}}r$ .  $R_G$ , like the hydrodynamic or Stokes' radius ( $R_S$ ), is shape-dependent and a poor measure of the actual molecular weight (volume) of the molecule of interest.  $R_G$  and  $R_S$  are different, however, in that  $R_S$  is the radius of an equivalent sphere that diffuses identically to the molecule of interest, hence  $R_S = r$  for a perfect sphere.

At low resolution, the scattering can be described by the Guinier approximation:

$$I(q) = I(0) \exp[-(q^2 R_G^2)/3].$$

The Guinier plot of  $\log(I(q))$  against  $q^2$  will give a straight line from which  $R_G$  and I(0) can be extracted (Fig. 4c; Guinier & Fournet, 1955). The q-range over which the Guinier approximation is valid  $(qR_G < 1.3 \text{ for globular proteins})$  is much larger for particles with small  $R_G$  than larger particles (Fig. 4c). In practice, this estimation of  $R_G$  must be performed iteratively or interactively (Konarev *et al.* 2003), since new estimates of  $R_G$  can alter the q-range for which the estimate can be made. Lack of linearity in the Guinier plot is a sign that more care needs to be taken to evaluate the sample (Section 5), or that samples are elongated. For these samples, other methods for estimating  $R_G$  may be more appropriate (Table 1). Similarly,  $R_G$  should not vary with concentration for well-behaved samples with no interparticle interference or aggregation.  $R_G$  shows some dependence on the contrast difference between bulk solvent and the sample comparisons of different samples should be performed in the same buffer.

The second important parameter that can be evaluated from the lowest q values is I(0), the intensity measured at zero angle (q=0), which must be determined by extrapolation, as it is coincident with the direct beam. On an absolute scale, I(0) is the square of the number of electrons in the scatterer and is unaffected by particle shape and is useful for molecular weight determination (Section 2.3.4). I(0) is equivalent to the value of I(0, 0, 0) in X-ray crystallography

(Section 2.2.3). For well-behaved samples, a plot of I(0) vs. concentration gives a straight line. Additionally, since I(0) depends on the square of the number of electrons (molecular weight), SAXS is particularly sensitive to the assembly state of the scatterers.

Higher q values contain details regarding molecular shape. For folded macromolecules, the intensity of the scattering falls off by Porod's law (Porod, 1951):

 $I(q) \propto q^{-4}.$ 

This relationship, however, assumes a uniform density for the scatterer, which breaks down at high q values when atomic resolution information begins to contribute significantly. Hence, Porod's law, like the Guinier approximation, holds only in a portion of the scattering curve, and we have observed some samples that possess little or no scattering following Porod's law. For arbitrary polymers, this region of scattering is typically termed the 'power law regime', where the resolution-dependence of the scattering can be expressed as:

 $I(q) \propto q^{-d_{\rm f}},$ 

where  $d_f$  is the fractal degrees of freedom. For example, scattering comprised of spheres has a  $d_f=4$ , flat (oblate) ellipsoids has a  $d_f=2$  in the high *q*-range, whereas scattering from needle-like (prolate) ellipsoids has a  $d_f=1$  in the high *q*-range. Random coils in 'good solvent' have  $d_f=5/3$ .

Thus, SAXS is an ideal method for identifying and characterizing polymers without folded domains. The Kratky plot  $[q^2I(q)]$  as a function of q, which can be calculated directly from the scattering curve, provides an excellent tool for evaluating the folding of samples. For folded domains, the Kratky plot yields a peak roughly shaped like a parabola. The position of the peak provides some information about its overall size; however, our experience has shown that the position is shape-dependent like  $R_G$  and thus cannot directly provide information regarding molecular weight. In contrast, extended semi-stiff polymers, such as random coil peptides, follow the Porod–Kratky worm-like chain model (Kratky & Porod, 1949). Random coil or unstructured peptides lack the characteristic folded peak and are linear with respect to q in the large q-region. At low resolutions the scattering can be described by (Brulet *et al.* 1996):

$$\frac{I(q)}{I(0)} = \left[\frac{2}{y^2}(y - 1 + e^{-y}) + \frac{b}{L}\left(\frac{4}{15} + \frac{7}{15y} - \left(\frac{11}{15} + \frac{7}{15y}\right)e^{-y}\right)\right]e^{\frac{-q^2R_c^2}{6}},$$

where  $y = q^2 Lb/6$  and  $R_c$  is the radius of gyration of the cross section, L is the total length of the polymer, and b is twice the persistence length, the maximum length that the polymer chain persists in any one direction (Table 1). This relationship holds in the resolution range q < 3/b. For peptides, b varies between 19 Å and 25 Å, yielding an average persistence length of 9·5–12·5 Å or roughly 3–4 amino acids (Perez *et al.* 2001). The expected  $R_G$  for the unfolded polypeptide can be calculated with the equation:

$$(R_{\rm G})^2 = b^2 \left[ \frac{x}{6} - \frac{1}{4} - \frac{1}{4x} + \frac{1}{8x^2} \right],$$

where x = L/b. This equation is useful as the  $R_G$  value for unfolded or chemically denatured samples is so large that the scattering region following Guinier's approximation is typically not recorded in normal beamline geometries (Calmettes *et al.* 1994).

#### 2.3.3 Pair-distribution function

The pair-distribution function P(r), also called the pair-density distribution function (PDDF; Fig. 4*d*) is the SAXS function corresponding to the Patterson function (Section 2.2.4). This autocorrelation function can be directly calculated through a Fourier transform of the scattering curve (Table 1), and the result provides direct information about the distances between electrons in the scattering particles in the sample, in a manner similar to the Patterson function. The P(r)function can also be calculated directly from the electron density:

$$P(r) = r^2 \left\langle \int_V \Delta \rho(r) \Delta \rho(u+r) \mathrm{d}r \right\rangle_{\Omega}.$$

The important differences between P(r) and the Patterson function are that the P(r) is radially averaged and lacks vectors corresponding to vectors between scattering particles, which gives rise to large 'origin peaks' at (0, 0, 0) and other positions corresponding to pure crystallographic translations in the Patterson function. Typically, the P(r) function is calculated by an indirect Fourier transformation to avoid problems due to discrete sampling of the I(q) curve over a finite range (Glatter, 1977). The indirect Fourier transform essentially constructs trial P(r) functions that are Fourier transformed and evaluated in comparison with the experimental scattering. In the GNOM program (Semenyuk & Svergun, 1991), a regularizing multiplier is used to balance the smoothness of the trial P(r) functions with the goodness of fit to the data. In the GIFT program (Bergmann *et al.* 2000), the inverse transformation is solved using Boltzmann simplex simulated annealing to solve the nonlinear dependencies of the scattering curve with the P(r)structure factor parameters and iteratively fits the parameters. Additionally, GIFT also simultaneously fits contributions from the scattering due to interparticle interactions (Brunner-Popela & Glatter, 1997).

Theoretically, the P(r) function is zero at r=0 and at  $r \ge D_{\text{max}}$ , where  $D_{\text{max}}$  corresponds to the maximum linear dimension in the scattering particle. For the processing of real data, the P(r)function is typically constrained in the calculation to be zero at these values. This constraint is often not necessary for well-behaved (globular) samples and can be an indicator of good quality data (Section 5.3). On the other hand unfolded proteins are often not zero at r=0 in unconstrained P(r) functions, and non-zero values at  $r=D_{max}$  may indicate aggregation or improper background subtraction. D<sub>max</sub> is useful for characterizing the sample; however, accurately determining  $D_{\text{max}}$  for samples can be difficult. The scattering data should be measured at  $q \ge 2\pi/2$  $D_{\text{max}}$ . More problematically, the indirect Fourier transformation methods to calculate P(r) rely upon the value of  $D_{max}$ , giving the value more importance than if the P(r) function could be calculated from a direct Fourier transformation. Moreover, the P(r) curves are typically small in the vicinity of  $D_{\text{max}}$ , and hence contribute little to the overall scattering. Thus, errors in estimates of  $D_{\text{max}}$  can be difficult to identify, including extended structures and globular structures with disordered extensions, such as unstructured N- and C-termini in proteins. In practice, estimation of  $D_{\rm max}$  by the inverse Fourier transformation involves choosing multiple  $D_{\rm max}$  values and evaluation of the resulting P(r) functions for their fit to the experimental scattering.

The P(r) function has many important usages. First, a value for  $R_G$  and I(0) can be calculated from the P(r) function that takes into account all of the collected data and is not limited to the small region about the direct beam that is used in the Guinier approximation. Thus, this real space approximation is likely to be a better estimate for samples complicated by small amounts of aggregation that most strongly affect the lowest resolution information. Second, P(r) functions



**Fig. 5.** Theoretical and experimental SAXS from a 76mer double-stranded DNA fragment. (*a*) Theoretical scattering from a 76mer B-DNA fragment without counter ions was calculated using CRYSOL (green; Svergun *et al.* 1995) and compared to the experimental scattering (black) or *ab-initio* reconstructions generated by GASBOR (red; Svergun *et al.* 2001a). The experimental data has a  $R_G$  of 69 Å as compared to the 67 Å for the theoretical B-DNA structure. (*b*) Comparison of the P(r) functions calculated from the theoretical and experimental scattering by GNOM. The early peak and roughly linear fall off of the P(r) are characteristic of a linear or extended molecule and can be seen with protein samples as well (Fig. 10). The observed  $D_{max}$  is 260 Å, whereas the  $D_{max}$  calculated for the theoretical DNA fragment is 255 Å, and the linear length expected for 76 bases of B-DNA is 250 Å. C. *Ab-initio* reconstructions of the DNA generated by GASBOR are quite similar in thickness and length to the size of B-DNA.

can be readily calculated from atomic models (Section 4.2.1). This has important implications for many different methods of using atomic models in conjunction with SAXS data (Section 4.2). The P(r) function can also give some initial indication of the overall shape from its overall shape, for example spherical objects with bell-shaped P(r) functions can be readily distinguished from rod-like shapes (Fig. 5) and are particularly useful in conjunction with Kratky plots (Section 2.3.2).

# 2.3.4 Information content in scattering curves

One of the central strengths of SAXS is that measurements are done in solution with little preparation relative to other techniques. The downside is that the measured data are orientationally averaged and cannot, for example, be used to distinguish between enantiomorphs. Furthermore, the scattering curve only has a small number of independent data points, typically estimated by the number of Shannon channels (Shannon & Moore, 1949):

$$N_{\rm s} = (q_{\rm max} - q_{\rm min}) D_{\rm max} / \pi$$

The number of independent values that can be extracted from the scattering (reciprocal space) has been shown to be equivalent to the number of independent data-points in real space (Moore, 1980). For most SAXS curves,  $N_s$  usually does not exceed 10–15. As the lowest-resolution value describes the overall size of the scatterer ( $R_G$ ), the first data point  $q_{\min}$  ought to be measured at  $q_{\min} \leq \pi/D_{\max}$ . An alternative measure for the number of experimentally determined parameters has been suggested that uses a maximum entropy method (Mueller *et al.* 1996). This method accounts for the problem of determining a uniquely defined  $q_{\max}$  in the presence of experimental noise; however, this measure does not substantially change the number of parameters available (Vestergaard & Hansen, 2006). SAXS data are dramatically over-sampled, in that  $\Delta q$  between two adjacent points measured in the scattering curve are, however, much less than  $\pi/D_{\max}$ . This

fact has been used to argue that the effective information content is higher than predicted from the number of Shannon channels (Koch *et al.* 2003).

Given this estimate of information content in a solution scattering curve it is remarkable that accurate shapes can be derived. This intuitively daunting feature may explain why the development of SAXS has not been pursued at the same rate as crystallography. Studies have demonstrated situations in which more and more detailed structural information can be extracted utilizing SAXS data when additional constraints are imposed on the reconstructions. Modern *ab-initio* algorithms include constraints that attempt to force final solutions to have protein-like properties. For example GASBOR enforces penalties on its shape reconstructions for compactness (Svergun *et al.* 2001b). The number of restraints added by this type of external information is not easily estimated. Regardless of the precise number of fitted parameters that can be justified, we would suggest that there is a clear analogy with the difficulties of refining X-ray crystal structures at different resolutions (Section 2.2.7) in which the best use of the available experimental data includes external information. Thus, using known crystal structures as a basis for fitting low-resolution SAXS data mirrors the use of chemical bonding parameters in the case of moderate or 'low'-resolution X-ray diffraction data, and we detail theoretical and practical methods to do this in Sections 4 and 5.

#### 2.3.5 Molecular weight and multimerization state in solution

In a monodisperse, ideal solution of identical particles, the observed scattering is linearly related to the number of particles, N, in the sample. The measured I(0) obtained after scaling for concentration corresponds to the scattering of the single particle and it is proportional to the square of the total excess scattering length in the particle. If the measurements are made on an absolute scale (cm<sup>-1</sup>), I(0) can be directly related to the molecular weight of the particle:

$$I(0) = Nm^2 (1 - \rho_0 \psi)^2,$$

where *m* is the number of electrons of the particle,  $\rho_0$  is the average electron density of the solvent, and  $\psi$  is the ratio of the volume of the particle to its number of the electron. If the scattering curve is scaled by concentration in units of the molarity of the particle, then I(0) is proportional to the mass squared. However typically only the molarity of the monomeric unit is known and the concentration of the target particle, *c*, is reported as mass per volume (mg/ml) and is  $c = N\mu m/N_A$ , where  $N_A$  is Avogadro's number, and  $\mu$  is the ratio M/m of the molecular weight to the number of electrons, which depends on the chemical composition of the particle (for proteins a good approximation is M/m = 1.87). Therefore,

$$I(0)/c = \frac{N_A M}{\mu^2} (1 - \rho_0 \psi)^2.$$

If  $\psi$ ,  $\rho_0$ , and *c* are known, and the intensity of the incident beam is known on an absolute scale, then the intensity at the origin provides a determination of the molecular weight (Vachette & Svergun, 2000; Koch *et al.* 2003).

An experimentally more tractable measurement of mass can be obtained using relative *I*(0) values after proper calibration with reference samples, such as lysozyme (14.3 kDa), bovine serum albumin (BSA, 66.2 kDa), and glucose isomerase (172 kDa) (Kozak, 2005; Mylonas & Svergun, 2007). Samples composed of multiple components with different average electron

densities, such as protein–DNA complexes, can be more problematic. Nevertheless even with mixed electron density systems, bounding the mass between values may be sufficient to establish the multimeric state as has been demonstrated for membrane protein systems collected above the critical micelle concentration of the solubilizing detergent (Columbus *et al.* 2006).

Each of these techniques requires accurate determinations of I(0) values from Guinier or Debye approximations (Table 1), or from estimates using the P(r) function:

$$I(0) = 4\pi \int_0^{D_{\max}} P(r) \mathrm{d}r.$$

The P(r) is calculated from the entire scattering curve. Thus, extracting I(0) from P(r) has several advantages over the I(0) measured from Guinier plots, particularly for data where only a few points have been measured in the Guinier region or where the Guinier region is affected by interparticle interactions. The P(r)-based I(0) value is typically reported by programs performing P(r) calculations (Svergun, 1992; Bergmann *et al.* 2000).

Since most macromolecules have fairly uniform densities, the molecular weight is also directly related to volume. Volume information can be derived from SAXS curves in experiments where neither absolute I(0) nor reference samples have been measured. In this approach the theoretical excluded volume calculated from sequence or from an atomic model, such as reported by the program CRYSOL (Svergun *et al.* 1995), can be compared to volumes generated by *ab-initio* shape-determination algorithms (Section 4.3) (Hammel *et al.* 2002; Krebs *et al.* 2004) and/or volumes derived from the scattering according the Porod law (Porod, 1982).

The volume of the macromolecule undergoing scattering can be calculated from I(0) and the Porod invariant Q (Porod, 1982):

$$V = 2\pi^2 \frac{I(0)}{Q},$$

where the invariant is calculated by:

$$Q = \int_0^\infty q^2 I(q) \mathrm{d}q.$$

This calculation does not require data normalization. For globular proteins, Porod volumes in nm<sup>3</sup> are typically twice the molecular masses in kDa and is a valuable conformation of mass estimates using I(0) (Petoukhov *et al.* 2003; Gherardi *et al.* 2006). These volume determinations are, however, subject to error as they rely on the accurate data over the entire *q*-range (due to extrapolation of high *q* using the fall off of intensity with  $q^{-4}$ ). The contribution of internal particle structure to scattering at larger angles becomes significant at *q* values above  $0.2 \text{ Å}^{-1}$ , and this contributes error to the calculation. Thus, the large angle portions of the curves should be discarded in the computation (Glatter, 1982). Additionally, this technique for extracting mass is very inaccurate for asymmetric particles. Excluded volumes can be easily calculated using programs such as PRIMUS (Konarev *et al.* 2003).

Importantly, the *I*(0) method for molecular weight determination and volumes derived from *ab-initio* shape determination (Hammel *et al.* 2002; Krebs *et al.* 2004) should yield consistent results (Petoukhov *et al.* 2003, 2006; Gherardi *et al.* 2006; Nemeth-Pongracz *et al.* 2007; Qazi *et al.* 2007), and thus each can be used to independently confirm the results from a single sample (Table 1). In practice, SAXS provides a powerful approach to determining the molecular weight

and assembly state in solution that can be extremely useful for crystallization efforts (Section 2.4), for modeling solution assemblies (Section 4), and for interpreting biochemical and mutational results.

#### 2.3.6 Contrast variation

Measurable scattering from a solute is contingent on the contrast in scattering density between the solute and solvent. For length scales larger then 15 Å the scattering density for biomolecules is approximately homogeneous. Thus, most internal structural features can be successfully ignored at resolutions lower than q=0.2 Å<sup>-1</sup>. SAXS can be used to extract internal structural features of systems comprised of two or more components with distinct average electron densities. Since SAXS is a contrast method, variation of the average solvent electron density can extract information about the inner structure of multicomponent systems (Stuhrmann, 1973, 1982). Choosing appropriate solvent electron densities with high concentrations of sugars, glycerol, or salt can mask out the scattering of one of the components (Pilz, 1982), and contrast variation studies have been successfully performed with SAXS (Muller *et al.* 1978). In practice, however, the dramatic difference in the interaction of neutrons with hydrogen atoms (<sup>1</sup>H) and deuterons (<sup>2</sup>H), makes neutron scattering (SANS) with specifically deuterated components, and not SAXS, the technique of choice for contrast variation studies.

SANS contrast variation studies on two component systems have been performed for samples such as the ribosome (Stuhrmann *et al.* 1976; Svergun *et al.* 1997a, b), human plasma lipoprotein (Stuhrmann *et al.* 1975), and DNA–protein complexes (Chamberlain *et al.* 2001). One recent SANS study used contrast variation to eliminate the contribution of the detergent molecules required to solubilize the hydrophobic membrane-binding protein apolipoprotein B-100 (apoB-100) (Johs *et al.* 2006). ApoB-100 is the protein component of human low-density lipoproteins (LDL), which triggers the receptor-mediated cellular uptake of LDL. Due to its size (550 kDa and 4536 residues) and hydrophobicity, the apoB-100 structure had not been well characterized. The authors have calculated low-resolution models (Fig. 6*a*, *b*) that reveal a pronounced cavity in the center of the molecule with alternating wide and narrow sections, which have been interpreted as folded domains connected by linkers. These linkers may confer flexibility, allowing apoB-100 to rearrange when forming the lipoprotein particle (Fig. 6*b*–*d*).

#### 2.4 SAXS characterizations that aid crystallography

# 2.4.1 Validating sample quality and assessing crystallographic targets

A frequently underappreciated feature of a SAXS experiment is that even in the absence of an atomic resolution structure, SAXS experiments are useful in directly assessing sample quality. In a structural genomics project involving protein complexes isolated from *Pyrococcus furiosis*, features of the SAXS were excellent indicators of sample folding, assembly, and aggregation. In turn each of these features was a useful predictor for identifying samples that were readily crystallized (G. L. Hura, J. A. Tainer, M. W. Adams unpublished observations).

Experimental evidence has shown that homogenous solutions of macromolecules with little or no aggregation have a tendency to readily crystallize both when multiple proteins (D'Arcy, 1994; Ferre-D'Amare & Burley, 1997) or multiple preparations of the same protein are compared (Habel *et al.* 2001). Dynamic light scattering (DLS) has emerged as an important technique to characterize macromolecules (Zulauf & D'Arcy, 1992) and screen for buffer conditions for



**Fig. 6.** Reconstructed three-dimensional *ab-initio* model of the detergent solubilized apoB-100 protein. (*a*) Ten independent single models with similar goodness of fit were restored by DAMMIN (Svergun, 1999). (*b*) The average envelope of apoB-100 was calculated for 10 independent DAMMIN models using the program DAMAVER (Volkov & Svergun, 2003), and the secondary structure information was mapped onto the low-resolutionmodel. (*c*) Hypothetical model of the spatial arrangement of apo-100B in LDL. (*d*) Model of the LDL particle with apoB-100 and a superimposed 250 Å sphere, representing the lipid components (after Johs *et al.* 2006).

crystallization (Jancarik *et al.* 2004) as it can evaluate the overall aggregation of the sample. As SAXS uses wavelengths that probe atomic resolution information, the presence of structure in the SAXS curves as compared to featureless decay has been used to distinguish between samples where aggregation can be ameliorated by varying conditions, centrifugation, or size-exclusion chromatography from those that are hopelessly aggregated.

Similarly, for some molecules that form complexes, control of the molecular assembly state has been important for crystallization, such as for the requirement of multimerization by the *Schizosaccharomyces pombe* cell cycle-regulatory protein suc1 for crystallization (Parge *et al.* 1993;

Bourne *et al.* 1995), although many others examples also exist based on anecdotal evidence. As has been described in Section 2.3.4, SAXS is exquisitely sensitive to the assembly state and thus can be used to identify buffer conditions that help stabilize particular assemblies for crystallization.

Finally, SAXS is also sensitive to the overall shape of the macromolecule, and samples that are unfolded are clearly visible in the Kratky plot (Section 2.3.2). This evaluation is particular important for engineering and characterizing truncation mutations for crystallographic studies, such as the  $\alpha$ -subunit of DNA polymerase III (Lamers *et al.* 2006), as well as identification of natively unfolded proteins (Shell *et al.* 2007).

Taken together, these uses of SAXS data can be tremendously helpful for characterizing proteins being expressed and purified for biochemical, biophysical and crystallographic studies. In general, the typical concentrations and total amounts of sample required for SAXS are readily accessible when crystallography or other biophysical experiments are being pursued. In addition to evaluating constructs and buffer conditions for aggregation or unfolding (Section 5), SAXS can be used to provide initial information regarding the shape and the assembly of complexes from *ab-initio* structure reconstruction as discussed below (Section 4.2) and direct information for establishing the biologically relevant solution assemblies as discussed below (Section 4.2.5).

# 2.4.2 Using low-resolution SAXS envelopes for phasing

From a one-dimensional experimental SAXS curve, it is possible to reconstruct a threedimensional envelope (Section 4.3.1). Even in the absence of a high-resolution structure, these envelopes may be useful for low-resolution phasing of crystallographic data. For example, specialized programs such as FSEARCH can place envelopes derived from EM or SAXS into the crystal by molecular replacement (Hao *et al.* 1999; Ockwell *et al.* 2000; Hao, 2001, 2006). This has been successful for low-resolution phasing of phytase (Liu *et al.* 2003) and lobster clottable protein (Kollman & Quispe, 2005). Many standard molecular replacement programs, such as AMoRe (Navaza, 2001), can also do molecular replacements using structure factors calculated from a low-resolution envelope in an appropriate cell with P1 symmetry (Urzhumtsev & Podjarny, 1995). *Ab-initio* phasing from fairly simple envelope models has been used for phasing viral particles. For these systems, success is due to the ability to perform rounds of phase extension with non-crystallographic symmetry averaging (reviewed in Rossmann, 1995).

For average macromolecular crystals lacking the extensive non-crystallographic symmetry of viruses, the major challenge for the use of low-resolution phasing techniques is to extend the phase information to a resolution high enough that atomic models can be built. One intriguing possibility that may be more accessible for general crystallographic problems might be to leverage the power of multicrystal averaging, if molecular replacement solutions can be found for each non-isomorphous crystal (Cowtan & Main, 1996, 1998). However, even in the absence of appropriate density modification techniques for phase extension, low-resolution phasing by envelopes has the potential to help locate heavy atoms, particularly heavy-atom clusters used for phasing crystals with large asymmetric units, and thereby allow the first heavy-atom positions to be identified from difference Fourier maps rather than the more difficult Patterson maps.

## 2.4.3 Structural database of SAXS

With the increase in speed of modern computers, one potential crystallographic phasing strategy is a molecular replacement search using all solved domains from the Protein Data Bank (PDB)



**Fig. 7.** An example of a successful DARA (Sokolova *et al.* 2003a, b) hit from a protein of unknown structure and function. Using SAXS data, an *ab-initio* GASBOR envelope was calculated that superimposes well with the crystal structure of the best DARA hit.

(Bernstein *et al.* 1977). These searches still take substantial amounts of computer time and one useful strategy would be to focus the molecular replacement search on models that are most like the protein of question. Measurement of a solution scattering curve might be able to provide sufficient structural information to focus such a search.

Recently, a database of calculated scattering curves, DARA, was created for a large portion of the structures deposited in the PDB (Sokolova *et al.* 2003a, b). DARA ranks experimental scattering curves relative to their fit to the precalculated DARA scattering curves. Theoretical scattering curves that were scored as similar tended to fit both overall shape and secondary structural features. For a brute-force molecular replacement search, these hits are likely to be much more useful for phasing.

There are a number of difficulties in performing these comparisons, even in the absence of experimental noise. First, determining appropriate criteria by which to compare scattering curves can be difficult (Section 4.1). Second, many PDB entries lack appropriate or correct information for the biologically relevant assemblies in the file meta-data, which are problematic for DARA searches. Moreover, multiple conformations of residues and heavy atoms in the crystal structures may cause problems for calculating the theoretical scattering.

These difficulties may be responsible for the mixed success in studies using the current version of DARA (Hamada *et al.* 2007). Nevertheless we have found it useful in several cases (Fig. 7), and it is currently unclear if success is dependent upon specific features of the macromolecules being investigated. However, the possibility of identifying similarly shaped molecules using SAXS curves has the potential to be particularly powerful for high-throughput applications. Fine tuning parameters for analysis of SAXS data may also be aided by applying the analysis on structures identified by DARA. For example, the database search may help identify hollow protein shells like ferritin (Trikha *et al.* 1995), which tend to be biased against in *ab-initio* shape restoration (Section 4.3).

## 3. Computational techniques for modeling macromolecular flexibility

A fundamental problem in combining solution SAXS with atomic resolution structures determined by crystallography is to find ways to model motions that are accessible to molecules outside of a crystal lattice (Section 3.1). Motions can be derived when multiple conformations are experimentally observed at high resolution (Section 3.2) or can be inferred through computational techniques such as normal modes analysis (Section 3.3), molecular dynamics (Section 3.4), and Monte Carlo-based techniques (Section 3.5). The current challenge is to identify and use the appropriate computational tools for modeling the results in any particular experiment. We expect that leveraging SAXS data using computation will not only substantially improve SAXS interpretation tools, but will also provide experimental feedback to improve the accuracy of modeling techniques.

# 3.1 Time scales of macromolecular motion

Macromolecular motions have been best studied in proteins, and the time scale at which the motion occurs is strongly dependent upon the nature of the conformational change. Fast motions on the femtosecond to picosecond time scales are primarily vibrational. Switching between side-chain rotamers tends to occur in the picosecond time range, and is best understood in tryptophan residues using their intrinsic fluorescence. Quasi-harmonic motions, such as twisting of  $\beta$ -sheets and the flexing of  $\alpha$ -helices in proteins occur in the sub-nanosecond time scale.

Slower conformational changes typically involve concerted flexing or movement of domains and are more difficult to study, as these motions are typically slower than the rotation of the molecule in solution (nanosecond time scale). Thus, the information regarding orientation of specific reporters that are imparted during the probing steps of both NMR and fluorescence experiments are lost faster than the conformational change occurs. Moreover, these motions also tend to be faster than the tens of milliseconds it takes to collect an NMR signal, and thus are not resolved as separate resonances. Thus, slower large-scale motions are problematic for both fluorescence and classical NMR experiments. At present, the best understood large-scale conformational changes are results of solving independent structures in which each structure is stabilized in a particular state. Stabilization can occur through binding ligands, mutagenizing individual residues, or, in crystallography, finding new crystal forms with different packing environments. Recent advances in measuring and interpreting residual internuclear dipolar couplings from weak alignment NMR, however, has the potential to provide a wealth of dynamic information for these kinds of motions in the biologically important time scale of  $10^{-8}$  to  $10^{-4}$  s (Bax & Grishaev, 2005).

# 3.2 Motion from experimentally determined structures

One method of understanding macromolecular flexibility is to extract information from experimentally determined crystal structures in different conformations. For systems where many structures are available, it is possible to apply the essential dynamics (ED) method (Amadei *et al.* 1993) to analyze the major motions that underlie conformational changes. ED was originally derived to extract information about motions from MD simulations of proteins, but has been applied to situations for which multiple (over 15) individual structures are known (van Aalten *et al.* 1997). In contrast to ED, other techniques such as those implemented in the MolMovDB server (Echols *et al.* 2003) attempt to extrapolate intermediates between two structurally determined endpoints by generating theoretical intermediates that are energy minimized. In some sense, the introduction of these intermediates is less satisfying than the direct analysis of experimental structures; however, this analysis only requires two endpoint structures. In general, techniques that attempt to derive the details of conformational changes from known crystal structures have a number of limitations. The analysis assumes that the appropriate conformations relevant to the solution conditions have been stabilized and characterized. For very flexible targets, all appropriate conformations are unlikely to have been observed. The analysis also assumes that the observed conformational changes will be relevant and are not, for example, artifacts stabilized by crystal-packing interactions. Despite these potential problems, any modeling of experimental SAXS data in which flexibility is suspected ought to include investigation of known experimental conformations (Section 4).

#### 3.3 Normal mode analysis

NMA is an effective computational tool for exploring the slow, large-scale motions by which macromolecules move. NMA derives 'fundamental' or 'essential' motions of a macromolecule via a simplifying assumption: distortions can be described by harmonic energy potentials (Brooks & Karplus, 1985). In essence, each atom in the structure is treated as if it were separated by springs from all other atoms in the structure, with the equilibrium position for each of these springs being defined as the distance observed in the static structure. These calculations can be performed in Cartesian or torsion angle space (Levitt *et al.* 1985). This description reduces the potentials to quadratic functions, making normal mode calculations much less computationally expensive than MD (Section 3.4), particularly for long time scales in which the MD trajectories can be inaccurate (Smith *et al.* 1995). The geometry of the atoms is more important for controlling the calculated normal modes than the details of the force field used, allowing further simplification in which interactions are only measured between adjacent atoms (Tirion, 1996) or between backbone atoms in reduced  $C\alpha$ -only representations of proteins (Hinsen, 1998; Bahar *et al.* 2001).

Most applications focus on the lowest frequency motions that describe the slower, larger scale motions, although the absolute magnitudes of these frequencies lack physical meaning as the force constants are arbitrary (Lindahl & Delarue, 2005). Additionally, experimental verification has been challenging, as the time scales of the large NMA motions are slower than rotational correlation times (Section 3.1). NMA is likely to be physically meaningful based on NMR relaxation data (Korzhnev *et al.* 2001) and analysis of conformational changes observed in high-resolution structures that can be modeled using combinations of the lowest frequency normal modes (Krebs *et al.* 2002). Normal modes can also provide the ability to generate systematically deformed molecules for agreement with experimental data and have been used for molecular replacement in protein crystallography (Suhre & Sanejouand, 2004), for refinement of crystal structures (Delarue & Dumas, 2004), and have real advantages for deforming atomic models to fit EM and SAXS data (Section 4.3.3).

## 3.4 MD simulations

MD has played an important role in many areas of structural biology (Karplus & McCammon, 2002). For bridging atomic resolution structures with SAXS data, it is most commonly used to generate a wide range of macromolecular conformations from which experimental signals are calculated and compared to measured results (Sections 4.2.2 and 4.4.2).

MD calculates the energies and dynamics of molecules with atoms being influenced by the classical equations of motion and forces due to bonding, electrostatics, and van der Waals

interactions. In most cases, the forces are not amenable to analytical solution, so they are solved numerically by evaluating the displacement of each atom for some tiny increment in time after which the forces are then re-evaluated. The force fields used in MD are, by necessity, approximations as most systems studied by MD far exceed the sizes that can be treated by quantum mechanical methods. Thus, force fields are created to define chemical parameters, including bond lengths, bond angles, torsion angles, planarity restraints, and non-bonded distances (MacKerell et al. 1998), as these features do not emerge from the calculation as they do from *ab-initio* quantum chemical calculations. The force fields can also be user-defined, which allows additional terms to be added or even changed during the course of the calculation. For example, the simulated annealing method of refining structures in X-ray crystallography and NMR (Nilsson et al. 1986; Brunger et al. 1987) applies a 'pseudo-energy' to the system to satisfy experimentally determined constraints in addition to those force-field terms necessary to satisfy chemical parameters. Deconvolution of the important motions from calculated trajectories (time-dependent changes in conformation) is also a non-trivial problem. ED, as described above (Section 3.2), has been used to reduce the population of structures into a subset of important motions dominating the simulation (Amadei et al. 1993). The large conformational changes have been useful for modeling in conjunction with SAXS data (Sections 4.2.2, 4.4.2).

## 3.5 Monte Carlo simulations

Monte Carlo simulations have also been used to sample configuration space in proteins (Binder & Heerman, 1992). The name comes from the application of random numbers to perturb features of the structure to prevent trapping of structures in local energy minima. Perturbations of the structure are accepted generally on the basis of an energy calculated for the new conformation. Classically, the Metropolis criterion is used, which evaluates steps on the basis of their energy change and the current 'temperature' of the system in the simulation (Metropolis *et al.* 1953), always allowing steps that decreases the energy of the system but applying a probabilistic rejection to steps that increase the energy of the system. As with MD simulations, a user-defined energetic force-field description is required to evaluate the conformation at each step; however, Monte Carlo simulations do not necessarily have to follow physically relevant trajectories to reach their final states. Monte Carlo algorithms have been extensively used in the study of protein folding (Hansmann & Okamoto, 1999), as well as modeling large rearrangements between protein domains (Maiorov & Abagyan, 1997). Monte Carlo simulations are proving quite useful for modeling of SAXS data, and we expect their use to allow continued advances in interpretation tools (Sections 4.4.2 and 5).

## 4. Using crystallography and SAXS to model structures

A variety of tools and techniques are employed to model SAXS data in conjunction with atomic models. A critical and non-trivial task is comparing scattering curves to each other (Section 4.1). This comparison is most commonly used to compare observed profiles to those calculated from atomic resolution models (Section 4.2). Several *ab-initio* methods have been developed to generate low-resolution envelopes, similar to those in EM reconstructions, that can be used for docking atomic structures (Section 4.3). More complicated analysis must be performed in cases where the macromolecules of interest have substantial flexibility and where the SAXS curve is

generated from a population of different conformers (Section 4.4). This final case is perhaps the most powerful use of SAXS, and methods are being actively developed. In each type of SAXS modeling, assessing the uniqueness of any particular solution is crucial; however, the inclusion of atomic resolution information provides strong constraints on possible solutions, making this type of modeling particularly powerful for understanding biological systems.

#### 4.1 Comparing SAXS profiles and assessing agreement

Fitting theoretical models to SAXS curves requires that a measure be established for determining the agreement between two scattering curves. In X-ray crystallography, this goodness-of-fit measure, the *R*-factor (Section 2.2.7) is well established. Lower *R*-factors correspond to the better fits between the calculated and experimental diffraction data. In contrast to X-ray crystallography, a multitude of different measures have been employed for SAXS. Although all of these measures are minimized on exact fits between two curves, these measures weight different portions of the scattering curve differently. Different weightings can strongly impact the results of modeling protocols.

A SAXS version of the *R*-factor has been employed in combining crystal and solution structures, which was defined by analogy with X-ray crystallography (Smith *et al.* 1990):

$$R = \frac{\sum_{i} \left| I(q)_{\exp} - I(q)_{calc} \right|}{\sum_{i} \left| I(q)_{\exp} \right|}.$$

The programs developed by Svergun and co-workers minimize the normalized discrepancy function  $\chi^2$ :

$$\chi^2 = \frac{1}{N_p - 1} \sum_{i} \left[ \frac{I(q_i)_{\exp} - cI(q_i)_{calc}}{\sigma(q_i)} \right]^2,$$

where *c* is a scaling factor and  $\sigma(q_i)$  is the experimental error. This measure clearly weights the lowest resolution data most strongly and is used in comparing experimental and theoretical curves in CRYSOL (Svergun *et al.* 1995). The database of theoretical protein scattering, DARA (Sokolova *et al.* 2003b; Section 2.4.3), implemented a 'weighted *R*-factor' to compare the scattering profiles  $I_1(q)$  and  $I_2(q)$ :

$$R_{I_1(q)I_2(q)} = \frac{\sum_i \left( \text{Weight}_i \cdot (\text{ScaFac}_i \cdot I_2(q_i)I_1(q_i)) \right)^2}{\sum_i I_1(q_i) \cdot \text{ScaFac}_i \cdot I_2(q_i).(\text{Weight}_i^2)},$$

where the scaling multiplier ScaFac yielding the best least-squares fit is

$$\operatorname{ScaFac} = \frac{\sum_{i} I_1(q_i) \cdot I_2(q_i) \cdot \operatorname{Weight}^2}{\sum_{i} I_2(q_i) \cdot \operatorname{Weight}^2},$$

with the weighting function Weight<sub>i</sub>= $q_i$ , corresponding to the weighting in the Shannon sampling theorem (Shannon & Moore, 1949), being the most sensitive. Furthermore, Sokolova *et al.* proposed to analyze separately the low angle range (q=0-0.15 Å<sup>-1</sup>), corresponding to the overall shape of the protein, and medium angle range (q=0.4-0.9 Å<sup>-1</sup>), which corresponds to information corresponding to internal structure.

The suite of SAXS modeling programs developed by Svergun and co-workers use target fitness functions of the form  $E = \chi^2 + \sum \alpha_n P_n$ , where  $\chi^2$  is the discrepancy and the penalty term  $\alpha_n P_n$  is used to apply external constraints on the solutions. For rigid-body modeling programs (Section 4.2), such as SASREF (Petoukhov & Svergun, 2005, 2006), the penalty terms  $\alpha_n P_n$ formulate the requirements of the absence of the overlaps between the subdomains. For *ab-initio* shape-restoration programs (Section 4.3), the penalty terms  $\alpha_n P_n$  can be used to enforce model connectivity. The program BUNCH, which combines both *ab-initio* and rigid-body modeling, has penalty terms that combine both types of constraints (Petoukhov & Svergun, 2005). In contrast, Chacon and colleagues tested several different fitness functions during the development of an *ab-initio* reconstruction program DALAI\_GA (Chacon *et al.* 1998, 2000). Of them, the comparison of intensities on a logarithmic scale rather than a linear scale by the *F*-factor gave the most promising results:

$$F = \left(\frac{1}{N_{\rm p}} \sum_{i} \left[\log\left(I_{\rm exp}(q_i)\right) - \log\left(I_{\rm calc}(q_i)\right)\right]^2\right)^{-\frac{1}{2}},$$

where  $N_{\rm p}$  is the number of points of the profile. In comparison to  $\chi^2$ , the *F*-factor gives a larger weight to the higher resolution data. The differences in the fitness functions in *ab-initio* programs have made it difficult to compare the scores of the final solution. In one study, a separate reciprocal space *R*-factor was used to make direct comparisons (Takahashi *et al.* 2003).

We are not currently convinced that a 'best' measure has been adequately developed; however, the proposed partitioning of the scattering profile in the 'weighted *R*-factor' scheme (Sokolova *et al.* 2003b) has the possibility of avoiding problematic fitting of the entire scattering profile. This parameter can deal with the problem that different parts of the scattering curve have different systematic and statistical errors with high *q* data ( $q > 0.4 \text{ Å}^{-1}$ ) being prone to large amounts of both. The existence of a standard *R*-factor measure in X-ray crystallography has been tremendously important in evaluating different crystallographic methods as well as providing insight for crystallographers as to 'acceptable' values for final refined structures. Hence, we believe that establishing a standard measure for SAXS will be important for the community as a whole. We note that as in X-ray crystallography reporting a standard 'goodness of fit' value can be independent of the particular fitness function used for refinement. Thus adoption of a standard need not constrain the development of new protocols for fitting atomic or *ab-initio* models to SAXS curves.

#### 4.2 Direct comparison of crystallographic structures with SAXS data

If the crystal structure of a macromolecular sample is known, a theoretical solution scattering pattern can be calculated from the atomic coordinates (Section 4.2.1). This provides the opportunity to evaluate computationally generated models under situations where the curve computed from the crystallographic coordinates displays significant deviations from the experimental scattering. Though this is one of the most straightforward uses of SAXS, the uniqueness of arrangements of atomic resolution structures that fit SAXS data must also be evaluated, especially in the case of molecules that have flexible linkers. Atomic structures greatly constrain possible SAXS models. These structures can be used to perform rigid-body modeling in situations where individual components, but not the assembly are known (Section 4.2.2). Computed

scattering curves are extremely valuable as they can also be used to screen through potential solutions generated by computational docking (Section 4.2.3), to evaluate how solution structures differ from crystal structures (Section 4.2.4), and to establish which contacts in a crystal structure are the biologically relevant solution assemblies (Section 4.2.5).

#### 4.2.1 Calculation of scattering curves from atomic models

Theoretical SAXS curves can be directly calculated from atomic models. The observed scattering profile is largely the difference between the scattering of the target molecule with its ordered solvation layer and the excluded volume that takes into account the missing scattering of bulk solvent due to the existence of the solute. The excluded volume term can be determined by defining the shape of the molecule and calculating the scattering from it as if it were filled by an electron density equivalent to bulk solvent (Fraser *et al.* 1978; Lattman, 1989; Svergun *et al.* 1995).

In principle, the P(r) function and hence the scattering from the solute can be calculated by evaluating all of the interatomic distances in the structure: a procedure that scales by the square of the number of atoms. This algorithm is inappropriate for fitting purposes for which calculated scattering must be calculated thousands of times. Faster alternatives include the calculation of the P(r) function using Monte Carlo integration routines (Zhao *et al.* 1998) and is implemented in ORNL\_SAS (Tjioe & Heller, 2007) or from spherical harmonics (multipole expansion) envelopes that cover the entire model as implemented in CRYSOL (Svergun *et al.* 1995). The spherical harmonics procedures scale as  $N(L_{max} + 1)^2$ , where the default values of  $L_{max}$  is typically in a range from 15 to 17 with different programs.

Although solute atoms dominate the scattering signal at small angles, the scattering from ordered solvent atoms must also be considered (reviewed in Perkins, 2001). For proteins, this ordered solvation layer corresponds to 0.3 g water per g protein and is 15% denser than bulk water (Merzel & Smith, 2002a) due to both geometric effects and changes in the water structure such as shorter oxygen–oxygen distances and increased water coordination numbers. Strongly bound water molecules are known to fill surface grooves and channels stabilizing their structures and smoothing the excluded volume (Kuhn *et al.* 1992). Computationally, the hydration shell has been modeled by explicitly placing water molecules on the surface (Hubbard *et al.* 1988; Grossmann *et al.* 1993; Fujisawa *et al.* 1994) or by surrounding the particle by a continuous envelope representing the solvation shell of 3 Å with a density that can differ both from bulk density and the solute (Svergun *et al.* 1995). Including the hydration shell improves the accuracy of the calculated scattering profiles; however, the contribution from the ordered solvent layer is several orders of magnitude lower than the scattering from the solute and the excluded volume.

Fitting experimental scattering at higher resolutions  $(q > 0.4 \text{ Å}^{-1})$  is more problematic for spherical harmonic reconstructions, as they do not account for the internal structure of the scattering particles. Even globular proteins, such as glucose isomerase, typically require extreme values for adjustable parameters in CRYSOL to fit the high-resolution portion of the experimental curve (Fig. 8*a*). Calculation of scattering profiles by algorithms that explicitly include all atoms are more accurate but require more intensive computation (Merzel & Smith, 2002b). For example the program solX (Tiede & Zuo, 2007) shows good agreement throughout the *q*-range with measured scattering profiles using default settings and values even without a solvation layer. Fitting proteins with flexibility at high *q* values is more problematic, as exemplified by a truncated form of Cel5A (Fig. 8*b*), for which multiple conformations must be



**Fig. 8.** A. Comparison of experimental scattering curves of glucose isomerase (black) with the scattering curve calculated from the atomic model using the program CRYSOL (Svergun *et al.* 1995) with the default parameters (red line; solvation shell contrast  $0.03 \text{ e/Å}^3$ , average atomic radius 1.61 Å, excluded volume  $2\cdot13 \times 10^5 \text{ Å}^3$ ) or adjusted parameters (blue line; solvation shell contrast  $0.005 \text{ e/Å}^3$ , average atomic radius 1.4 Å, exclude volume  $2\cdot29 \times 10^5 \text{ Å}^3$ ). Theoretical scattering calculated with program solX using allatom methods is shown in green line (Tiede & Zuo, 2007). B. Comparison of the experimental scattering curves of truncated Cel5A (black) with the scattering curves calculated using CRYSOL with default parameters (red line; solvation shell contrast  $0.03 \text{ e/Å}^3$ , average atomic radius 1.61 Å, excluded volume  $5\cdot35 \times 10^4 \text{ Å}^3$ ), CRYSOL with adjusted parameters (blue line; solvation shell contrast  $0.01 \text{ e/Å}^3$ , average atomic radius 1.8 Å, excluded volume  $5\cdot48 \times 10^4 \text{ Å}^3$ ), or solX using all atom-methods (green line).

explicitly included (Section 4.4). Moreover, the theoretical calculation of higher resolution scattering  $(q > 0.25 \text{ Å}^{-1})$  is influenced by atomic displacement factors (Section 2.2.8) in the atomic model (ADFs; Zhang *et al.* 2000a). The crystallographically refined ADFs may not be appropriate for molecules in solution; hence, proper calculation of higher resolution scattering curves remains an unsolved problem.

In many of the modeling schemes described in the following sections, the methods for manipulating the structures of interest are directly tied to the calculation engines for generating the theoretical scattering curves for the models. In some sense, this tight coupling and limited scripting abilities have introduced pratical limitations for altering modeling schemes. In X-ray crystallography, X-PLOR and its successor CNS (Brunger *et al.* 1998) were designed to be highly modifiable, which we believe is part of its success as becoming an important crystallographic refinement package. Thus, we are excited about the stated intent of the authors of ORNL\_SAS (Tjioe & Heller, 2007) to ensure that the scattering-curve calculation engine is easily scriptable for integration so that other programs can be readily integrated into SAXS modeling even if these programs were never written with X-ray scattering in mind.

#### 4.2.2 Rigid-body modeling with SAXS data

When atomic resolution structures of individual domains are known, SAXS can be used to determine the relative orientation and placement of these domains in a complex by maximizing the agreement between the theoretical and experimentally observed scattering (reviewed in Wall *et al.* 2000). In rigid-body modeling, the domains are considered static objects and only their relative orientations are changed. In general, building an assembly with M subunits has 6(M-1)
Program	Method	Reference
MASSA, ASSA DIMFOM	Manual domain docking Fast algorithm for modeling globular homodimeric or heterodimeric structures.	Kozin <i>et al.</i> 1997; Konarev <i>et al.</i> 2001 Petoukhov & Svergun, 2005
GLOBSYMM	Brute-force rigid-body modeling of symmetric oligomers	Petoukhov & Svergun, 2005
SASREF	Rigid-body modeling of a complex that can simultaneously fit scattering from subcomplexes	Petoukhov & Svergun, 2005, 2006

Table 2. Rigid-body modeling SAXS programs

fittable parameters, as one subunit can be fixed and M-1 subunits are mobile with three translations and three rotations each. The number of parameters can be reduced if symmetry is present, such as with homo-oligomers. Depending on the problem, rigid-body modeling with atomic models can reduce the dimensionality of the fitting problem and be a more economical and powerful way to use the information content in the SAXS curve.

Evaluation of each trial rotation and translation in the rigid-body search can be computationally expensive. A brute-force search scales as  $n^{6(M-1)}$ , where *n* is the number of steps searched for each parameter. Different simplifications, however, can be performed for rapid screening. An important simplification is the representation of rigid bodies in forms from which scattering profiles are easily computed. For example, individual components have been modeled using triaxial ellipsoids (Gallagher *et al.* 1999), such as in the case of a heterodimeric cAMPdependent protein kinase which was used as a basis for construction of an atomic model by energy minimization (Tung *et al.* 2002). Spherical harmonic representations calculated from atomic models have also been applied (Svergun *et al.* 1995) using the algorithm of Svergun and Stuhrmann (Svergun & Stuhrmann, 1991).

Several programs are freely available for rigid-body docking using these spherical harmonic envelopes of atomic domains (Table 2). Interactive docking can be performed using ASSA (Kozin *et al.* 1997) or MASSHA (Konarev *et al.* 2001); however, in many cases, being able to generate automated rigid-body modeling is preferable (Koch *et al.* 2003; Petoukhov & Svergun, 2005; Svergun, 2007). For two-component complexes, DIMFOM exhaustively searches rotation space by rolling one monomer over the other. For larger symmetric complexes that reduce the dimensionality of the search space, GLOBSYMM is appropriate. For complexes containing several subunits that may or may not be symmetrically related, a heuristic search algorithm has been implemented in the program SASREF. An advantage of SASREF is the ability to use additional constraints to incorporate other information about the system, such as known subunit interfaces (Mattinen *et al.* 2002; Grishaev *et al.* 2005). Recently SASREF was used to simultaneously fit X-ray and neutron-scattering curves that included contrast variation data from selectively deuterated complexes (Gherardi *et al.* 2006).

With symmetry constraints, rigid-body modeling has been successful at constructing the dimeric complexes of hepatocyte growth factor and tyrosine kinase with p2 point-group symmetry (Gherardi *et al.* 2006) or glyceraldehyde-3-phosphate dehydrogenase (Ferreira da Silva *et al.* 2006) and glutamate synthase (Petoukhov *et al.* 2003) with p222 point-group symmetry. Even with the small number of parameters of rigid-body modeling, the small number of independent data values can lead to overfitting (Section 2.3.4). From a practical standpoint, many models should

Program	Method	Reference
GLOOPY	Individual residues are modeled with spring forces between neighbors	Petoukhov et al. 2002
CHADD	Dummy atoms with spring forces between neighbors	Petoukhov et al. 2002
CHARGE	Restore structures as random-walks with secondary-structural features added when known	Petoukhov et al. 2002
CREDO	Dummy atom construction of missing loops	Petoukhov et al. 2002
BUNCH	Brute force rigid-body modeling asymmetric oligomers with reconstruction of missing linkers and domains	Petoukhov & Svergun, 2005

Table 3. Reconstruction of missing components

be generated independently and examined for uniqueness and tested by other biophysical techniques. For example, automated rigid-body modeling of bacterial release factor 1 (RF1) generated multiple solutions with similar fits to the experimental data (Vestergaard *et al.* 2005). These models were evaluated by SAXS analysis of additional deletion constructs and by cryo-EM studies.

Rigid-body modeling can also be combined with *ab-initio* modeling (Table 3). The unknown portions of the macromolecule may be modeled as in *ab-initio* methods (Section 4.3.1) where each amino acid is represented by 'dummy residues' (also called 'beads' or 'big atoms'). The program BUNCH models full-length proteins as rigid-body domains linked by their termini to flexible chains or domains (Petoukhov & Svergun, 2005) and determines a best-fit conformation through a simulated annealing protocol (e.g. Fig. 9). Flexible regions have also been modeled using a Monte Carlo dihedral angle sampling of hinge regions (Akiyama et al. 2004) and an 'automated constrained fit' procedure generates thousands of possible models by applying MD on the linker region in exhaustive search of the best-fit conformation (Boehm et al. 1999) applied to a number of human complement regulating proteins (Aslam & Perkins, 2001; Aslam et al. 2003; Sun et al. 2004; Gilbert et al. 2005, 2006) and the cellulosome (Hammel et al. 2005). Frequently, the simulation is performed on the linker regions at very high temperature ( $\sim 1500$  K) to prevent the molecule from becoming trapped in a local minimum (Leach, 2001). Different conformations of the protein were produced at regular intervals along the trajectory of subsequent calculations of the theoretical SAXS profiles. The comparison of the MD-generated configurations with the experimental data enabled discrimination of a finite number of structures with the best fit and with RG closest to the experimental values (Hammel et al. 2005). This modeling, however, is likely to be inappropriate for macromolecules with substantial flexibility that must be treated as conformational ensembles lacking a 'best-fit' structure (Section 4.4).

## 4.2.3 Computational docking combined with SAXS data

Atomic structures have long held the promise of allowing the structures of unknown complexes to be determined computationally. Substantial effort has been directed towards using only chemical and geometric parameters to determine docking sites (Mandell *et al.* 2001; Bonvin, 2006); however, this problem has not proven to be simple. Computational docking has two major challenges: (1) dealing with flexibility in macromolecules in a computationally practical manner and (2) determining an appropriate scoring function that successfully distinguishes



**Fig. 9.** Experimental SAXS curves (black circles) and scattering profiles computed from the models of a multidomain protein reconstructed by the rigid-body modeling using BUNCH (red line; Petoukhov & Svergun, 2005) and *ab-initio* model using the program GASBOR (green line; Svergun *et al.* 2001a). The secondary structural elements of the known atomic structures of the single modules are shown by a multicolor ball representation (yellow, gold, gray, red). Restored linker conformations between the modules are displayed by a blue beads representation and varied from run to run. Single best-fit models did not fit the smallest *q* region (q < 0.03 Å<sup>-1</sup>) well, but fits involving multiple models did. The position of the unknown structure of the small domain has been modeled as the globular region at the position correlated with its position in the sequence (marked with the arrow). One GASBOR model is shown as gold beads superposed on the average shape shown as light blue spheres.

correct solutions. A number of different strategies have been developed to introduce flexibility. Protocols allowing some degree of interpenetration of the molecules ('soft docking'), side-chain flexibility, and truncating surface side-chains have been developed (Schnecke *et al.* 1998; Palma *et al.* 2000; Chen *et al.* 2003; Heifetz & Eisenstein, 2003; Li *et al.* 2003). Additionally, docking has also employed ensembles of conformation (Smith *et al.* 2005) from NMR structure determination (Dominguez *et al.* 2004), MD simulations (Rajamani *et al.* 2004), NMA (Mustard & Ritchie, 2005), or residual dipolar coupling (Tai, 2004). A number of web servers generate conformational ensembles specifically for docking (Lei *et al.* 2004; Suhre & Sanejouand, 2004; Barrett & Noble, 2005). Regardless of the protocol involved, an energy minimization is frequently run for top solutions to resolve steric clashes at the interface (Li *et al.* 2003).

SAXS can provide an experimental target function to score hits from the docking search and can substantially aid in the problem of development of scoring functions that successfully distinguish between good and bad solutions. For example, this combination has been used to show that the histone-like domain of the Ras activator son of sevenless folds onto a helical linker between two other son-of-sevenless domains (SOS; Sondermann *et al.* 2005). In this work, the docking of the SOS<sup>DH-PH-cat</sup> and SOS<sup>Histone</sup> crystal structures was performed separately from comparisons to SAXS results. Of the top 40 docking solutions, the top-scoring solution was also the best fit to the SAXS. Computation docking was also used to generate 100 dimeric structures of purine nucleoside phosphorylase and to verify that the crystallographic trimer was the

### 230 C. D. Putnam et al.

assembly state in solution (Filgueira de Azevedo *et al.* 2003). We anticipate that computational docking of the supramolecular complexes and validation of models by SAXS will become a particularly powerful combination.

### 4.2.4 Differences in crystallographic and solution structures

The ability to directly compare atomic models to SAXS curves is a remarkably useful tool for deciphering the influences that the crystal lattice has on the observed structure. The crystal structure is likely the lowest energy state within the lattice and under crystallization conditions; however, it not necessarily the lowest energy state in solution. Many studies have suggested that the effects of the crystal lattice do not alter the folding of domains, but rather influence the conformations adopted by flexible termini or linkers between domains. Moreover, proteins that undergo allosteric rearrangements and have multiple stable conformations separated by small energy differences can be affected by lattice forces. The classic example for incompatibility between crystal lattices and conformational states is the cracking of reduced deoxyhemoglobin crystals upon the addition of oxygen (Perutz *et al.* 1964). By their very nature, crystal structures tend not to reveal the flexibility of the crystallized macromolecules. Crystal structures, such as for the c-Abl protein kinase (PDB id 10pl; Nagar *et al.* 2003) that has two molecules in the asymmetric unit with substantially different domain arrangements tend to be the exceptions, so we favor continued research to optimize the combination of computational searches and SAXS data to characterize solution conformations.

A naive assumption is that crystallization will mostly force macromolecules to adopt a more compact structure. SAXS has revealed that crystallographically induced compaction is likely to be frequent, such as with the ligand-bound 'relaxed' state (but not the unliganded 'tense' state) of aspartate transcarbamoylase. These differences, however, could be modeled by small rigid-body subunit rotations on the order of 10 degrees (reviewed in Koch *et al.* 2003). Despite these sorts of results, compaction by the lattice is not the only possible effect. For example, in a study of four thiamine diphosphate-dependent enzymes, one was found to be identical in the crystal and in solution, two appeared to be less compact in solution, and one complex with weaker subunit interactions was clearly more compact in solution (Svergun *et al.* 2000).

The differences between solution and crystal structures can be important for deciphering biological mechanisms. For example, the single conformation of the dimeric bacterial DNA mispair recognition protein MutS is likely constrained by lattice forces, despite the fact that multiple states are expected from biochemical results. MutS recognizes mispairs in DNA when in a nucleotide-free or ADP-bound state. Binding of ATP by bacterial or eukaryotic versions of the DNA-MutS complex induces a state in which the protein acts like a rigid ring that can freely diffuse along the length of the DNA and is no longer restricted to the mispair (Mendillo et al. 2005). Crystal structures of MutS-DNA complexes with ATP or ATP analogs have yet to reveal the conformational changes anticipated from biochemical studies (Lamers et al. 2004). These predicted domain motions that would allow the two halves of the composite ATP-binding pockets to engage with their bound nucleotide have been anticipated by the effects of dominant mutations in the eukaryotic homologs (Hess et al. 2002) and studies in related systems for which both states are known (Hopfner et al. 2000). Regardless of whether or not these conformational changes are incompatible with the crystal lattice or due to the concentration effects of the trapped DNA, attempts to decipher these conformational changes have not yet been successful through crystallography alone. Properly assessing these conformational changes in the case of

the efficient DNA repair machinery is important as toxins that interfere with the repair machinery are expected to be more dangerous to public health than toxins that directly damage DNA (McMurray & Tainer, 2003).

### 4.2.5 Determining biological assemblies in crystal structures

Crystallization typically requires the formation of symmetric contacts that exist solely in the crystal and may not be important in biology. However biologically relevant multimers can also be symmetric, and biologically relevant symmetry operators can also be part of the underlying symmetry of the crystal structure so the contents of the asymmetric unit cannot be used as a guide for the biological assembly. Thus, crystallographers must decipher which macromolecular contacts mediate multimerization in solution and which are only crystal contacts.

In many cases identifying the biological multimer can be a difficult problem. Systematic investigation of atomic resolution structures has shown that authentic interfaces tend to be large and involve hydrophobic interactions on the surfaces of the protein (Jones & Thornton, 1996) and tend to be better packed than crystal contacts (Li *et al.* 1985; Getzoff *et al.* 1986; Bahadur *et al.* 2004; Keskin *et al.* 2005). Additionally, these interfaces are less hydrated (Rodier *et al.* 2005) and tend to have greater sequence conservation (Valdar & Thornton, 2001a, b) with frequent tryptophans, phenylalanines, and methionines (Ma *et al.* 2003). These analyses, although useful in the absence of other information, are theoretical in nature. SAXS data collected on the biologically relevant assemblies in solution has the potential to provide experimental information to help test and identify the biologically relevant interfaces in the crystal.

Identifying biological dimers is particularly difficult, as crystallographic symmetries tend to generate many dimeric interactions. For example, the bacterial mismatch repair protein MutL is known to form functional dimers. MutL is comprised of an N-terminal ATPase domain, a flexible internal linker, and a C-terminal domain that mediates constitutive dimerization (Drotschmann *et al.* 1998). The dimeric C-terminal domain crystallized in the space group P4<sub>3</sub>22 with four different dimer interfaces (PDB id 1x9z; Guarne *et al.* 2004). Two interfaces have remarkably small buried surface areas, but two others were more sizable. The dimer interface proposed by Guarne *et al.* later proved to be incorrect, whereas the other large dimer interface was consistent with crosslinking experiments and identifiable by computations analysis of packing (Kosinski *et al.* 2005). The biologically relevant dimer is more extended than the dimers generated by crystal contacts, and theoretical calculations suggest that SAXS could have easily distinguished it by size ( $R_G = 34 \cdot 1$  Å,  $D_{max} = 120$  Å *vs.*  $R_G = 28 \cdot 1$  Å,  $D_{max} = 80$  Å) as well as by the P(r) function that is characteristic of an elongated shape, having a peak at short distances and a long extended tail ( $R_G = 28 \cdot 1$  Å,  $D_{max} = 80$  Å) (Fig. 10).

A more insidious challenge is when the authentic multimer does not exist in the crystal at all. Generating a solution assembly model given only the high resolution structure of a subunit is possible with SAXS data, and modeling assemblies with SAXS data will be discussed below. But before modeling can be performed, it is critical to identify that the biologically relevant assembly is not present in the crystal and to recognize that modeling must be done in the first place. In these cases, the lack of agreement between SAXS data collected on multimers in solution and theoretical scattering curves calculated from assembles observed in the crystals could be an important clue. For example, the hexameric assembly for the Holliday junction ATPase motor RuvB could be characterized by SAXS results in combination with the crystal structure of the subunit from a non-biologically relevant screw assembly (Putnam *et al.* 2001).



**Fig. 10.** SAXS could have readily distinguished between alternative dimer structures of the C-terminus of MutL observed in the crystal structure (PDB id 1x9z; Guarne *et al.* 2004). (*a*) Each of the four different dimers has remarkably different overall shapes, giving rise to measurable differences in SAXS and parameters such as  $R_G$  and  $D_{max}$ . Dimer 1, with a buried surface area of the monomer of 755 Å<sup>2</sup>, is the asymmetric unit of the crystal, whereas dimer 2, with a buried surface area of 923 Å<sup>2</sup>, is the solution dimer assembly (Kosinski *et al.* 2005). (*b*) Theoretical P(r) functions calculated for dimer 1 (black) and dimer 2 (red) are readily distinguished. Dimer 1 has a characteristic globular P(r) that is bell-shaped, whereas dimer 2 has a characteristic extended P(r) with an early peak and a long tail.

One particularly difficult problem was identifying the biological dimer of ATP-binding cassette (ABC) transporters (Fig. 11). ABC transporters couple ATP hydrolysis with the transport of ligands across membranes and include the cystic fibrosis transmembrane regulator. The nucleotide-binding domains of the ABC ATPase transporters were known to dimerize. The first of these structures, the *Salmonella typhimurium* histidine permease HisP was crystallized in the space group P4<sub>3</sub>2<sub>1</sub>2 with two symmetry-related dimers in the crystal packing (PDB id 1b0u;



**Fig. 11.** SAXS has the potential to indicate mismatches between the dimers in the ABC ATPase assemblies observed in crystal and with their solution conformation, if dimeric states could have been stabilized for solution scattering. (*a*) The biological dimeric assemblies of the ABC ATPases were first observed with the DNA repair proteins Rad50 (PDB id 1f2u; Hopfner *et al.* 2000) and MutS (PDB id 1e3m, Lamers *et al.* 2000; PDB id 1ewq, Obmolova *et al.* 2000) that were normally stable as dimers in solution. Subunits are displayed as yellow and red ribbons; bound nucleotides are orange ball-and-stick models, and blue ribbons indicate the position of the signature motif that forms the second half of the ATP-binding surface. (*b*) Different crystallographic dimers were observed for HisP, MalK, MJ0796, and MJ1267 [PDB ids 1b0u, 1g29, 1f3o, 1g6h (Hung *et al.* 1998; Diederichs *et al.* 2000; Karpowich *et al.* 2001; Yuan *et al.* 2001] that lack special positioning of the signature motifs. Theoretical *P*(*r*) scattering curves clearly distinguish between the crystallographically observed dimers and those modeled using the Rad50 assembly.

Hung *et al.* 1998); the largest dimer interface was proposed to be the biologically relevant one. Other transporter ABC ATPases were solved later, including the *Thermococcus litoralis* maltose transporter MalK (PDB id 1g29; Diederichs *et al.* 2000), and the *Methanococcus jannaschii* ABC ATPases MJ0796 (PDB id 1f30; Yuan *et al.* 2001) and MJ1267 (PDB id 1g6h and 1gaj, Karpowich *et al.* 2001). Remarkably, each of these ABC ATPases had dimers generated by



**Fig. 12.** The two different HslUV protease/chaperone assemblies observed crystallographically have different orientations of the HslU hexamers (green) that interact on both sides of the double-ringed HslV dodecamer (gold). In the first, the globular domains of HslU are separated from HslV, giving a bimodal P(r) function with a  $R_G$  of 90 Å and a  $D_{max}$  of 265 Å (PDB id 1doo; Bochtler *et al.* 2000). In the second, the globular domains of HslU pack against HslV, giving a monomodal P(r) function with a  $R_G$  of 65 Å and a  $D_{max}$  of 220 Å (PDB id 1g3i; Sousa *et al.* 2000). The experimental P(r) function (not shown; Sousa *et al.* 2000) strongly argues for a monomodal distribution and the more compact 1g3i-like assembly (after Sousa *et al.* 2000).

crystallographic symmetry, yet none shared common dimer interfaces. The biologically relevant dimer was first proposed using sequence analysis and the structure of the HisP monomer (Jones & George, 1999). Dimerization generates an 'ATP sandwich' with the active site being made up of a Walker A ATP-binding and hydrolysis site on one molecule and an ABC ATPase 'signature' motif from the other subunit. These interfaces were first observed crystallographically in the ABC ATPase domains from the DNA repair proteins Rad50 (PDB id 1f2u; Hopfner *et al.* 2000) and MutS (PDB id 1e3m; Lamers *et al.* 2000; PDB id 1ewq; Obmolova *et al.* 2000), and the Rad50 structure in particular was subsequently used for modeling of the ABC transporters. The inference that the membrane transporters shared the Rad50/MutS dimer assembly was later established by the crystal structure of the *E. coli* BtuCD heterotetramer (PDB id 117v; Locher *et al.* 2002).

SAXS can be used to distinguish between alternative assemblies in different crystal structures. The bacterial HslUV chaperone/protease complex had been observed in two distinct crystallographic assemblies (Fig. 12; Bochtler *et al.* 2000; Sousa *et al.* 2000). In the center of both assemblies was an HslV hexamer, but in the two different crystal forms the globular N- and C-terminal domains of HslU were either packed against HslV (PDB id 1g3i; Sousa *et al.* 2000) or held away from HslV by an extended internal domain (PDB id 1doo; Bochtler *et al.* 2000). The P(r) functions for these different assemblies and hence their X-ray scattering were easily distinguishable and SAXS revealed that in solution the globular N- and C-terminal domains of HslU pack against HslV and that the extended HslU domain extends into solvent (Sousa *et al.* 2000).

Finally, SAXS can establish the validity of weak assemblies observed in crystal packing that might be ignored as artifacts of the crystallization process. The *E. coli* mismatch repair protein MutS forms a dimer that recognizes mispairs in DNA. At high protein concentrations, MutS forms tetramers that depend upon the presence of the C-terminal 53 amino acids (Bjornson *et al.* 2003). Fusions of this C-terminal domain onto maltose binding protein (MBP) mediated its tetramerization, and the crystal structure revealed a helix–loop–helix domain that made a symmetric interaction to form a tight dimer (PDB id 20k2; Mendillo *et al.* 2007). A potential tetramer contact was observed in the crystal form; however, the interface was small and weakly packed. Despite this, SAXS data collected on the MBP fusion protein and subsequent mutagenesis of salt bridges that stabilized it revealed that the tetramer in solution closely resembled the assembly observed crystallographically. Similarly, a combination of SAXS, ultracentrifugation, and EM was used to validate the biologically relevant Mre11<sub>2</sub>/Rad50<sub>2</sub> heterotetrameric DNA processing head used in double-strand break repair, whose structure was inferred from the atomic resolution structures of Mre11, Rad50, and biochemical results (Hopfner *et al.* 2001).

### 4.3 Modeling atomic assemblies using ab-initio SAXS structures

Substantial theoretical and practical work has gone into establishing that three-dimensional reconstructions can be derived from the one-dimensional SAXS curve (Section 4.3.1). Successful reconstructions generate low-resolution envelopes that are analogous to averaged reconstructions generated by EM. Similarly, many of the tools used in EM for fitting known atomic assemblies into these envelopes can also be used for docking atomic structures in a rigid-body fashion (Section 4.3.2) or using methods that allow flexibility to be introduced into the atomic models (Section 4.3.3). Introducing flexibility in the atomic model is valuable in situations where the three-dimensional reconstruction involves a state of the macromolecule that is different from the state that was used for determining the atomic resolution structure. The *ab-initio* envelope reconstruction protocols substantially differ from those using atomic models described above (Section 4.2). In these protocols, all of the independent experimental values extracted from the SAXS curves are used for generating an overall envelope, which frequently has a uniqueness problem. Depending on the system, we suggest that rigid-body modeling of a small number of subunits is a more powerful use of the information content in SAXS curves than ab-initio envelope reconstruction; however, these modeling methods are independent and can be performed in parallel. Similar answers deduced by each technique can help give confidence in the solutions; although we believe that *ab-initio* models are best treated as hypotheses to be tested by additional experimentation (Fig. 13).

## 4.3.1 Calculation of ab-initio SAXS envelopes

Several programs have been created for the purpose of calculating so called *ab-initio* shapes from scattering profiles (Table 4). In practice many of these programs use information external to the scattering profiles and the term *ab initio* solely refers to lack of a pre-defined input structure. These programs implicitly or explicitly assume the shape is a continuous object, which substantially reduces the search space. We do not attempt to provide detailed descriptions of these



Fig. 13. SAXS models of the complexed cellulase can be reconstructed using different methods. (*a*) *Ab-initio* model reconstructed by spherical harmonics using SASHA (Svergun *et al.* 1996). (*b*) Densely packed beads model using DAMMIN (Svergun, 1999). (*c*) Model calculated with GASBOR (Petoukhov *et al.* 2002) (*d*) Reconstruction of one missing module using dummy residues with CREDO (green) (Petoukhov *et al.* 2002). The secondary structural elements of fixed known atomic structure are in gray. (*e*) Rigid-body modeling applied on the knowing atomic structures (gray) in the combination with *ab-initio* models, by applying molecular dynamics on the linker region (blue) have been used in an exhaustive search of the best-fit conformation. One hundred conformations (yellow) are shown superimposed on the catalytic module (gray) (Hammel *et al.* 2004a).

Program	Characterization	Advantage	Reference
DALAI_GA	Fits the scattering profile computed from dummy beads models of the molecule, using the Debye formula. Bead models are optimized with a genetic algorithm that searches the huge space of possible mass distributions and evolves convergent models	Missing connectivity penalty lead to excellent fit for unfolded protein	Chacon <i>et al.</i> 1998, 2000
DAMMIN	Searches for a compact dummy beads configuration minimizing the discrepancy function applying the non-compactness penalty using the simulating annealing starting from a random configuration	Optional use of symmetry constraints, loose and disconnected shapes are penalized, in the expert mode adjusting of penalties	Svergun, 1999
ELLSTAT	Builds models using combination of simple shapes including ellipsoids and cylinders	Not an over-parameterized reconstruction of the observed scattering	Heller, 2006
GASBOR	Similar to DAMMIN. The protein structure is represented not by densely packed dummy beads but rather by an ensemble of dummy residues forming a chain-compatible model. The spatial positions of these residues aim at approximately corresponding to those of the $C\alpha$ atoms in the protein structure. The number of residues should be equal to that in the protein	Beads are on a chain with centers 3.8 Å apart to simulate protein and provide additional constraints, realistic models for multidomain particles with extended linker or hinge region	Svergun <i>et al.</i> 2001b
GA_STRUCT	Uses a genetic algorithm and Monte Carlo approaches, no fixed grid and dummy beads radii are fixed by resolution of data		Heller <i>et al.</i> 2003
SASHA	Allows determining the low-resolution shape of a simply shaped homogeneous particle applying the spherical harmonics. No extra information is required	Fast calculation, not an over-parameterized reconstruction of the observed scattering	Svergun <i>et al.</i> 1996
SASMODEL	Uses Monte Carlo approaches, chain of ellipsoids that rotate about attachment point and no fixed grid	Volume constraints	Zhao <i>et al.</i> 1998
SAXS_3D	Based on Monte Carlo approach without any limitation of search space using 'give and take' approach	Straightforward modeling without any a priori information, fast calculation	Walther <i>et al.</i> 2000

Table 4. Ab-initio SAXS envelope reconstruction programs

programs here and the citations listed in Table 4 should be referenced. The general approach taken by these programs is to propose shapes, calculate scattering curves or P(r) functions and optimize the agreement to the experimental data (Fig. 13).

Many of the early programs restricted searches to shapes defined by a small number of parameters. For example,  $R_G$  can be thought of as describing an *ab-initio* model with a single parameter: a spherical envelope of radius  $r=(5/3)^{\frac{1}{2}}R_G$ . Whole-body methods approach the problem by attempting to fit X-ray scattering using spheres, oblate, prolate, and triaxial ellipsoids and are also used for modeling hydrodynamic properties (Rallison & Harding, 1985). A modern

version of this algorithm that can also combine several simple shapes is available with the program ELLSTAT (Heller, 2006). Spherical harmonics also have been used (Svergun & Stuhrmann, 1991) as implemented in SASHA (Fig. 13*a*; Svergun *et al.* 1996). However, spherical harmonics descriptions cannot properly represent all shapes, such as structures with cavities or holes. In the specialized case of icosohedrally symmetric virus particles, icosohedral harmonics have also been employed to describe low-resolution structures (Zheng *et al.* 1995).

Recently, a number of over-parameterized methods have been used for *ab-initio* shape determination (Volkov & Svergun, 2003). These methods attempt to generate a bead model (or dummy atom model) to fill a volume consistent with the experimental scattering. External constraints, such as smoothness, connectivity, and particle symmetry have been used to reduce the search space (Koch *et al.* 2003). The number of possible arrangements increases combinatorially, so all programs use computational tricks to search through the solution space including Monte Carlo approaches, genetic algorithms, and simulated annealing (Table 4). Several programs fit directly to the scattering data involving a Fourier transform for each shape proposed. This becomes computationally intensive and cheap methods of calculating scattering curves from proposed shapes have been developed (Section 4.1).

Shape-restoration programs using bead models do not uniquely define the position of each bead within a volume, nor do the beads represent positions of specific residues. Rather, the bead positions are non-unique and define a volume for the scattering particle, as illustrated by six independent GASBOR reconstructions of the human OGG1 DNA glycosylase/lyase (Fig. 14). Each of the reconstructions fits the experimental scattering equally well and describes similar shapes. GASBOR is designed specifically for proteins (though modeling of other macromolecules is possible); dummy atoms have radii approximating amino acids with penalties imposing chain connectivity. In the case of OGG1, the envelope generated by averaging the different GASBOR runs fits the volume and shapes the truncated crystal structure (Fig. 14).

The final resolution attainable from *ab-initio* envelopes varies with data quality, the program used, particle size, shape and flexibility (Section 5.6). Several studies have compared available programs (Takahashi *et al.* 2003; Zipper & Durschschlag, 2003); however, a systematic study on the accuracy of *ab-initio* envelopes has not been conducted. Figure 6 shows 10 *ab-initio* reconstructions and the average shape from the highly flexible detergent solubilized apoB-100 generated by DAMMIN. A larger variation exists between the models in comparison to OGG1 (Fig. 14); although, the overall dimensions are in agreement and suggest relative domain motions. Very flexible structures typically have fewer features in their scattering profiles, which can be fit by a greater variety of shapes. A comparison of the output of multiple independent modeling runs (at least 6–10) provides some measure of the uniqueness of the models.

One parameter used to characterize the agreement among models is the normalized spatial discrepancy (NSD) (Kozin & Svergun, 2001). Briefly, if models 1 and 2 are expressed in two sets of points  $P_1 = \{p_{1i}, i=1, ..., N_1\}$  and  $P_2 = \{p_{2i}, i=1, ..., N_2\}$ , then, NSD between  $P_1$  and  $P_2$  is defined as

$$\rho(P_1, P_2) = \left\{ \frac{1}{2} \left[ \frac{1}{N_1 d_2^2} \sum_{i=1}^{N_1} \rho^2(p_{1i}, P_2) + \frac{1}{N_2 d_1^2} \sum_{i=1}^{N_2} \rho^2(p_{2i}, P_1) \right] \right\}^{\frac{1}{2}},$$

where  $N_i$  is the nuber of points in  $P_i$ , d is the average distance between neighboring points in  $P_i$ , and  $\rho(p_{1i}, P_2)$  is the distance from arbitrary points,  $p_{1i}$ , in  $P_1$  to the nearest point in  $P_2$ . NSD



**Fig. 14.** GASBOR reconstructions of OGG1, a 39 kDa protein critical to the recognition and repair of oxidized guanine in double-stranded DNA by the DNA base-excision repair pathway. (Protein courtesy of Tapas Hazra and Sankar Mitra University of Texas, Galveston, reconstructions for functional analyses in collaboration with Cynthia McMurray.) (*a*) Six independent runs of GASBOR showing the variation among models all of which produce scattering curves in agreement with the experimental data. Each model is composed of dummy atoms whose radii approximate those of amino acids. Data were collected at a concentration of 5 mg/ml with a 30-s exposure. (*b*) The crystal structure of a truncated version of OGG1 fits well within the *ab-initio* SAXS envelope defined by the average of the six GASBOR runs. Each run takes around two hours on computers typically found in most laboratories.

is 0 for identical models. NSD enables quantitative comparison of similarities of models if the modes have the same resolutions, but it is not straightforward to compare similarities of models with different resolutions because  $N_i$  and d are very different. In these cases, using the program SITUS, which was originally developed for EM, is more appropriate (Wriggers *et al.* 1999; Wriggers & Chacon, 2001).

The combination of multiple runs and external constraints has been remarkably successful for the low-resolution envelope reconstruction of many systems (Fig. 15). Imposing correct symmetry greatly enhances the resolution of final results. In general, symmetry information will need to be introduced from external information. However, in the case of the cytosolic portion of a voltage-gated potassium channel (Pioletti *et al.* 2006), the fourfold symmetry could be derived from the scattering using GASBOR, as shown by the result of imposing various symmetries (Fig. 16). Each of the five symmetries shown was run eight times resulting in a total of 40 GASBOR runs. Each run required 12 h of computer time for the 1288 amino-acid complex. In this case, the overall shape could be reconstructed even when fourfold symmetry was not explicitly enforced. Imposed symmetries below and including fourfold-derived shapes with



Fig. 15. Crystallographic and SAXS based models of DNA ligase-PCNA complex. (*a*) A model of human DNA-ligase-PCNA postulated from two independent crystal structures: human PCNA (purple surface PDB id 1w69; Kontopidis *et al.* 2005) and human ligase I (green) complexed with nicked DNA (orange, PDB id 1x9n; Pascal *et al.* 2004). The model assumes the two ring-like structures are co-axial around DNA. B. The averaged *ab-initio* SAXS model derived from experimental SAXS data of the *Sulfolobus solfataricus* ligase-PCNA complex without DNA (purple transparent surface) superposed on the complex reconstructed from the ligase (PDB id 2hiv) and PCNA (PDB id 2hii) models using SASREF (Petoukhov & Svergun, 2005). *Ab-initio* shapes were calculated by GASBOR (Svergun *et al.* 2001a) and average using DAMAVER (Volkov & Svergun, 2003) (after Pascal *et al.* 2006).

similar  $\chi^2$  agreement to the experimental data. Imposing higher symmetries resulted in noticeably higher  $\chi^2$  values and marginally poorer fits to the data. On the other hand symmetries of fourfold and higher had low NSD values, implying a much smaller variation in shapes. An imposed fourfold symmetry optimized both the  $\chi^2$  agreement and NSD values (Fig. 16*f*). We note that the mass was a critical parameter required as input for this study, and the ability to determine overall symmetry by SAXS will likely depend greatly on the overall shape of the target. In this case, the dramatic X-shape of the potassium channel, which is apparent even at extremely low resolutions, was likely critical for determining the proper symmetry.

One of the temptations of *ab-initio* programs is that they require much less information and are easier to run. However, atomic resolution information provides substantially more restraints on possible solutions and is in our opinion the future of SAXS computational development. *Ab-initio* derived models can support models built with or proposed by other methods. Moreover, *ab-initio* structures can be used directly to build atomic models through rigid-body (Section 4.3.2) or flexible (Section 4.3.3) docking into the envelopes.

## 4.3.2 Rigid-body docking into low-resolution envelopes

A candidate *ab-initio* SAXS structure is essentially a low-resolution envelope to which an atomic structure may be docked. Finding a manual best fit using specifically developed SAXS software (Kozin *et al.* 1997; Konarev *et al.* 2001) or with standard molecular graphics programs using envelopes expressed as atomic positions can be a surprisingly challenging process. The program SUPCOMB (Kozin & Svergun, 2001) automatically superimposes atomic structures with dummy atom models; however, this problem is essentially the same as that faced when fitting high-resolution structures into EM maps. Substantial progress has been made in the EM community in this area (Volkmann & Hanein, 1999; Roseman, 2000; Rossmann, 2000; Wriggers & Chacon, 2001; Chacon & Wriggers, 2002; Navaza *et al.* 2002; Craig *et al.* 2006), and the EM program SITUS (Wriggers *et al.* 1999) has also been used for SAXS (Rosenberg *et al.* 2005). In SITUS



**Fig. 16.** SAXS envelopes calculated with different point-group symmetries determined from experimental data from the cytosolic portion of a voltage-gated potassium channel (a tetramer generated from two heterodimers) and overlaid on the crystal structure (Pioletti *et al.* 2006). Each of the envelopes is the result of averaging eight GASBOR (Svergun *et al.* 2001a) runs and is shown in two orientations, 90° apart. Models were calculated with the point group symmetries of (*a*) p1 (no symmetry), (*b*) p2 (one twofold), (*c*) p222 (three perpendicular twofolds), (*d*) p4 (one fourfold), and (*e*) p8 (one eightfold). For each applied symmetry indicated by the value in the circle, the discrepancy between the individual models (NSD; Section 4.3.1) is plotted against the average agreement with the experimental scattering ( $\chi^2$ ; Section 4.1). The model generated with fourfold symmetry has nearly equivalent  $\chi^2$  with lower symmetry models but also has low NSD values. All models suggest a fourfold symmetric structure, excepting the model forced to have p8 point-group symmetry.

the distribution of atoms within the high-resolution structure as well as the low-resolution reconstructions are approximated by a small number of vectors that are calculated by vector quantization. Vector quantization-based fitting is limited to cases in which all density in the SAXS envelope is accounted for by the atomic model. In practice missing or disordered regions of the atomic model need to be modeled before vector quantization-based fitting can be applied (Volkmann & Hanein, 2003). Additionally, more accurate correlation-based approaches have been proposed. In these methods the solution sets lead to the possibility of defining confidence intervals and error margins for the fitting parameters, which is particularly important in the context of docking atomic structures into low-resolution density maps as no independent information is available on how a correct fit should look (Volkmann & Hanein, 2003).

The ability to align crystal structures into SAXS envelopes is particularly informative when there are functionally related conformational changes. For example, both symmetry and the

## 242 C. D. Putnam et al.

existence of atomic resolution structures combined with SAXS analysis have provided substantial information regarding the mechanism by which the AAA+-ATPase p97 undergoes ATP-coupled conformational changes. *Ab-initio* reconstruction of SAXS envelopes for this hexameric ATPase with p3 or p6 point-group symmetry provided substantial information on conformational changes induced upon nucleotide binding (Davies *et al.* 2005). States induced by the non-hydrolysable analog AMP-PNP, the transition-state like mimic ADP-AIF<sub>x</sub>, ADP, and no nucleotide were readily distinguishable in GASBOR-generated envelopes and consistent with previous EM studies (Zhang *et al.* 2000b; Rouiller *et al.* 2002; Beuron *et al.* 2003). Using the crystal structure of the ADP-AIF<sub>x</sub> bound state, different conformations were proposed, revealing a coordinated mechanism for transferring the energy of nucleotide hydrolysis through two parallel rings of domains.

Modeling atomic resolution structures into SAXS envelopes has been used to study the conformational states of the ATPase domain of the *Aquifex aeolicus* enhancer-binding protein NtrC1 in various nucleotide-bound states. NtrC1, which is also an AAA + -ATPase, acts upon a quiescent  $\sigma^{54}$ -RNA polymerase complex to activate transcription initiation. Superposition of the ADP-bound crystal structure of the heptameric ATPase domain with low-resolution envelopes derived from SAXS with enforced sevenfold symmetry could be used to model the nucleotide control of different conformational states. In conjunction with EM, the authors demonstrated an ATP-bound state that stabilized the EBP-<sup>54</sup>-RNA polymerase complex, while subsequent hydrolysis and phosphate release drove the conformational changes necessary to generate an open polymerase/promoter complex (Chen *et al.* 2007).

#### 4.3.3 Flexible docking into low-resolution envelopes

In addition to rigid-body fitting of domains into low-resolution envelopes, *ab-initio* SAXS envelopes, like experimental EM maps, may correspond to states that are different from the crystallized form of the macromolecule. Hence, computational techniques that can take single conformational states and account for potential flexibility during the docking protocols become quite useful. This problem has been faced frequently in EM studies and a variety of computational methods are currently being applied (Suhre *et al.* 2006). Large conformational changes have been shown to frequently correspond to highly collective movements that can be described by a small number of low-frequency normal modes of protein (Section 3.3; Tama & Sanejouand, 2001), and structures modified by NMA (Section 3.3) have been used for fitting EM density maps (Tama *et al.* 2006) allows flexible fitting to EM maps and is well suited for SAXS envelopes; however, the SAXS envelopes generally need to be converted into synthetic EM envelopes with SITUS first (Wriggers *et al.* 1999).

The elastic normal modes computed based on the low-resolution envelopes compare well with the normal modes obtained at atomic resolution (Tama *et al.* 2002; Chacon *et al.* 2003). Thus the motions of large macromolecular assemblies can be directly extracted from low-resolution envelopes derived from SAXS or EM, as has been shown for the DNA-dependent protein kinase and pyruvate dehydrogenase (Boskovic *et al.* 2003; Kong *et al.* 2003). NMA has also been used to study the motion in free and complexed cellulase built from SAXS models (Hammel *et al.* 2004a) when submitted to the ElNémo server (Suhre & Sanejouand, 2004). Computational methods other than NMA could be used to deform molecules to fit into these low-resolution envelopes;

however, NMA has proven to be quite useful due to the efficiency for which altered conformations can be calculated.

# 4.4 Flexibility and conformational disorder measured by SAXS

Several forms of flexibility are crucial for the function of many macromolecular complexes and enzymes (Boehr *et al.* 2006). A common example are flexible linkers, such as those controlling the overall domain conformations and activation of protein kinases (Nagar *et al.* 2006). In another example, analysis of the P(r) function demonstrated that the flexibly linked primase subunits dissociated when ionic strength was increased. This response to changes in ionic strength was linked to primase function (Corn *et al.* 2005). A more dramatic form of flexibility is exemplified by a new class of important regulatory proteins, which do not have a single three-dimensional conformation and are either intrinsically unstructured or natively unfolded (Dyson & Wright, 2005). It has been estimated that over 50% of eukaryotic proteins contain unstructured regions that are over 40 amino acids in length (Vucetic *et al.* 2003), and growing evidence suggests that macromolecular flexibility will be an important part of the regulatory mechanism in many different biological systems.

While samples containing multiple conformations can be challenging to crystallize and tend to reach a single low-energy conformation when they do, they present no difficulties for data collection by SAXS. In combination with domain structures from X-ray crystallography and NMR and recent advances in computational approaches, SAXS has the potential to provide realistic information regarding large-scale structural rearrangements. The techniques described in Sections 4.2 and 4.3 focus on determining a single 'best' conformation. For particularly flexible samples, this conformation is unlikely to exist, and any single conformation of the macromolecule would be predicted to fit the scattering poorly. The real challenges for modeling this flexibility are the dramatic increase in the number of fittable parameters and the difficulties in incorporating multiple models during model refinement. The simplest case of conformational heterogeneity in SAXS is the presence of multiple well-defined conformations (Section 4.4.1), such as are observed during allosteric rearrangements. In contrast, assemblies containing linkers that allow for free, continuous motions of individual domains are more challenging to model (Section 4.4.2) and must be distinguished in SAXS from samples suffering from aggregation (Section 4.4.3).

#### 4.4.1 Multiple well-defined conformations

Assuming that different conformational states do not interact, the resulting scattering from a mixed sample is a population-weighted average of scattering from individual states. This scattering poses a significant challenge for all shape reconstruction techniques. Nevertheless, data determined from the consensus population in solution, under a variety of conditions can be advantageous. A concern with cryo-EM is that an observed minor population overly biases the determined shape. This was dramatically demonstrated by cryo-EM and SAXS DNA bound and free structures of the important DNA damage response protein p53, which is mutated in 50% of all human cancer (Tidow *et al.* 2007).

In general, the characterization of these mixed states by SAXS is most straightforward if the different states can be isolated independently and treated as homogeneous samples. Driving the population to a single conformation might be induced by buffer conditions or through binding specific ligands or substrates. For example, the dodecameric  $Ca^{2+}/calmodulin-dependent protein kinase II (CaMKII) converts from a compact autoinhibited state to a loosely tethered state with independent kinase domains upon <math>Ca^{2+}$  binding (Rosenberg *et al.* 2005).

Alternatively, direct comparisons of different conformational states with theoretical scattering calculated from atomic-resolution structures has been quite successful in identifying and deconvoluting the relative fractions in the sample, such as with the archaeal secretion ATPase GspE (Fig. 17). The structure was solved as a mixed hexamer of open and closed conformational states of the component monomers (Yamagata & Tainer, 2007). Fitting of the solution X-ray scattering curves of the Mg<sup>2+</sup> and AMP-PNP bound enzyme fit the experimental crystal structure less well than a computational hexamer generated from the closed state of the monomers alone. Moreover, the scattering of the ADP-bound state was well fit by a mixture of scattering from the crystal structure, the all-open model, and the all-closed model, demonstrating considerable flexibility in the system only using experimentally determined states of the monomer for modeling.

Identifying that a solution contains a mixed population of macromolecules can be challenging and often requires additional information from other techniques such as native gel electrophoresis. Since  $R_G^2$  corresponds to the average square distance of each scatterer from the center of the particle contributing to the scattering, mixed samples containing components with different  $R_{\rm G}$  values will yield observed  $R_{\rm G}$ 's that are the square root of the weighted sum of the  $R_{\rm G}^2$ values of the different components. Thus, mixed populations continue to have linear Guinier regions (Heller, 2005). Moreover, the signal indicating heterogeneity is highly dependent upon how much the conformational changes alter SAXS (Heller, 2005). In the calculation of the collapse of calmodulin, for example, the 2.6 Å difference in  $R_{\rm G}$  and 20 Å differences in  $D_{\rm max}$ were readily apparent, even when simulated noise was added to the scattering curves. Ab-initio reconstructions of mixed populations show substantial conformational components of both extended and collapsed states. In contrast, the conformational changes involving protein kinase A, with a change in  $R_{\rm G}$  of 0.13 Å, were lost when noise was added to the theoretical scattering (Heller, 2005). Therefore, SAXS studies of mixed populations that give rise to large changes in SAXS will be the most straightforward. For example, scattering power in the small-angle region goes by the square of the mass; hence, mixed populations of different multimeric states will be more readily identified. Similarly, unfolding of macromolecules involves large-scale changes in the overall structure and will dramatically change the observed scattering; thus, SAXS has been a method of choice for studying protein folding (Provencher & Glockner, 1983; Doniach, 2001; Perez et al. 2001).

The best way to identify heterogeneity is by following it through titration of the states from one form to another or through situations in which atomic models are available so that they can be directly compared to the observed scattering. For example, SAXS has been used to follow large-scale changes due the pH-induced maturation of viral capsids of the HK97 bacteriophage and *Nudaurelia capensis*  $\omega$  virus, which give rise to substantial changes in the SAXS curve (Canady *et al.* 2001; Lee *et al.* 2004). Moreover, several schemes have been used to prove the existence of transient, substrate-induced conformational changes and characterize them (Akiyama *et al.* 2004; Goettig *et al.* 2005; Graille *et al.* 2005; Vestergaard *et al.* 2005; Nowak *et al.* 2006a, b).

Using experimentally determined SAXS and theoretical scattering from individual components (form factors), volume fractions in each conformation can be determined by solving a systems of linear equations with the program OLIGOMER (Konarev *et al.* 2003). The bacterial class I



**Fig. 17.** Determination of the solution conformation of the hexameric archaeal secretion ATPase GspE from a combination of SAXS and crystallography. (*a*) Side view of the 2 alternating configurations of the ATPase GspE monomer bound to AMPPNP found in the hexameric structure determined via crystallography. (*b*) Top views of the crystal structure and two models proposed by modifying subunits to adopt either all open (brown) or all closed (green) conformations. (*c*) In solution the ATPase in excess AMPPNP adopts a conformation most similar to the all closed model while a solution with excess ADP is best described as a mixture of all the models (after Yamagata & Tainer, 2007).

release factors, for example, adopt a compact structure in the crystal, but unlike the crystal, the solution scattering is consistent with a population containing 92.5% in an open conformation and only 7.5% in the compact form (Vestergaard *et al.* 2005). Similarly, the transcriptional antiterminator LicT exhibits a heterogeneous population in solution with 61% being open and 39% being compact and active (Graille *et al.* 2005).

# 246 C. D. Putnam et al.

In addition to systems of linear equations, single-value decomposition (SVD; Press *et al.* 1992) can deconvolute SAXS data from mixtures. SVD was introduced into SAXS in the early 1980s (Fowler *et al.* 1983) and has been applied to the problem of protein folding (Chen *et al.* 1996; Perez *et al.* 2001) as well as deconvoluting mixtures of protein–RNA complexes (Bilgin *et al.* 1998) and transient protein conformations (Fetler *et al.* 1995). SVD requires multiple scattering curves collected from samples containing different populations of states. Unlike solving systems of linear equations, SVD does not require scattering curves from individual components, but in order to extract real scattering curves external information may be required, such as thermodynamic models for transitions.

In SVD, all the collected scattering profiles  $I_n(q)$  are reduced to a common minimal basis set as described in the following equation:

$$I_n(q) = \sum_j^N w_j b_j^n u_j,$$

where  $w_j b_j^n$  is the weighting contribution of  $\mathbf{u}_j$  basis vector to  $I_n(q)$ , the *n*th experimentally determined scattering profile. Determining the values of basis vectors and the relative weights is accomplished through creating an  $M \times N$  matrix  $\mathbf{A}(M \times N)$  where the N columns are the intensity values of the scattering profiles determined at M values of q. Such a matrix may be represented as  $\mathbf{A}(M \times N) = \mathbf{U}(M \times N)$   $\mathbf{W}(N \times N)$   $\mathbf{B}(N \times N)$  where U is also an  $M \times N$  matrix containing the  $\mathbf{u}_j$  basis vectors, W is a diagonal  $N \times N$  matrix composed of the so-called singular values  $w_j$  and B is an  $N \times N$  matrix containing  $b_j^n$ . Although U contains N vectors, only important basis vectors will have associated significant  $w_j$  values. Many vectors in U will fit noise in the data rather than the significant parts of scattering profiles. Several commercially available mathematical packages have built-in SVD routines.

SVD will identify the minimum number of curves required to describe all the scattering profiles and can readily distinguish a system with a single transition from one with multiple transitions. Only two basis vectors were necessary to describe the scattering datasets collected from the allosteric enzyme aspartate transcarbamoylase measured in the presence of various substrate analogs, activators, and inhibitors (Fetler *et al.* 1995).

In an illustrative use of SVD, the temperature denaturation of neocarzinostatin was characterized by SAXS (Perez *et al.* 2001). At least three SVD basis vectors were required for a complete description of the unfolding, suggesting the presence of at least one intermediate state. With the scattering from the starting protein and the final unfolded state, they attempted to derive the scattering from the intermediate state by reprojecting the basis set. However, the derived scattering curve was degenerate, and the authors concluded that the folding pathway of neocarzinostatin involves an ensemble of related flexible intermediate states.

Another promising application of SVD is toward characterizing detergent solubilized membrane proteins (Lipfert & Doniach, 2007). The experimental matrix for SVD used SAXS data collected from different concentration ratios of membrane protein to detergent. All concentrations of detergents were above the critical micelle concentration. They reprojected the SVD vectors as scattering due to membrane protein–detergent complex, micelles, a micelle–micelle interaction component and a membrane protein–detergent to micelle interaction component. Of course the component of interest is the membrane protein–detergent complex. Although their experimental application using membrane protein TM0026 did not provide details on structural parameters other than  $R_{\rm G}$ , the potential to isolate the scattering

profile of a membrane protein-detergent complex is very exciting and well worth further research effort.

SVD is potentially very powerful; however, some care must be taken in employing the technique. Robust analysis requires many scattering curves and relatively error-free data. Systematic errors from background subtraction among the data sets may cause particularly insidious problems for SVD, as some of the basis sets may be required to fit error rather than be truly representative of solutions. If background-subtracted data is used in the SVD analysis, great care must be taken in titration experiments to guarantee the buffer is properly matched to the solution for background subtraction. SVD is also particularly powerful for time-resolved experiments of induced conformational changes. Unfortunately, time-resolved experiments with short exposures also suffer from weaker signals. Finally, SVD identifies the minimum number of scattering curves and not necessarily all states that may exist (Koch *et al.* 2003), which might be particularly problematic when a continuum of states exist, such as with partially unfolded peptides.

### 4.4.2 Conformational disorder explored by SAXS

In contrast to samples that can be described as containing multiple well-defined conformations, many multidomain proteins and protein complexes contain flexible linkers that allow them to adopt large numbers of conformations. This situation is substantially more complex to model; however, it is likely to be quite common for large numbers of eukaryotic proteins. For these types of samples, attempting shape reconstructions to derive a single model with a 'best-fit' conformation can be misleading and at best provides a model representing an average of the conformations. At times an averaged model can be informative. For example, SAXS studies of the c-Abl tyrosine kinase that is dysregulated by gene fusions in chronic myelogenous leukemia (Nagar et al. 2006) have provided additional insight into the autoinhibition of the kinase. The N-terminal half of Abl is comprised of three domains connected by flexible linkers. Binding of the myristoylated N-terminus within a binding pocket on the protein generates a compact autoinhibitated state. SAXS data revealed that mutants predicted to disrupt the autoinhibited state were much more elongated than wild-type Abl that closely matched the more compacted crystal structure. Fully extended models, created from the crystal structure (PDB id 10pl; Nagar et al. 2003), fit the averaged SAXS envelope well, despite the fact that the molecule was proposed to be in a conformational ensemble with a wide range of heterogeneous states. Further interpretation about the extent of flexibility would require analysis beyond *ab-initio* shape restoration.

In some cases, the lack of convergence of *ab-initio* models has been correlated with flexibility. A number of models calculated by CREDO from scattering data collected from the cellulase Cel48F did not generate a single conformation (Fig. 18; Hammel *et al.* 2004a). Combining and weighting the scattering of individual models with OLIGOMER improved the overall fit (Fig. 18). In this case, more parameters were added which makes an improvement in the fit unsurprising; however, it is noteworthy that the individual models were generated independently of each other and from the weighting and merging steps. What is truly remarkable, however, is the fact that CREDO, a program that generates an over-parameterized *ab-initio* model (Section 4.3.1), was unable to come up with a better single model fit to the raw data. In the case of scattering from a heterogeneous population, the measured scattering is derived from the population-weighted thermodynamic ensemble and describes some population-weighted distribution



Fig. 18. Partial *ab-initio* models of free and complexed cellulase Cel48F. (*a*) Five typical CREDO (Petoukhov *et al.* 2002) models of linker–dockerin region of free Cel48F are displayed in different colors together superposed on the crystal structure of Cel48F catalytic domain. (*b*) Two restored models (green and blue) of the dockerin/cohesin complex of the complexed Cel48F using the program CREDO superposed on the crystal structure of Cel48F (PDB id 1fbw; Parsiegla *et al.* 2000). The corresponding sub-domains are schematically represented below each construct. The observed displacements between individual runs of partial *ab-initio* restoration are highlighted with the orange arrows. (*i*) Experimental SAXS profiles of free (bottom curve) and complexed (upper curve) fitted by the averaged form factors of the CREDO models obtained by the program OLIGOMER (red line; Konarev *et al.* 2003), and SAXS profiles calculated from the single CREDO models (green line) (after Hammel *et al.* 2004a).

of electrons. The inability of CREDO to generate and converge to a 'best fit' model likely derives from incompatibilities between the constraints that CREDO has in generating models and those model features, such as partial occupancy, necessary to completely describe this population-weighted electron density.

The biggest challenge in trying to model conformationally very flexible systems using SAXS data is to avoid overfitting the raw data. One strategy to avoid overfitting the raw data with multiple models is to leverage existing atomic structures to reduce the parameter space of the model by describing the ensemble as a set of most probable structures. To minimize potential problems with overfitting, individual conformations to be tested as probably components of the population ought to be generated independently of the SAXS data. A number

of creative techniques have begun to address the problem of quantitatively modeling flexible macromolecules observed in SAXS experiments (Akiyama *et al.* 2004). Various modeling approaches can be used to generate atomic models that sample conformational space for use in fitting experimental SAXS curves. Monte Carlo techniques (Buey *et al.* 2007; Shell *et al.* 2007), exploring the dihedral space of linkers (Tai, 2004), CONCOORD, a non-dynamical method of generating conformation sets (Schlick, 2001) and MD (Levy & Becker, 2002) have all been employed.

Conventional MD methods are computationally intensive; however, several advances have increased the size of tractable conformational changes (Yuzawa et al. 2001). These new techniques include multiple time-step MD and high temperature MD (Boehm et al. 1999; Aslam & Perkins, 2001; Yuzawa et al. 2001; Aslam et al. 2003; Hammel et al. 2005; von Ossowski et al. 2005; Gilbert et al. 2006) in which simulations are run at very high temperatures ( $\sim 1000$  K) to prevent molecules from becoming trapped in local minima (Leach, 2001). Similarly, many of the simulations are sped up by including only van der Waals terms and distance restraints, but not electrostatic terms (Losonczi et al. 1999). Comparison of rigidbody modeling, ab-initio shape reconstruction, and reconstruction of missing domains using CREDO have given similar results to MD sampling (Marino et al. 2006). SAXS profiles calculated using MD trajectories are particularly useful for determining the average overall shape of the minicellulosomes and for proposing plausible atomic conformations they may adopt (Fig. 19). Given the restricted amount of independent data values from SAXS experiments, the best these techniques can provide is a set of conformations that are consistent with experimental data. In the absence of additional experimental evidence, the combination of SAXS and atomic models generated by MD trajectories cannot prove the existence of any particular conformation.

Once generated from atomic structures, these models can then be evaluated for their relative contribution to the scattering using SVD or the algorithm underlying OLIGOMER (Section 4.4.1). In the case of  $\beta_2$ -glycoprotein I ( $\beta_2$ GPI), SAXS fits poorly with the crystallized conformation or individual models with systematic modified conformations (Hammel *et al.* 2002) but does fit well using a multiple model approach (Fig. 20). Not only is the inherent flexibility in  $\beta_2$ GPI likely to be important to its function, but it also illustrates the common case in which external information must be introduced in order to generate the correct types of models to fit the observed scattering.

We are enthusiastic about a new approach to analyze the presence of multiple conformations of proteins contributing to the experimental scattering profile (Bernado *et al.* 2007). Bernado and co-workers define an ensemble optimization method (EOM) in which a pool of possible conformations (N>1000) is randomly generated to cover the possible conformational space. A genetic algorithm is then applied to select subsets (N=50) of configurations that fit the experimental scattering. The advantage of this method is the use of quantitative criteria for analyzing the EOM-selected models and for determining the optimal number of conformers in the subset. The best subsets are then selected for further evolution. Using both theoretical and experimental data for unstructured and multidomain proteins, EOM was able to distinguish between rigid and flexible proteins and assess interdomain contacts.

We also are enthusiastic about the potential offered by the recently developed residual dipolar coupling (RDC) NMR, which has been used to identify relative orientations in multidomain proteins relative to an external coordinate system (the 'alignment tensor'). For multidomain proteins, the orientation of each domain can be determined separately within this coordinate



Fig. 19. The solution structure of the cellulosome. (a) The average envelope shape of different cellulosome constructs were calculated with GASBOR (Svergun *et al.* 2001a). The corresponding modules are schematically represented on the top of each shape. (b) Restored partial *ab-initio* models of the different cellulosomes constructs calculated with CREDO (Petoukhov *et al.* 2002). The CREDO models are displayed in surface representation (yellow). The fixed known atomic structures are in  $C_{\alpha}$  tube representation. The secondary structural elements of superimposed atomic structures are shown as  $C_{\alpha}$  tubes. (c) Best-fit models were calculated by MD of different constructs and superimposed on cohesin from S4 or complexed Cel48F (highlighted with the black circle). Three models colored by blue, brown, green for each model, except for FtS4Fc complex where only two models (green and blue) for better visualization, are presented (after Hammel *et al.* 2005).



Fig. 20. Solution structure of  $\beta_2$ -glycoprotein I ( $\beta_2$ GPI). (a) The crystal structure of  $\beta_2$ GPI (left) is shown with attached sugars (red). (b) The best single-conformation model derived from experimental SAXS data is superposed with the averaged *ab-initio* model calculated by program DAMMIN (Svergun, 1999) displayed as gray cages. (c) The theoretical scattering profile calculated for the mixture of different conformations improved the fit to the experimental data (root mean squared deviation for single and multiconformation fit are  $3.0 \times 10^{-3}$  and  $8.5 \times 10^{-3}$ , respectively). Three different atomic conformations were constructed by simple rotations between domains CCP2 and CCP3 of the best fit single structure (after Hammel *et al.* 2002).

frame, and the relative interdomain orientation can be deduced (Bernado *et al.* 2005). RDC data allow for unbiased determination of interdomain orientations in solution, albeit with a fourfold degeneracy, but require structural models for interpretation. In the study of the first two Ig domains of titin, Z1Z2, SAXS data was used to resolve the RDC degeneracy (Marino *et al.* 2006). Calculations of the 200 RDC conformers showed that the 50 models with the lowest-discrepancy values were a well-defined cluster of conformations that superimposed with the *ab-initio* SAXS model, whereas the more compact crystal structure failed to fit the solution structures. Furthermore, conformational sampling using simulated averaged RDCs was applied in the analysis of SAXS profiles of partially unfolded protein. The close agreement of experimental and simulated RDC to SAXS data validated the conformational sampling results and provided a description of local structure, dynamics and average dimensions of the ensemble of unfolded protein. Thus, the synergy of SAXS with molecular modeling, crystallography, and NMR promises to provide unique insights into the structural characterization of proteins with intrinsic flexibility (Mattinen *et al.* 2002; Bax & Grishaev, 2005).

### 4.4.3 Flexibility, oligomerization, or aggregation?

In general multidomain proteins with long linkers or that adopt extended conformations have smooth scattering profiles with few prominent features at high resolution and extended tails in P(r) functions. These proteins also typically have heterogeneous conformations with a variety of  $R_{\rm G}$  and  $D_{\rm max}$  values. Guinier plots of these macromolecules are linear over a smaller region:  $qR_{\rm G} < 0.8$  instead of  $qR_{\rm G} < 1.3$  which is more typical for globular samples



**Fig. 21.** The experimental P(r) of the rabbit 15-lipoxyganase-1 (black). Theoretical P(r) calculated for a mixture of two protein conformations adopting different extensions in the N-terminal regions with the indicated occupancies (red) and for a mixture of 80% monomeric and 20% oligomeric assemblies (blue) are shown. The crystal structures used in the P(r) calculation (PDB id 1lox; Gillmor *et al.* 1997) using in the P(r) calculation are shown as surfaces using the same coloring (after Hammel *et al.* 2004b).

(Table 1). Unfortunately, many of these features are also observed in the presence of small amounts of aggregation, which primarily affects the same low-resolutionregion of the scattering curve.

Studies on the soybean lipoxygenase-1 and rabbit 15-lipoxygenase-1 illustrate the need for careful interpretation of SAXS data in cases where flexibility is proposed (Fig. 21). The major discrepancies between the experimental scattering curve of rabbit lipoxygenase in solution and the curve calculated from the atomic coordinates were interpreted in terms of a large movement of the N-terminal domain with respect to the C-terminal domain (Hammel *et al.* 2004b). Rigid-body modeling was applied, and the improvement of the fit in the entire scattering profile was observed using a mixture of different swing out conformations (Fig. 21, red, two conformations are shown; Hammel *et al.* 2004b). The modeling of rabbit lipoxygenase was entirely dependent on establishing that the solution studied was monodisperse. In a study of soybean lipoxygenase, Dainese and co-workers found that the modeled structure was quite similar to the crystal structure and that slightly aggregated samples would give rise to elongated signals, as reported for rabbit lipoxygenase (Dainese *et al.* 2005). Re-evaluation of rabbit lipoxygenase indicated that partial (20%) aggregation could also explain the scattering of rabbit lipoxygenase (Fig. 21).

Therefore, samples that are suspected of possessing intrinsic flexibility must be carefully characterized to ensure monodispersity prior to any SAXS modeling. Native gels can be very useful as an indicator of the homogeneity of samples. We have also found dynamic light scattering (DLS) useful for providing overall size and sample polydispersity measurements under the same concentrations and buffer conditions as for SAXS experiments. We have used DLS as an important pre-screening tool to optimize samples and buffers for maximum monodispersity. Our

empirically based approaches to dealing with intrinsic flexibility, heterogeneous conformations, aggregation, and multidomain proteins are presented in Section 5 below.

# 5. Strategy and tactics for SAXS experiments

We designed and built the SIBYLS synchrotron beamline (http://www.bl1231.als.lbl.gov) at the Advanced Light Source to interconvert between a SAXS and a crystallography endstation quickly (under an hour). One of our goals in the stewardship of this beamline has been to encourage the larger community of crystallographers to use SAXS. Experience has demonstrated several pitfalls in this process. Typically crystallographers are excited about the initial results from *ab-initio* shape restorations only to later become disillusioned because of perceived uncertainty. In particular, some crystallographers feel that they have biased their results in a certain way. As mentioned throughout this article, in contrast to crystallography, SAXS suffers from few standardized ways of validating results. In crystallography, visibly attractive but poorly ordered crystals will immediately result in poor images in the diffraction experiment and thereby indicate data collection and processing will not be fruitful. In contrast, every solution sample gives scattering that could be processed and modeled whether or not the results are valid. Both SAXS and crystallographic analysis packages are becoming more and more like 'black boxes' and are frequently designed using ideal data giving few indicators about problematic data. Without a clear understanding of what information SAXS can and cannot resolve and of the metrics for judging results and the steps along the way there is a real danger of generating incorrect results from problematic data.

With increased general use of SAXS as a powerful tool to examine biological samples, it becomes necessary to have both strategic concepts and tactical plans for the assessment, processing, and interpretation of the SAXS results collected from every sample regardless of quality. Here we provide a strategic basis to proceed with data evaluation and processing for different types of data, where data-quality assessments play an essential role (Fig. 22). It is difficult to overstress the paramount importance of collecting high-quality SAXS data, validating the level of quality by several tests, and remaining diligent in regard to possible systematic errors and other possible data problems.

Based upon our experience with many types of samples, we recommend the strategy and tactics shown schematically in Figs 22–25. Conceptually the SAXS experiment can be divided into four major steps: data collection, data evaluation (Fig. 23), data analyses (Fig. 24), and solution structure modeling (Fig. 25), each of which is detailed below.

#### 5.1 Data collection

At the SIBYLS beamline, 15  $\mu$ l of a well-behaved 100 kDa protein in typical buffers can provide SAXS data with noise levels below 1% out to q=0.3 Å<sup>-1</sup> at a wavelength of 1 Å in a 30-s exposure. As described below collecting data as a function of concentration is an important quality control in SAXS analysis and four concentrations between 10 and 1 mg/ml at the same volume are recommended. A minimum amount of sample for the same protein for interpretable data under the same conditions is 15  $\mu$ l at 1 mg/ml. Typical data collection times are 1–100 s although millisecond time-resolved experiments are possible.

A number of parameters affect the amount of sample required for data collection. The difference in average electron density between the sample and bulk solvent plays a key role. For example, RNA and DNA, which have greater scattering contrasts than protein, require lower



Fig. 22. Schematic for SAXS data collection, evaluation, analysis, modeling, and interpretation. We favor the combination of atomic structures with SAXS envelope modeling (lower right) where possible. Using the SAXS experimental observations identifies possible flexibility along with shape, assembly and conformation in solution.

sample concentrations for equivalent signals (Fig. 26). Similarly, osmolites in buffers can decrease the scattering contrast with macromolecules. Salt concentrations, particularly above 1 M NaCl, will require higher sample concentrations than low salt buffers for equivalent signals. Detergents, which are commonly used for stabilizing and purifying hydrophobic molecules, can also be problematic, as detergent micelles scatter very well. Collecting data at detergent concentrations below the critical micelle concentration is required for most standard analysis. Similarly,



Fig. 23. Data collection and evaluation.

macromolecular size is also very important. The total scattering from the same mg/ml concentrations of a small macromolecule is equivalent to that of a larger macromolecule; however, the angular distribution of scattering is not the same. At the same mg/ml concentration, larger macromolecules have stronger scattering in smaller *q*-ranges than smaller macromolecules. If small angle information is desired from larger macromolecules for determination of mass,  $R_G$ ,  $D_{max}$ , and aggregation, lower sample concentrations will be required. The amount of sample required also varies with incident beam size, wavelength and whether static or flow cells are used.

# 256 C. D. Putnam et al.

To obtain an ideal scattering curve for the entire q-range,





o Infinite dilution

Highest protein concentration

Fig. 24. Data extrapolation, merging, and analysis. Proper data analysis as outlined is essential to any further modeling and interpretation.

Many SAXS instruments can adjust the q-range over which data may be collected in one single exposure. This may be done by adjusting the sample to detector distance, the incident wavelength, or the offset of the detector relative to the direct beam. Changing the q-range can require a prohibitive amount of time and one stable and well-calibrated configuration is often required.

Properly choosing beamline geometries to collect the appropriate scattering information is important. The first consideration is to capture the q-range necessary to accurately determine



Fig. 25. Solution structure modeling and interpretation.

values for both  $R_{\rm G}$  and  $D_{\rm max}$  while obtaining sufficient information for being able to use the data for structural modeling purposes (Fig. 23). In theory the minimum q necessary for determining  $R_{\rm G}$  is  $q_{\rm min} \leq \pi/D_{\rm max}$ ; however, for determining  $R_{\rm G}$  from Guinier plots (Section 2.3.2), multiple points are required and the linear region ( $q_{\rm min}R_{\rm G} < 1.3$  for globular



**Fig. 26.** SAXS from RNA samples. (*a*) Experimental scattering curves of SAM-riboswitch at 1 mg/ml (red) and a 95-bp RNA molecule at 0.7 mg/ml (courtesy of Robert Rambo) in comparison to lysozyme 2.5 mg/ml (gray). For better comparison the curves have been normalized by concentration and molecular weight. (*b*) The calculated *P*(*r*) functions for the samples shown in panel (*a*) are shown with the maximum value normalized to 1. (*c*) Average *ab-initio* model of the SAM-riboswitch (red) superposed on the theoretical dimer of the crystal structure (PDB id 2gis; Montange & Batey, 2006) (green and cyan). Average *ab-initio* model of the 95-bp RNA (blue wireframe) superposed on single DAMMIN model of (blue beads). The beads models reconstructed with DAMMIN (Svergun, 1999) were transformed into wireframe model using SITUS (Wriggers *et al.* 1999).

samples and  $q_{\min}R_G < 0.8$  for elongated samples) can be more constrictive than the theoretical constraints. For large complexes, getting sufficiently low q data by increasing the sample-to-detector distance or decreasing the incident X-ray wavelength may be required. For example, the small well-folded maltose-binding protein with  $R_G = 22.1$  Å and  $D_{\max} = 52.0$  Å, requires a  $q_{\min}$  of < 0.059 Å<sup>-1</sup>, which is readily achievable in most default SAXS station geometries, whereas the *E. coli* MutS tetramer has a  $R_G = 80.8$  Å and  $D_{\max} = 250.0$  Å and requires a minimum q of < 0.013 Å<sup>-1</sup> (Mendillo *et al.* 2007), which may require more careful experimental setup. In contrast, in order to sufficiently constrain  $D_{\max}$ , a maximum q of  $q_{\max} \ge 2\pi/D_{\max}$  is required. For large complexes, such as the MutS tetramer, this required limit  $q_{\max} \ge 0.025$  Å<sup>-1</sup> is typically not a problem for normal geometries; however, for small proteins this limit should be checked.

Larger q values inherently provide more information and give structural reconstruction algorithms more information to fit which helps constrain the resulting solutions (Fig. 23). Unfortunately, since scattering intensity falls off rapidly, higher concentrations (which may not be possible) and/or longer exposures may be required for adequate signal to noise in the high q region. While high q data can be very valuable, especially for particle reconstruction, care should be taken so as not to sacrifice accurate low q data for extremely noisy high q data, especially in the case of X-ray sensitive samples.

Much like UV spectroscopy, the scattering from the buffer must be subtracted from the sample. There are several technical challenges in the subtraction for SAXS. First, the signal in SAXS data lies very close (typically within a few degrees) of the incident beam. In order to measure the data, the sample to detector distance is typically on the order of a meter, and the incident X-ray beam must be tightly collimated and focused and carefully blocked, as it typically has 10<sup>10</sup>-fold higher intensity than the scattered X-rays. Stray rays of the primary beam can dominate the scattering. Often some portion of these stray rays are present during data collection but can be subtracted with the buffer. Second, small buffer differences, even a micromolar difference of salt concentration, can cause large differences relative to signal at high resolutions. Thus, the sample is typically exchanged into the buffer that is used as a blank prior to the experiment by dialysis or size-exclusion chromatography. Third, to avoid systematic errors due to small differences in X-ray path lengths, the samples and blanks are measured in the same cuvette, which has windows made of a material that scatters X-rays poorly, such as mica, kapton, or beryllium. Fourth, the fact that the identical cuvette is used for the sample and blank prevents them from being measured simultaneously. Even at relatively stable synchrotron X-ray sources, changes in the intensity of the X-ray beam can easily be larger than 10<sup>-3</sup> or 10<sup>-4</sup> over the timeframes in which samples are measured and the cuvette is washed. Thus, the resulting scattering must be normalized by the intensity of the incident X-rays. Fifth, correction for the difference in X-ray absorption by the sample and the buffer can be applied, although the correction is small and is commonly ignored.

One of the first quality checks that must be performed during data collection is to determine the sensitivity of the sample to X-ray irradiation damage. Changes between SAXS profiles collected by multiple short exposures can indicate radiation sensitivity. Additionally, samples with radiation damage tend to aggregate, showing increasing  $R_G$  and I(0) as a function of total X-ray exposure. We also routinely check samples before and after longer exposures even for samples that pass initial screening of radiation sensitivity. Several options exist to deal with samples that have inherent radiation sensitivity. Cooling the sample, diluting the sample, and adding free radical scavenging compounds or protectants like glycerol have helped prevent aggregation (Kuwamoto *et al.* 2004). Flow cells can also significantly improve data collection for very sensitive samples, although flow cells can require much more material. Alternatively, data from multiple short exposures can be summed where each exposure is performed on a fresh sample.

For samples without radiation damage, individual frames can be averaged. Background/buffer scattering should be measured both before and after the measurement of the sample. Scattering due to macromolecules is then calculated by subtracting the buffer blanks from the sample. The resulting scattering curve calculated using buffer blanks measured both before and after the sample should be identical. Data collected from several types of SAXS set-ups require 'de-smearing' due to incident beam properties. This is not required for sources with point focusing at the detector.

# 5.2 Data evaluation

One of the attractive features about SAXS data collection is that data evaluation can be performed during data collection. This is particularly useful with limited time and/or material.



**Fig. 27.** Characteristic scattering of aggregates in SAXS. A fully aggregated sample is shown in black. This aggregate has no features in the scattering curve and indicates a poorly behaved sample. Lack of features in the scattering curve can also be observed with unfolded samples. Unfolded and aggregated samples can be distinguished using a Kratky plot. A partially aggregated sample is shown in red. In this scattering curve, only the lowest resolution scattering is affected, and this type of scatter can be observed both through the low-resolution shape of the scattering and disagreement between I(0) and  $R_G$  calculated from the Guinier plot and from the P(t) function by indirect Fourier transformation. Passage of the sample through a filter with a 100 kDa cut-off removes the aggregated material and allows an aggregation-free scattering curve to be collected (green line).

For example if samples shows signs of aggregation at an intermediate concentration, higher concentration measurements are unlikely to be fruitful (Fig. 23). A better use of the available material might be to dilute the concentrated samples into different buffers to screen for conditions that give better scattering.

Often the first parameter extracted from data evaluation is  $R_G$  (Table 1), which can be determined from the slope in a Guinier plot  $[\log(I(q)) vs. q^2]$  (Fig. 23). A Guinier plot can be easily generated from the raw data using the program PRIMUS. Nonlinear behavior in the Guinier plot in the range  $q < 1.3/R_G$  indicates the presence of aggregation. The scattering from the aggregation influences the entire dataset and any further data processing should proceed with caution. Some samples may show nonlinearity over a small region within the Guinier region and the remaining data is linear. Data clipped at the lowest q values where aggregation is most apparent may be processed further. However, when possible, varying buffer conditions, centrifugation and filtration should be attempted to remove the aggregation for further analysis as the aggregation may have subtle effects throughout the scattering profile (Fig. 27).

With ideal samples, scattering profiles from a concentration gradient should be superimposable when scaled by concentration. In this case individual scattering particles are not interacting with one another (Fig. 23). In some cases target macromolecules interact with one another either repulsively or attractively adding additional and unwanted correlations in solution. These correlations affect the scattering profile, usually in the lowest resolution region  $(q < 0.1 \text{ Å}^{-1})$ . For example, decreasing intensity at very small q with increasing protein concentration indicates the presence of repulsion forces. In these cases, the observed scattering is treated as a product of the scattering curve of the particle in ideal solution (the 'form factor' of the particle) and interactions between particles (the 'structure factor' of the solution). An interference-free scattering due to the 'form factor' can then be extracted by measuring at least three different concentrations. Lowering the concentration increases the average distance between particles and decreases the strength and effects of interparticle interactions. Thus, these curves can be used to extrapolate the scattering to infinite dilution. Long-range interactions can also be removed or diminished experimentally by changing the buffer, such as by screening electrostatic interactions by increasing salt concentration.

# 5.3 Data analysis

For SAXS data that is free of aggregation, radiation damage, and long-range interactions, data analysis may proceed (Fig. 24). Determination of the experimental parameters  $R_G$ , I(0),  $D_{\rm max}$ , excluded volume, and molecular weight is the first step in reconstruction of the solution structure (Table 1).  $R_G$  and I(0) can be determined from the SAXS curve using the Guinier plot or through calculation of the pair-distribution function, P(r) (Section 2.3.3; Table 1, Fig. 24). Disagreement between the values may be a sign of improper assignment of  $D_{\rm max}$  for the indirect Fourier transformation or other problems such as heterogeneity or unfolding.

Extended and globular macromolecules can be distinguished by P(r), as globular particles have bell-shaped functions, whereas extended particles have functions with a maximum at short distances and a long extended tail (Figs 5 and 10). Unfolded samples not only have P(r) functions consistent with extended molecules, they also possess characteristic Kratky plots  $[I(q)q^2 vs. q]$ lacking bell-shaped peaks and having a plateau or a slowly increasing curve at large q values (Section 2.3.2, Fig. 24). For proteins, if  $R_G$  is substantially larger than the theoretical value for a globular protein of the same molecular weight then the protein is likely in an extended conformation or potentially an oligomer (Fig. 24). The theoretical  $R_G$  can be calculated by:

$$R_{G, \text{ globular protein}} = (3/5)^{\frac{1}{2}} [\text{MW}(\text{Daltons})/(0.44 \text{ Daltons}/\text{Å}^3) * (3/4\pi)]^{\frac{1}{3}}.$$

Determining the oligomeric state requires a determination of molecular weight, such as methods relying upon I(0) (Section 2.3.5). Most often the monomeric molecular weight is known and non-integral stoichiometries from I(0) imply heterogeneous multimerization. The excluded volume calculated through the Porod invariant can be converted to a mass (Table 1; Section 2.3.5). For proteins, a rule of thumb is to estimate the mass by dividing the excluded volume derived this way by 2. This accounts for the hydration layer as well as the protein volume. This estimate holds up well for large globular proteins (>70 kDa) but fails for proteins with unusual shapes and is particularly problematic for small proteins.

## 5.4 Structure modeling and interpretation

If a solution of folded macromolecules is monodisperse, then the measured scattering profiles can be used for solution structure determination (Fig. 25). *Ab-initio* shape determination can be applied to reconstruct the low-resolution envelopes of the protein if atomic structures are not available (Section 4.3). For this purpose programs DAMMIN, DALAI\_GA, SAXS3D, or

### 262 C. D. Putnam et al.

GA\_STRUCT can be used. If the number of residues of the protein is known, the program GASBOR has the advantages of providing significantly more sophisticated penalties that constrain the models and of not prohibiting the generation of cavities and other anisotropic shapes (Svergun *et al.* 2001b). Comparison of multiple reconstructions is extremely important to verify the stability of the solution. Multiple repetition of the modeling process significantly decreases the risk of inferring erroneous shapes. The program DAMMAVER aligns, averages, checks the uniformity, and computes a probability map of the given *ab-initio* models.

Ab-initio shape restoration programs allow further constraint on the solution by enforcing known symmetry. Ab-initio shape-determining programs are known to provide accurate solutions given the correct symmetry (Table 4, Fig. 25). In fact, correctly enforced higher symmetry improves the resolution of final models. Unfortunately given incorrect symmetries *ab-initio* programs will often find shapes that also fit the scattering profiles. Assuming higher symmetry with knowledge of only the molecular weight must be carefully justified. Identification of the fourfold symmetry of the potassium channel by comparing NSD and  $\chi^2$  values derived from the SAXS data alone (Fig. 16) may not be generally applicable, especially for less anisotropic assemblies.

If the atomic structure of the sample is known or an atomic model has been proposed, comparison of theoretical SAXS profile with the experimental data is the first step in the structure evaluation (Fig. 25). The theoretical SAXS profile can be calculated with the program CRYSOL, using the  $\chi^2$  agreement to evaluate the best models. Agreement between theoretical SAXS curves from crystal structures and experimental SAXS data of  $\chi^2 < 3.0$  are not uncommon. *Ab-initio* shapes and atomic resolution models may be superimposed by using the programs SUPCOMB or SITUS.

One possible reason for disagreement between atomic resolution models and SAXS results is the presence of heterogeneous multimerization. This may be apparent if the molecular weight is determined. Nevertheless the program OLIGOMER can take the calculated scattering profiles of several proposed multimers and determine the fractional concentration of each multimer required to best fit the data. An excellent fit with OLIGOMER and a poor fit with CRYSOL implies heterogeneous multimerization. Other possibilities for disagreement between models and SAXS data are flexibility and truncated loops (common in crystal structures where disordered regions are not built into models). Generating a variety of conformations of the flexible regions and using OLIGOMER may improve the final fit.

If the sample is a multimeric assembly and atomic models of all subunits are available but arrangement is unknown, then rigid-body modeling can be applied (Table 2, Fig. 25). Quaternary structure modeling of a complex against the SAXS data can be performed manually using the program MASSHA or fully automatically using program SASREF. If the full-length protein represents an assembly of the domains connected by linker regions, a combination of rigid-body modeling of the subdomains and *ab-initio* modeling of the missed linker using flexible chains of interconnected residues can be applied. The program BUNCH allows determination of three-dimensional structures of multidomain assemblies based on multiple scattering data sets from deletion mutants when the structure(s) of individual domains are available. Other conformational sampling methods can be applied (Table 3, Fig. 25) to model the quaternary structure of multidomain assemblies. Final models are evaluated in terms of the goodness of the fit of their calculated scattering curves to the experimental data. Comparison of models generated from rigid-body modeling and *ab-initio* shape restorations are also very useful in establishing the validity of the results (Fig. 25).
Parameters	Assessment
Experimental	
q-range	Range must be suitable through the entire spatial resolution required for determined models
Guinier plot	Non-linear behavior indicates aggregation or inappropriate q-range
R <sub>G</sub>	Consistency of extracted $R_{\rm G}$ with multiple methods (Table 1) increases confidence in not only $R_{\rm G}$ but also assigned $D_{\rm max}$
I(0)	Should correlate with molecular weight and concentration
$D_{\max}$	Proper description of the range of $D_{\text{max}}$ for well behaved $P(r)$ functions
P(r)	High frequency oscillations or discontinuities in <i>P</i> ( <i>r</i> ) may indicate problematic Fourier transform process
Structure modeling	
Goodness of fit (R or $\chi^2$ )	Validates <i>ab-initio</i> or atomic models agreement with experiment
NSD	Verifies stability and convergence of modeling
$R_{\rm G}$ model	Atomic model validation
$D_{\max}$ model	Atomic model validation
P(r) model	Atomic model validation

Table 5. SAXS parameters for data validation and interpretation

# 5.5 Criteria for evaluation of SAXS results

In preparing or evaluating publications with SAXS results, measurements and analyses it is critical to have metrics identifying the reliability of extracted information. In the long term, establishing standards for SAXS data analyses will be important for judging accurate SAXS measurements and modeling, and for the larger community in the interpretation and publication of SAXS data. Whereas universally accepted quality control parameters for SAXS results comparable to those currently used for crystallography do not yet exist, we suggest here some empirical criteria to consider for SAXS data-quality control and to help avoid over-interpreting SAXS results (Table 5). We propose this information should be presented in studies concerned with SAXS solution structure modeling of biological macromolecules.

The first crucial parameters for SAXS data validation comes from the linearity of the Guinier plot and  $R_{\rm G}$  value calculated from this plot (Figs 23, 24). Ideally the concentration dependence of the  $R_{\rm G}$  value should be presented and a corrected  $R_{\rm G}$  value calculated if necessary (Section 5.2). The P(r) function and derived  $D_{\rm max}$  value provide a second set of important parameters. If the P(r) function gives an extremely elongated tail, accurate characterization of  $D_{\rm max}$  may not be possible and a possible symptom of heterogeneity (Fig. 24). Further structural interpretation of these SAXS data may be suspect.  $D_{\rm max}$  can be validated by calculating the P(r) function in the larger r range where  $r_{\rm max}$  is greater than the expected  $D_{\rm max}$  and/or by calculating the P(r) function without the constraints that enforce  $P(D_{\rm max}) = 0$  (Section 2.3.3). I(0) and  $R_{\rm G}$  may also be calculated from the P(r) function and should correspond to those derived from the Guinier plot.

Molecular weight determination using I(0) is directly related to the excluded volume, and if SAXS is used to determine the oligomerization state, then accurate calibration of I(0) is necessary (Section 2.3.5, Fig. 24). These calibrations are particularly useful in identifying sub-stoichiometric amounts of aggregation or mixed assembly states that need to be analyzed and modeled as mixed populations and not homogenous samples that can be easily used for low-resolution structure reconstruction.

## 264 C. D. Putnam et al.

In solution structure modeling from SAXS measurements, it is important not only to validate the fit of the calculated to the experimental SAXS data using R or  $\chi^2$ , but also to compare multiple modeling runs (Fig. 25). Multiple trials of the modeling process significantly decrease the risk of over interpretation of underdetermined models. We suggest that at least 10 repetitions are required to obtain an accurate average model for evaluation. Probably the easiest way to validate differences in the models is to use a NSD parameter to validate the stability and convergence of the multiple modeling rounds (Section 4.3). For cases where rigid-body refinement is used, comparisons of  $R_G$ ,  $D_{max}$ , and the P(r) functions of the models to the experimentally determined values can also provide substantial insight into the quality of the model as well as the differences between any pseudo-atomic model and the experimentally observed scattering (Sections 4.2.4, 4.2.5). Handedness cannot be resolved using SAXS structures alone. Conclusions where chirality is important should be carefully examined.

## 5.6 Accuracy and resolution of SAXS experiments, measurements, and models

A critical issue for both SAXS and crystallography is defining the level of accuracy of resulting models. Essentially, what is the resolution of the model? For crystallography, the resolution of the structure is defined as the resolution of the data used to determine the structure. Robust quantitative measures are used to define the resolution of the resulting structure, even if the exact values of some criteria are debated. For example, a common set of criteria to define a 2 Å resolution crystal structure would expect over 50% of the data with Bragg spacings of 2 Å to be measured with a signal-to-noise ratio of more than 2 and good agreement between multiply observed reflections ( $R_{sym} \ll 50\%$ ).

A SAXS resolution *per se* is more difficult to define, as there are two different types of resolution that need to be addressed: (1) the nominal resolution of the experimental scattering curve, and (2) the effective resolution of the model. The nominal resolution of the experimental scattering curve can described in a manner related to that of crystallography. The relationship between q and Bragg spacing (d) is  $d = \lambda/(2 \sin(\theta)) = 2\pi/q$ . Thus,  $q_{\min}$  sets the largest dimensions observable in an experiment  $(q_{\min} = 0.006 \text{ Å}^{-1} \text{ or } d = 1000 \text{ Å}$  is often possible). The limiting experimental factor is how close reliable data can be determined near the primary beam. Similarly,  $q_{\max}$  sets the smallest Bragg spacing observable in an experiment  $(q_{\max} = 0.6 \text{ Å}^{-1} \text{ or } d = 1000 \text{ Å}$  is often possible). The limiting experimental consideration is the noise in the experimental data. At Bragg spacings smaller than 10 Å, the signal due to organization of bulk solvent begins to overpower the signal from the relatively small population of solutes (Head-Gordon & Hura, 2002), as the difference between the buffer solution and protein solution becomes very small at larger values of q.

The resolution of models derived from SAXS reconstructions is frequently not described in the literature, and models are simply termed 'low-resolution' structures. Part of the difficulty in determining the resolution of SAXS envelopes or SAXS-based models arises from the fact that fitting of the SAXS curve does not provide a unique solution, particularly for *ab-initio* envelopes. Averaging is needed to construct a reasonable model for the observed scattering. The resolution of the final averaged model depends both on reciprocal resolution defined by measured *q*-range and data quality, as well as on the molecular size, shape, and flexibility (Section 4.3.1). For example, averaging of different models can eliminate higher resolution details, particularly if molecules are flexible such as shown in Fig. 9. The single *ab-initio* model (orange beads) matches the experimental data to highest resolution 13 Å ( $q_{max}$ =0.47 Å<sup>-1</sup>); however, the model's structural details do not reach this spatial resolution, and the averaged model (blue transparent beads) is at an even lower resolution. Alternately, more details can be visualized through the incorporation of atomic-resolution models in conjunction with SAXS data (Fig. 9), and can allow details of the model to more closely match the experimental resolution, which can be critical for determining different types of structural details (Fig. 28). In general, if the synchrotron-based high-angle scattering is combined with crystal structure even the dynamic aspects of reaction-linked changes in protein conformation can be quantitatively monitored (Tiede *et al.* 2002). Thus in a well-defined SAXS experiment, the effective resolution can be impressive and certainly sufficient to address questions of conformational state (Davies *et al.* 2005; Chen *et al.* 2007; Yamagata & Tainer, 2007), see for example Fig. 17. In contrast, flexible multi-domain systems will show broad variations from the averaged structures but can be improved by introducing atomic models (Fig. 19).

Although the nominal experimental resolution sets a limit on the maximum possible resolution for the SAXS model, the resolution of the data does not define the effective resolution of the model. Determining the effective resolution is more difficult and has strong similarities for the determination of the resolution of models derived from EM. In EM, resolution is estimated by the Fourier shell method, which indicates the highest resolution shells that show agreement in reciprocal space for two independently processed sets of data. This assumes coherence in Fourier space and random Gaussian error in electron density maps. In principle, this method could be used for SAXS, but validation of this approach is currently a research problem, as the degree of systematic error is not as well defined as it is for EM or crystallography. In NMR, structures are not defined as having any particular resolution, but are rather given a confidence level equivalent by the root-mean-squared derivation from the averaged structure. A SAXS equivalent would be to examine the variance of the single model volume from the averaged volume envelope. As we favor the use of high-resolution structures as restraints in the SAXS experiments, another approach would be to define the variability of a fit of known domain structures into the SAXS envelope. This would be analogous to the accurate correlation-based approaches used for EM, where the solution sets are used to define confidence intervals and error margins for the fitting parameters (Volkmann & Hanein, 2003). The definition of valid and robust measures of resolution will be important advance for SAXS experiments. A true measure of resolution will allow researchers know whether to trust or ignore the finer features of SAXS shapes and conformational changes as well as enable them to identify to what extent additional runs of shape modeling programs will enhance their final model. We recommend that publications of SAXS models make explicit the sources of error, intended level of accuracy, and estimated model resolution with the bases for the estimates stated. Ab-initio SAXS-based models should be used for experimental design or in combination with other types of information, as they are generally not sufficiently over determined by the SAXS experiment alone.

We note that the resolution required in any experiment depends upon the biological question of interest. The lower resolution diffraction from B-form DNA turned out to be more useful than the higher resolution A-form DNA diffraction for defining the structure of the double helix (Franklin & Gosling, 1953a, b). In cases involving conformational flexibility and transitions, lower resolution results that can be used to describe the flexibility can potentially answer biological questions of interest better than high-resolution experiments that enforce a single conformation. In general, the great strength of SAXS experiments is that they characterize the overall molecular shape and assembly in solution, and delineate the architectural arrangements needed to place high-resolution structures of components. This is precisely why the



**Fig. 28.** The resolution range required to identify structural features in SAXS data is qualitatively illustrated showing the theoretical X-ray profile calculated by CRYSOL (Svergun *et al.* 1995) for proteins with A. different oligomerization states, B. different domain conformations and C. different structural fluctuations. The resolution ranges are highlighted with blue box and the upper axis of the inset graphs indicate the spatial resolution ( $\Delta = 2\pi/q$ ) of this range. (*a*) The low angle scattering information (500–30 Å resolution) is determined particle size and overall shape, as illustrated by a monomeric (black) and dimeric (red) assemblies of the extracellular fibrinogen-binding protein (PDB id 2gom; Hammel *et al.* 2007). (*b*) The medium angle scattering (40–16 Å resolution) provides information about domain motions, as illustrated by the scattering profiles of human glucokinase without (black; PDB id 1v4s) and with (red) a glucose analog (PDB id 1v4t; Kamata *et al.* 2004). (*c*) The high angle scattering information (16–7 Å resolution) provides information about small structural fluctuations, as illustrated using the different conformations of the oxidized horse heart cytochrome c determined by crystallography (black; PDB id 1cr; Sanishvili *et al.* 1995) and NMR (red; PDB id 1akk; Banci *et al.* 1997).

combination of SAXS with a high-resolution method such as crystallography becomes so powerful.

## 6. Prospects and conclusions

# 6.1 General biological and biophysical implications

The use of both X-ray crystallography and SAXS is poised to be tremendously relevant for addressing the different biological pathways that control cells and offers enormous potential impacts in areas ranging from bio-energy to medicine. The current initiatives in genome sequencing along with the resulting whole-genome bioinformatics studies have revealed that not only are the components of signaling pathways modular (Bahattacharyya *et al.* 2006), but so are the proteins that make them up (Apic *et al.* 2001). These modular architectures within complexes and within individual proteins allow independent activities to be functionally coordinated. At the extreme, there are important proteins playing dynamic roles where interactions between multiple partners are exchanged to coordinate important cellular processes. Dynamic interactions within cells may be important in several respects: (1) to increase recognition probability, via significantly enlarging the area and three-dimensional size of the target, (2) to reduce disruptive interference among macromolecular steps and pathways, (3) to create highly effective local concentrations of components at target sites, (4) to promote pathway coordination, and (5) to allow a degree of self-regulation.

Biophysical studies have indicated that macromolecular complexes are formed through local contacts dominated by short-range interactions at the binding surface. Shape and chemical complementarity are critical features that control the mechanisms of assembly. Thus, the control of interface shape provides simple structure-based mechanisms to control cellular processes. Moreover, these studies also predict the existence of molecules that are primarily required as molecular scaffolds and jigs that bring together appropriate active components within complexes and may possess substantial flexibility. In some cases these scaffolding molecules and domains are dispensable with the introduction of appropriate fusions. For example, telomeres can be properly maintained when Cdc13 and Stn1 are replaced with a fusion of the Cdc13 DNA binding domain onto Stn1 (Pennock et al. 2001). Similarly, a Cdc13-Est2 fusion protein eliminates the requirement for the Est1 protein for telomerase function (Evans & Lundblad, 1999). The presence of these scaffolding domains and proteins in normal cells, however, provides the opportunity for additional cellular levels of control. For example, fusion of the Sld3 and Dpb11 proteins bypasses the requirement for the phosphorylation of Sld3 by the budding yeast cyclindependent kinase and the phospho-binding BRCT domains of Dpb11 (Zegerman & Diffley, 2007). However, this fusion comes at the cost of the regulation that the cyclin-dependent kinase normally performs in coordinating the yeast cell cycle (reviewed in Mendenhall & Hodge, 1998). Understanding these types of mechanisms in the context of the native systems will require structural characterization of the various conformational states these proteins adopt, regardless of whether or not a particular molecule or conformational state is suitable for high-resolution structural studies.

Decision-making by biological pathways involves dynamic molecular interactions. These interactions are pathway specific and can include the formation of cooperative ensembles, use of interface mimicry and exchange, switching of states in chemo-mechanical assemblies, and flexing by unstructured regions. Studying these processes by any single biophysical technique can be remarkably challenging. Thus, many of these systems are ideal targets for the combination of SAXS with high-resolution structures and computational techniques.

### 6.2 Needs and prospects

Combining data from solution scattering with atomic resolution structures holds tremendous promise for addressing biophysical details of how specific complexes and flexibility drive biological processes. The challenge has been that this combination requires understanding of biology, structures, solution scattering, and computational methods. Thus, the goal of this review has been to introduce sufficient detail for each of these techniques so that they can be productively applied in the context of this problem. We note that advances continue to be made in each of these fields, and we predict that several areas in SAXS data collection and processing are positioned to become important developments in the near future.

One issue that would benefit from sustained theoretical and practical study is the development of a universally accepted SAXS assessment factor or factors that are equivalent to the *R*-factor in crystallography. One of the fundamental requirements in the modeling is to be able to successfully measure the fit of proposed models to solution scattering curves. As detailed in Section 4.1, we do not consider this to be a solved problem, and we expect that multiple measures might be required depending on the types of modeling that are performed. An equally important statistical measure for evaluating this fitting is an equivalent to the crystallographic  $R_{\rm free}$ . However, a perfect analog is unlikely, as the limited number of independent parameters measured in a SAXS experiment make it unwise to extract a sufficient number for use as an unrefined reference set.

Additionally, we believe that efforts to automate SAXS data collection and processing will be profoundly important. The requirement of using the same cuvette for buffers and samples typically means that cuvette washing and filling take up a substantial fraction of the synchrotron time allocate for experiments. Thus, the use of flow cells, robotics, and temperature-controlled sample holders offer the possibility of streamlining the SAXS data collection process and can allow for rapid screening of many conditions. These can include different buffer conditions to control aggregation, titration conditions to understand specific conformational changes, or small molecules from libraries to search for lead compounds that inhibit or promote complex formation, control the formation of active states, or control the folding of specific targets. Similarly, the possibility of using SAXS in conjunction with an inline size-exclusion chromatography system may allow for separation and characterization of samples with dynamic heterogeneity in their assembly states. We are excited about these types of technological advances, as our experience has suggested that the ease for which experiments can be automated directly controls the type and scale of experiments that will be attempted.

We also anticipate that the use of SAXS in the study of nucleic acids, particularly RNA enzymes and riboswitches will become increasingly important, such as illustrated by glycine riboswitch (Lipfert *et al.* 2007b) and the SAM riboswitch (Fig. 26). RNA scatters X-rays approximately five times more strongly than proteins (Fang *et al.* 2000), allowing for useful characterizations of less concentrated samples. Further, SAXS can provide helpful constraints onto the large number of potential secondary and tertiary structures that are predicted by folding algorithms such as MFOLD (Zuker *et al.* 1999) and frequently use sequence conservation and nuclease sensitivity to be validated. RNA activity often involves conformational switching

that is well defined by SAXS. SAXS allows the examination of multiple RNA states to define the active conformation, whereas the low-energy states determined in macromolecular crystallography have frequently been of inactive conformations. As SAXS provides direct measures of shape in solution, it may furthermore aid in the discovery of proteins that regulate processes by mimicry of DNA and RNA structures (Putnam *et al.* 1999; Putnam & Tainer, 2005).

SAXS appears to offer significant potential advantages for examining the structure of membrane proteins in lipid bilayers and detergents, although this application of SAXS is still an emerging technology. The size and complexity of membrane protein complexes with detergent and lipids are major challenges to crystallization and NMR experiments, but provide potential advantages for SAXS experiments by providing enhanced contrast compared to aqueous complexes. The central challenge for SAXS is that the buffer, proteins, and lipid/detergent mixture all have different electron densities, complicating the analysis. Matching the scattering of the lipid and detergent in SAXS or SANS provides one way to address these multicomponent systems (Bu & Engelman, 1999). SVD has also been used to extract the scattering profile of protein-detergent complexes at concentrations above the critical micelle concentrations (Section 4.4.1; Lipfert et al. 2007a). Embedding membrane proteins in cubic lipid phases can also control the nature of the lipid/detergent systems allowing for the generation of more closely matched blank samples (Caffrey, 2000; Lunde et al. 2006). Another approach has been the use of mutant lipid-carrying proteins (Denisov et al. 2004; Bayburt et al. 2006) to encircle the hydrophobic lipid/membrane protein complexes to form 'nano-disks' (Nath et al. 2007). This system can potentially yield monodisperse and homogeneous particles whose size can be controlled by varying lipoprotein size; however, sample preparation and control of particle homogeneity are non-trivial challenges. While all of these current approaches will benefit from further use and development, they demonstrate the potential of SAXS to aid in the characterization of membrane proteins in lipid bilayers and detergents.

Using SAXS data as a source of experimental restraints for modeling macromolecular flexibility and computational docking is an exciting and relatively underdeveloped possibility. SAXS data from appropriate samples can provide important experimental feedback, and could be usefully extended to include dynamic conformational changes characterized by time-resolved experiments. Time-resolved measurements require very high X-ray flux and fast detectors designed for rapid electronic shuttering. Both are now available, and SAXS, unlike traditional NMR and fluorescence experiments, is not affected by the molecular rotation times, so time-resolved SAXS can be performed in an equivalent manner to the traditional static experiments. Millisecond-resolved SAXS experiments performed to date have primarily focused on following changes in R<sub>G</sub> from the highest intensity region of the scattering curve during the folding of RNAs and proteins (e.g. Kwok et al. 2006; Uzawa et al. 2006). A natural complement to the global shape and conformation from SAXS will be residue level information from advancing techniques of enhanced hydrogen/deuterium exchange mass spectrometry, which can approach single-residue resolution as shown for the photocycle changes of photoactive yellow proteins (Brudler et al. 2006). Thus, SAXS is well positioned to become an important player along with new weak-field aligned NMR and fluorescence experiments that can probe samples in the biologically interesting millisecond time-frame. With appropriate resources for directed efforts, SAXS can provide complementary experimental data on flexibility in macromolecular interactions with widespread impacts.

### 6.3 Expectations and predictions

A fundamental strength of SAXS is that it provides the overall structure including both architectural arrangements and conformations in the 50–10 Å resolution range in near physiological conditions to yield the information needed to place high-resolution structures of components in macromolecular complexes. Thus, SAXS has the potential for addressing many important issues in fundamental biology and human disease. We believe that SAXS can play an important role in identifying which of the >10 million polymorphisms identified in humans cause structural defects in macromolecules.

One major class of such disease-causing mutations is thought to cause defects in folding and structure, which can be readily observed in SAXS. Mutations in this class have already been identified to affect a number of medically relevant proteins such as the DNA repair protein BRCA1, where mutations are associated with predisposition to breast cancer, the DNA damage response protein p53, where mutations are associated with predispositions to numerous cancer types, the Werners syndrome helicase-nuclease where mutations cause rapid aging and the reactive oxygen control enzyme superoxide dismutase, where mutations are associated with neurodegenerative disease (Deng et al. 1993; Foster et al. 1999; DiDonato et al. 2003; Williams et al. 2003; Wu & Hickson, 2006; Perry et al. 2006, 2007). SAXS experiments provide an efficient means to ask if polymorphisms may cause structural defects as seen for the mitochondrial superoxide dismutase (Borgstahl et al. 1996). The ability to do rapid SAXS analyses will also allow effective use of comparative genomics experiments to distinguish different assemblies that maintain similar activities among different organisms, such as occurs for eukaryotic and microbial Cu,Zn superoxide dismutases (Bourne et al. 1996). Similarly SAXS provides a basis to examine the structural conversion of the cellular prion protein into a misfolded isoform prion shape that causes human disease (Redecke et al. 2007). As SAXS has intrinsic advantages for examining both folding and shape in solution, it offers not only critical technology for characterization of such defects, but also an efficient and powerful means of drug discovery by SAXS-based screening for compounds that bind to and stabilize the native shape and assembly. SAXS offers obvious powerful advantages for identifying even low-affinity small molecule binders without requiring any labeling or the development of target specific assays. For the right problems, SAXS can be an important tool for identifying lead compounds.

A second major class of disease-causing mutations is likely to localize to macromolecular interfaces. Since these interfaces involve more residues than active site regions, more polymorphisms will likely alter interface residues. Polymorphisms that cause these surfaces to become altered or disrupted will be readily identifiable with SAXS. An important interface type involves acceptor sites that can bind multiple partners by interface mimicry and exemplified by the PCNA peptide-binding surface (Chapados *et al.* 2004; Sakurai *et al.* 2005; Dore *et al.* 2006; Pascal *et al.* 2006) or by exchange of similar interactions by multiple proteins as controlled by a third component such as DNA such as by the Rad51 polymerization interface (Shin *et al.* 2003). Atomic arrangements derived from SAXS can provide additional information that can clearly resolve the altered nature of these assemblies, such as distinguishing between open trimeric versus dimeric PCNA rings that are more difficult to identify by other techniques. For example, SAXS shape predictions for PCNA accurately predicted a trimeric ring assembly while also allowing accurate prediction of the folded region of a full-length DNA repair glycosylase including structural alterations resulting from crystal contacts and truncation of a large unstructured region (Tsutakawa *et al.* 2007). Thus, the characterization of static and dynamic

complexes by SAXS may be equally or more important than high-resolution structures of component active sites for understanding the implications of the genome project in cell biology and human health.

Results from the structural genomics initiatives provide compelling evidence for the utility of advanced SAXS technologies. These efforts indicate that about half of all eukaryotic proteins have unstructured regions of over 40 residues in length and many are at least partially unfolded without their specific protein partners. Given the ubiquity of flexibility in macromolecules, it is almost certain that deciphering the mechanistic details of biological pathways will require the integration of techniques, like SAXS. SAXS can define the overall shape, conformation, and architecture, including unstructured regions and those structures that may not adopt single states suitable for high-resolution structural studies. As more folded domain structures are being solved, the best use of available SAXS experimental information will incorporate these domains as external information, analogously to the use of residue stereochemical constraints to obtain accurate atomic models from 2.5 Å to 3.5 Å diffraction data. Similarly, for RNA and multidomain proteins that undergo functionally important conformational changes, SAXS can test and validate active conformational states in solution. For the many membrane bound proteins that remain difficult to crystallize, SAXS provides a possible general solution to the structural characterization of membrane proteins in their hydrophobic environments with the added advantage that lipids and detergents scatter less than protein. Their contribution to the scattering may be minimized by appropriate choice of buffer. SAXS provides appropriate experimental feedback for computational modeling of conformational landscapes, docking, denatured proteins, and the folding of proteins and RNA, which should improve the accuracy of computational simulations and predictions. SAXS furthermore provides an experimental basis to identify structural similarity even in the absence of sequence homology and thereby direct efficient molecular replacement efforts for phasing crystal structure data and also validate the relevance of cross-genomic structural modeling.

SAXS is an effective and important complement to crystallography as it can provide information on every sample, faster data collection than EM or NMR, and native structural analysis in solution. Furthermore, SAXS results provide an efficient and powerful way to identify experimentally testable models for macromolecular interactions and conformations in solution. With the millions invested in crystallography to develop high-throughput structural analyses, it is tragic that SAXS, which certainly has the capacity to be a true high-throughput technique, has been neglected by comparison. The investment in SAXS is surprisingly small, as SAXS can evaluate samples and to aid in the efficient optimization of constructs for crystallography. However, given the limited funding for SAXS in the USA, it is no surprise that much of the development and software for SAXS has been accomplished in Europe and Japan. Yet, this is an exciting time in the development of SAXS and the modeling of structures using a combination of SAXS and atomic resolution structures. Much of the technological and computational infrastructure is currently in place, which substantially lowers the barriers for new experimenters to use SAXS and interpret SAXS data in their biophysical experiments. In the next decade, any funding in SAXS will more than pay for themselves in real results and substantial technological advances that address important unsolved problems in biology, medicine, nanotechnology, and biotechnology. Thus, as these SAXS tools and technologies evolve and become widely adopted, we expect that they will be applied in novel ways to not only to solve existing problems in structural biology but also to play an active role in pushing the cutting edge of research in structural biology.

### 7. Acknowledgments

We thank Betsy Goldsmith, Lorena Beese, Wei Yang, Anna-Maria Hays Putnam, Meindert Lamers, Marc Mendillo, Ken Frankel, Susan Tsutakawa, Elizabeth Getzoff, Niels Volkmann, Michael Wall, Scott Brown, Eva Nogales and Brian Chapados for comments on the manuscript. We are grateful to our collaborators and co-workers and in particular Richard Kolodner, Marc Mendillo, Scarlet Shell, Veronique Receveur Brechot, Tom Ellenberger, Hiro Tsuruta, Robert Rambo and Henri-Pierre Fierobe for contributions to the projects described. X-ray scattering and diffraction technologies and their applications to macromolecular shapes and conformations in solution at the SIBYLS beamline at Lawrence Berkeley National Laboratory are supported in part by the DOE program Integrated Diffraction Analysis Technologies (IDAT) and the DOE program Molecular Assemblies Genes and Genomics Integrated Efficiently (MAGGIE) under Contract Number DE-AC02-05CH11231 with the U.S. Department of Energy. Efforts to apply SAXS and crystallography to characterize eukaryotic pathways relevant to human cancers are supported in part by National Cancer Institute grant CA92584. J.A.T. research on flexible protein interactions and assemblies is supported in part by NIH grants AI22160, CA81967, CA104660, and GM46312. C.D.P. was funded as a Robert Black fellow by the Damon Runyon Cancer Research Foundation.

# 8. References

- AKIYAMA, S., FUJISAWA, T., ISHIMORI, K., MORISHIMA, I. & AONO, S. (2004). Activation mechanisms of transcriptional regulator CooA revealed by small-angle X-ray scattering. *Journal of Molecular Biology* 341, 651–668.
- AMADEI, A., LINSSEN, A. B. & BERENDSEN, H. J. (1993). Essential dynamics of proteins. *Proteins* 17, 412–425.
- ANDRESEN, K., DAS, R., PARK, H. Y., SMITH, H., KWOK, L. W., LAMB, J. S., KIRKLAND, E. J., HERSCHLAG, D., FINKELSTEIN, K. D. & POLLACK, L. (2004). Spatial distribution of competing ions around DNA in solution. *Physical Review Letters* **93**, 248103.
- APIC, G., GOUGH, J. & TEICHMANN, S. A. (2001). An insight into domain combinations. *Bioinformatics* 17, S83–S89.
- ASLAM, M., GUTHRIDGE, J. M., HACK, B. K., QUIGG, R. J., HOLERS, V. M. & PERKINS, S. J. (2003). The extended multidomain solution structures of the complement protein Crry and its chimeric conjugate Crry-Ig by scattering, analytical ultracentrifugation and constrained modelling: implications for function and therapy. *Journal of Molecular Biology* 329, 525–550.
- ASLAM, M. & PERKINS, S. J. (2001). Folded-back solution structure of monomeric factor H of human complement by synchrotron X-ray and neutron scattering, analytical ultracentrifugation and constrained molecular modelling. *Journal of Molecular Biology* **309**, 1117–1138.
- BAHADUR, R. P., CHAKRABARTI, P., RODIER, F. & JANIN, J. (2004). A dissection of specific and non-specific protein-protein interfaces. *Journal of Molecular Biology* 336, 943–955.

- BAHAR, I., ATILGAN, A. R. & ERMAN, B. (2001). Anisotropy of fluctuation dynamics of proteins with an elastic network model. *Biophysical Journal* 80, 505–515.
- BAHATTACHARYYA, R. P., REMENYI, A., YEH, B. J. & LIM, W. A. (2006). Domains, motifs, and scaffolds: the role of modular interactions in the evolution and wiring of cell signaling circuit. *Annual Reviews of Biochemistry* 75, 655–680.
- BANCI, L., BERTINI, I., GRAY, H. B., LUCHINAT, C., REDDIG, T., ROSATO, A. & TURANO, P. (1997). Solution structure of oxidized horse heart cytochrome c. *Biochemistry* 36, 9867–9877.
- BARRETT, C. P. & NOBLE, M. E. (2005). Dynamite extended: two new services to simplify protein dynamic analysis. *Bioinformatics* 21, 3174–3175.
- BAX, A. & GRISHAEV, A. (2005). Weak alignment NMR: a hawk-eyed view of biomolecular structure. *Current Opinion in Structural Biology* **15**, 563–570.
- BAYBURT, T. H., GRINKOVA, Y. V. & SLIGAR, S. G. (2006). Assembly of single bacteriorhodopsin trimers in bilayer nanodiscs. *Archives of Biochemistry and Biophysics* 450, 215–222.
- BERGMANN, A., FRITZ, G. & GLATTER, O. (2000). Solving the generalized indirect Fourier transformation (GIFT) by Boltzmann simplex simulated annealing (BSSA). *Journal of Applied Crystallography* 33, 1212–1216.
- BERNADO, P., BLANCHARD, L., TIMMINS, P., MARION, D., RUIGROK, R. W. & BLACKLEDGE, M. (2005). A structural model for unfolded proteins from residual dipolar couplings and small-angle X-ray scattering. *Proceedings of the National Academy of Sciences USA* **102**, 17002–17007.

- BERNADO, P., MYLONAS, E., PETOUKHOV, M. V., BLACKLEDGE, M. & SVERGUN, D. I. (2007). Structural characterization of flexible proteins using small-angle X-ray scattering. *Journal of the Americal Chemical Society* 129, 5656–5664.
- BERNSTEIN, F. C., KOETZLE, T. F., WILLIAMS, G. J. B., MEYER, E. F., BRICE, M. D., RODGERS, J. R., KENNARD, O., SHIMANOUCHI, T. & TASUMI, M. (1977). The Protein Data Bank: a computer-based archival file for macromolecular structures. *Journal of Molecular Biology* **112**, 535–542.
- BEURON, F., FLYNN, T. C., MA, J., KONDO, H., ZHANG, X. & FREEMONT, P. S. (2003). Motions and negative cooperativity between p97 domains revealed by cryoelectron microscopy and quantized elastic deformation model. *Journal of Molecular Biology* 327, 619–629.
- BILGIN, N., EHRENBERG, M., EBEL, C., ZACCAI, G., SAYERS, Z., KOCH, M. H., SVERGUN, D. I., BARBERATO, C., VOLKOV, V., NISSEN, P. & NYBORG, J. (1998). Solution structure of the ternary complex between aminoacyltRNA, elongation factor Tu, and guanosine triphosphate. *Biochemistry* 37, 8163–8172.
- BINDER, K. & HEERMAN, D. (1992). Monte Carlo Simulation in Statistical Physics, 2nd edn. Berlin: Springer-Verlag.
- BJORNSON, K. P., BLACKWELL, L. J., SAGE, H., BAITINGER, C., ALLEN, D. & MODRICH, P. (2003). Assembly and molecular activities of the MutS tetramer. *Journal of Biological Chemistry* 278, 34667–34673.
- BLOW, D. M. & CRICK, F. H. C. (1959). The treatment of errors in the isomorphous replacement method. *Acta Crystallographica* 12, 794–802.
- BLUNDELL, T. L. & JOHNSON, L. N. (1976). Protein Crystallography. London: Academic Press.
- BOCHTLER, M., HARTMANN, C., SONG, H. K., BOURENKOV, G. P., BARTUNIK, H. D. & HUBER, R. (2000). The structures of HslU and the ATP-dependent protease HslU-HslV. *Nature* 403, 800–805.
- BOEHM, M. K., WOOF, J. M., KERR, M. A. & PERKINS, S. J. (1999). The Fab and Fc fragments of IgA1 exhibit a different arrangement from that in IgG: a study by X-ray and neutron solution scattering and homology modelling. *Journal of Molecular Biology* 286, 1421–1447.
- BOEHR, D. D., MCELHENY, D., DYSON, H. J. & WRIGHT, P. E. (2006). The dynamic energy landscape of dihydrofolate reductase catalysis. *Science* 313, 1638–1642.
- BONVIN, A. M. (2006). Flexible protein-protein docking. Current Opinions in Structural Biology 16, 194–200.
- BORGSTAHL, G. E., PARGE, H. E., HICKEY, M. J., JOHNSON, M. J., BOISSINOT, M., HALLEWELL, R. A., LEPOCK, J. R., CABELLI, D. E. & TAINER, J. A. (1996). Human mitochondrial manganese superoxide dismutase polymorphic variant Ile58Thr reduces activity by destabilizing the tetrameric interface. *Biochemistry* 35, 4287–4297.
- BOSKOVIC, J., RIVERA-CALZADA, A., MAMAN, J. D., CHACON, P., WILLISON, K. R., PEARL, L. H. & LLORCA, O. (2003). Visualization of DNA-induced conformational changes

in the DNA repair kinase DNA-PKcs. *EMBO Journal* 22, 5875–5882.

- BOURNE, Y., ARVAI, A. S., BERNSTEIN, S. L., WATSON, M. H., REED, S. I., ENDICOTT, J. E., NOBLE, M. E., JOHNSON, L. N. & TAINER, J. A. (1995). Crystal structure of the cell cycle-regulatory protein suc1 reveals a betahinge conformational switch. *Proceedings of the National Academy of Sciences USA* 92, 10232–10236.
- BOURNE, Y., REDFORD, S. M., STEINMAN, H. M., LEPOCK, J. R., TAINER, J. A. & GETZOFF, E. D. (1996). Novel dimeric interface and electrostatic recognition in bacterial Cu,Zn superoxide dismutase. *Proceedings of the National Academy of Sciences USA* 93, 12774–12779.
- BROOKS, B. & KARPLUS, M. (1985). Normal modes for specific motions of macromolecules: applications to the hinge-bending mode of lysozyme. *Proceedings of the National Academy of Sciences USA* 82, 4995–4999.
- BRUDLER, R., GESSNER, C. R., LI, S., TYNDALL, S., GETZOFF, E. D. & WOODS JR., V. L. (2006). PAS domain allostery and light-induced conformational changes in photoactive yellow protein upon I2 intermediate formation, probed with enhanced hydrogen/deuterium exchange mass spectrometry. *Journal of Molecular Biology* 363, 148–160.
- BRULET, A., BOUE, F. & COTTON, J. P. (1996). About the experimental determination of the persistence length of wormlike chains of polystyrene. *Journal of Physics II* (*France*) 6, 855–891.
- BRUNGER, A. T. (1992). Free R value: a novel statistical quantity for assessing the accuracy of crystal structures. *Nature* 355, 472–475.
- BRUNGER, A. T., ADAMS, P. D., CLORE, G. M., DELANO, W. L., GROS, P., GROSSE-KUNTSLEVE, R. W., JIANG, J. S., KUSZEWSKI, J., NILGES, M., PANNU, N. S., READ, R. J., RICE, L. M., SIMONSON, T. & WARREN, G. L. (1998). Crystallography & NMR system: a new software suite for macromolecular structure determination. *Acta Crystallographica* **D54**, 905–921.
- BRUNGER, A. T., KURIYAN, J. & KARPLUS, M. (1987). Crystallographic R factor refinement by molecular dynamics. *Science* 235, 458–460.
- BRUNNER-POPELA, J. & GLATTER, O. (1997). Small-angle scattering of interacting particles. I. Basic principles of a global evaluation technique. *Journal of Applied Crystallography* **30**, 431–442.
- BU, Z. & ENGELMAN, D. M. (1999). A method for determining transmembrane helix association and orientation in detergent micelles using small angle x-ray scattering. *Biophysical Journal* 77, 1064–1073.
- BUEY, R. M., MONTERROSO, B., MENENDEZ, M., DIAKUN, G., CHACON, P., HERMOSO, J. A. & DIAZ, J. F. (2007). Insights into molecular plasticity of choline binding proteins (Pneumococcal surface proteins) by SAXS. *Journal of Molecular Biology* 365, 411–424.

- BURMEISTER, W. P. (2000). Structural changes in a cryocooled protein crystal owing to radiation damage. Acta Crystallographica D56, 328–341.
- CAFFREY, M. (2000). A lipid's eye view of membrane protein crystallization in mesophases. *Current Opinions in Structural Biology* 10, 486–497.
- CALMETTES, P., DURAND, D., DESMADRIL, M., MINARD, P., RECEVEUR, V. & SMITH, J. C. (1994). How random is a highly denatured protein? *Biophysical Chemistry* 53, 105–114.
- CANADY, M. A., TSURUTA, H. & JOHNSON, J. E. (2001). Analysis of rapid, large-scale protein quaternary structure changes: time-resolved X-ray solution scattering of *Nudaurelia capensis* omega virus (NΩV) maturation. *Journal of Molecular Biology* **311**, 803–814.
- CHACON, P., DIAZ, J. F., MORAN, F. & ANDREU, J. M. (2000). Reconstruction of protein form with X-ray solution scattering and a genetic algorithm. *Journal of Molecular Biology* 299, 1289–1302.
- CHACON, P., MORAN, F., DIAZ, F. J., PANTOS, E. & ANDREU, J. M. (1998). Low-resolution structures of proteins in solution retrieved from X-ray scattering with a genetic algorithm. *Biophysical Journal* 74, 2760–2775.
- CHACON, P., TAMA, F. & WRIGGERS, W. (2003). Megadalton biomolecular motion captured from electron microscopy reconstructions. *Journal of Molecular Biology* 326, 485–492.
- CHACON, P. & WRIGGERS, W. (2002). Multi-resolution contour-based fitting of macromolecular structures. *Journal of Molecular Biology* **317**, 375–384.
- CHAMBERLAIN, A. K., RECEVEUR, V., SPENCER, A., REDFIELD, C. & DOBSON, C. M. (2001). Characterization of the structure and dynamics of amyloidogenic variants of human lysozyme by NMR spectroscopy. *Protein Science* 10, 2525–2530.
- CHAPADOS, B. R., HOSFIELD, D. J., HAN, S., QIU, J., YELENT, B., SHEN, B. & TAINER, J. A. (2004). Structural basis for FEN-1 substrate specificity and PCNA-mediated activation in DNA replication and repair. *Cell* **116**, 39–50.
- CHEN, B., DOUCLEFF, M., WEMMER, D. E., DE CARLO, S., HUANG, H. H., NOGALES, E., HOOVER, T. R., KONDRASHKINA, E., GUO, L. & NIXON, B. T. (2007). ATP ground- and transition states of bacterial enhancer binding AAA+ ATPases support complex formation with their target protein, *σ54. Structure* **15**, 429–440.
- CHEN, L., HODGSON, K. O. & DONIACH, S. (1996). A lysozyme folding intermediate revealed by solution X-ray scattering. *Journal of Molecular Biology* 261, 658–671.
- CHEN, R., LI, L. & WENG, Z. (2003). ZDOCK: an initialstage protein-docking algorithm. *Proteins* 52, 80–87.
- COLUMBUS, L., LIPFERT, J., KLOCK, H., MILLETT, I., DONIACH, S. & LESLEY, S. A. (2006). Expression, purification, and characterization of *Thermotoga maritima* membrane proteins for structure determination. *Protein Science* 15, 961–975.

- CORN, J. E., PEASE, P. J., HURA, G. L. & BERGER, J. M. (2005). Crosstalk between primase subunits can act to regulate primer synthesis in trans. *Molecular Cell* 20, 391–401.
- COWTAN, K. & MAIN, P. (1996). Phase combination and cross validation in iterated density-modification calculations. *Acta Crystallographica* D52, 43–48.
- COWTAN, K. & MAIN, P. (1998). Miscellaneous algorithms for density modifications. *Acta Crystallographica* D54, 487–493.
- CRAIG, L., VOLKMANN, N., ARVAI, A. S., PIQUE, M. E., YEAGER, M., EGELMAN, E. H. & TAINER, J. A. (2006). Type IV pilus structure by cryo-electron microscopy and crystallography: implications for pilus assembly and functions. *Molecular Cell* 23, 651–662.
- D'ARCY, A. (1994). Crystallizing proteins a rational approach? Acta Crystallographica D50, 469–471.
- DAINESE, E., SABATUCCI, A., VAN ZADELHOFF, G., ANGELUCCI, C. B., VACHETTE, P., VELDINK, G. A., AGRO, A. F. & MACCARRONE, M. (2005). Structural stability of soybean lipoxygenase-1 in solution as probed by small angle X-ray scattering. *Journal of Molecular Biology* 349, 143–152.
- DAUTER, Z. (1997). Data collection strategy. *Methods in Enzymology* **276**, 326–344.
- DAVIES, J. M., TSURUTA, H., MAY, A. P. & WEIS, W. I. (2005). Conformational changes of p97 during nucleotide hydrolysis determined by small-angle X-ray scattering. *Structure* 13, 183–195.
- DELARUE, M. & DUMAS, P. (2004). On the use of lowfrequency normal modes to enforce collective movements in refining macromolecular structural models. *Proceedings of the National Academy of Sciences USA* 101, 6957–6962.
- DENG, H.-X., HENTAI, A., TAINER, J. A., IQBAL, Z., CAYABYAB, A., HUNG, W.-Y., GETZOFF, E. D., HERZFELDT, B., ROOS, R. P., WARNER, C., DENG, G., SORIANO, E., SMYTH, C., PARGE, H. E., AHMED, A., ROSES, A. D., HALLEWELL, R. A., PERICAK-VANCE, M. A. & SIDDIQUE, T. (1993). Amyotropic lateral sclerosis and structural defects in Cu,Zn superoxide dismutase. *Science* 261, 1047–1051.
- DENISOV, I. G., GRINKOVA, Y. V., LAZARIDES, A. A. & SLIGAR, S. G. (2004). Directed self-assembly of monodisperse phospholipid bilayer Nanodiscs with controlled size. *Journal of the American Chemical Society* **126**, 3477–3487.
- DEPRISTO, M. A., DE BAKKER, P. I. & BLUNDELL, T. L. (2004). Heterogeneity and inaccuracy in protein structures solved by X-ray crystallography. *Structure* 12, 831–838.
- DIDONATO, M., CRAIG, L., HUFF, M. E., THAYER, M. M., CARDOSO, R. M., KASSMANN, C. J., LO, T. P., BRUNS, C. K., POWERS, E. T., KELLY, J. W., GETZOFF, E. D. & TAINER, J. A. (2003). ALS mutants of human superoxide

dismutase form fibrous aggregates via framework destabilization. *Journal of Molecular Biology* **332**, 601–615.

- DIEDERICHS, K., DIEZ, J., GRELLER, G., MUELLER, C., BREED, J., SCHNELL, C., VORNRHEIN, C., BOOS, W. & WELTE, W. (2000). Crystal structure of MalK, the ATPase subunit of the trehalose/maltose ABC transporter of the archaeon *Thermococcus litoralis*. *EMBO Journal* **19**, 5951–5961.
- DOMINGUEZ, C., BONVIN, A. M., WINKLER, G. S., VAN SCHAIK, F. M., TIMMERS, H. T. & BOELENS, R. (2004). Structural model of the UbcH5B/CNOT4 complex revealed by combining NMR, mutagenesis, and docking approaches. *Structure* 12, 633–644.
- DONIACH, S. (2001). Changes in biomolecular conformation seen by small angle X-ray scattering. *Chemical Reviews* 101, 1763–1778.
- DORE, A. S., KILKENNY, M. L., JONES, S. A., OLIVER, A. W., ROE, S. M., BELL, S. D. & PEARL, L. H. (2006). Structure of an archaeal PCNA1-PCNA2-FEN1 complex: elucidating PCNA subunit and client enzyme specificity. *Nucleic Acids Research* 34, 4515–4526.
- DOUBLIE, S. (1997). Preparation of selenometionyl proteins for phase determination. *Methods in Enzymology* 276, 523–530.
- DRENTH, J. (1994). *Principles of Protein X-ray Crystallography*. New York: Springer-Verlag.
- DROTSCHMANN, K., ARONSHTAM, A., FRITZ, H.-J. & MARINUS, M. G. (1998). The *Escherichia coli* MutL protein stimulates binding of Vsr and MutS to heteroduplex DNA. *Nucleic Acids Research* 26, 948–953.
- DUBOCHET, J., ADRIAN, M., CHANG, J. J., HOMO, J. C., LEPAULT, J., MCDOWALL, A. W. & SCHULTZ, P. (1988). Cryo-electron microscopy of vitrified specimens. *Quarterly Reviews in Biophysics* 21, 129–228.
- DYSON, H. J. & WRIGHT, P. E. (2005). Intrinsically unstructured proteins and their functions. *Nature Reviews of Molecular Cell Biology* 6, 197–208.
- ECHOLS, N., MILBURN, D. & GERSTEIN, M. (2003). MolMovDB: analysis and visulation of conformational change and structural flexibility. *Nucleic Acids Research* 31, 478–482.
- ENGH, R. A. & HUBER, R. (1991). Accurate bond and angle parameters for X-ray protein structure refinement. *Acta Crystallographica* A47, 392–400.
- EVANS, S. K. & LUNDBLAD, V. (1999). Est1 and Cdc13 as comediators of telomerase access. *Science* 286, 117–120.
- FANG, X., LITTRELL, K., YANG, X. J., HENDERSON, S. J., SIEFERT, S., THIYAGARAJAN, P., PAN, T. & SOSNICK, T. R. (2000). Mg<sup>2+</sup>-dependent compaction and folding of yeast tRNAPhe and the catalytic domain of the *B. subtilis* RNase P RNA determined by small-angle X-ray scattering. *Biochemistry* **39**, 11107–11113.
- FAURE, P., MICU, A., PERAHIA, D., DOUCET, J., SMITH, J. C. & BENOIT, J. P. (1994). Correlated intramolecular motions and diffuse X-ray scattering in lysozyme. *Nature Stuctural Biology* 1, 124–128.

- FERRE-D'AMARE, A. R. & BURLEY, S. K. (1997). Dynamic light scattering in evaluating crystallizability of macromolecules. *Methods in Enzymology* 276, 157–166.
- FERREIRA DA SILVA, F., PEREIRA, P. J., GALES, L., ROESSLE, M., SVERGUN, D. I., MORADAS-FERREIRA, P. & DAMAS, A. M. (2006). The crystal structure and solution structures of glyceraldehyde-3-phosphate dehydrogenase reveal different quaternary structures. *Journal of Biological Chemistry* 281, 33433–33440.
- FETLER, L., TAUC, P., HERVE, G., MOODY, M. F. & VACHETTE, P. (1995). X-ray scattering titration of the quaternary structure transition of aspartate transcarbamylase with a bisubstrate analogue: influence of nucleotide effectors. *Journal of Molecular Biology* 251, 243–255.
- FILGUEIRA DE AZEVEDO JR., W., DOS SANTOS, G. C., DOS SANTOS, D. M., OLIVIERI, J. R., CANDURI, F., SILVA, R. G., BASSO, L. A., RENARD, G., DA FONSECA, I. O., MENDES, M. A., PALMA, M. S. & SANTOS, D. S. (2003). Docking and small angle X-ray scattering studies of purine nucleoside phosphorylase. *Biochemical and Biophysical Research Communications* 309, 923–928.
- FOSTER, B. A., COFFEY, H. A., MORIN, M. J. & RASTINEJAD, F. (1999). Pharmacological rescue of mutant p53 conformation and function. *Science* 286, 2507–2510.
- FOWLER, A. G., FOOTE, A. M., MOODY, M. F., VACHETTE, P., PROVENCHER, S. W., GABRIEL, A., BORDAS, J. & KOCH, M. H. (1983). Stopped-flow solution scattering using synchrotron radiation: apparatus, data collection and data analysis. *Journal of Biochemical and Biophysical Methods* 7, 317–329.
- FRANKLIN, R. E. & GOSLING, R. G. (1953a). Evidence for 2-chain helix in crystalline structure of sodium deoxyribonucleate. *Nature* **172**, 156–157.
- FRANKLIN, R. E. & GOSLING, R. G. (1953b). Molecular configuration in sodium thymonucleate. *Nature* 421, 400–401.
- FRASER, R. D. B., MACRAE, T. P. & SUZUKI, E. (1978). An improved method for calculating the contribution of solvent to the X-ray diffraction pattern of biological molecules. *Journal of Applied Crystallography* 11, 693–694.
- FUJISAWA, T., URUGA, T., YAMAIZUMI, Z., INOKO, Y., NISHIMURA, S. & UEKI, T. (1994). The hydration of Ras p21 in solution during GTP hydrolysis based on solution X-ray scattering profile. *Journal of Biochemistry* (*Tokyo*) **115**, 875–880.
- FURNHAM, N., BLUNDELL, T. L., DEPRISTO, M. A. & TERWILLIGER, T. C. (2006). Is one solution good enough? *Nature Stuctural Biology* 13, 184–185.
- GALLAGHER, S. C., CALLAGHAN, A. J., ZHAO, J., DALTON, H. & TREWHELLA, J. (1999). Global conformational changes control the reactivity of methane monooxygenase. *Biochemistry* 38, 6752–6760.
- GARMAN, E. F. & SCHNEIDER, T. R. (1997). Macromolecular crystallography. *Journal of Applied Crystallog*raphy 30, 211–237.

- GETZOFF, E. D., TAINER, J. A. & OLSON, A. J. (1986). Recognition and interactions controlling the assemblies of beta barrel domains. *Biophysical Journal* 49, 191–200.
- GHERARDI, E., SANDIN, S., PETOUKHOV, M. V., FINCH, J., YOULES, M. E., OFVERSTEDT, L. G., MIGUEL, R. N., BLUNDELL, T. L., VANDE WOUDE, G. F., SKOGLUND, U. & SVERGUN, D. I. (2006). Structural basis of hepatocyte growth factor/scatter factor and MET signalling. *Proceedings of the National Academy of Sciences USA* 103, 4046–4051.
- GIACOVAZZO, C., MONACO, H. L., VITERBO, D., SCORDARI, F., GILLI, G., ZANOTTI, G. & CATTI, M. (1992). *Fundamentals of Crystallography*. Oxford: Oxford University Press.
- GILBERT, H. E., ASOKAN, R., HOLERS, V. M. & PERKINS, S. J. (2006). The 15 SCR flexible extracellular domains of human complement receptor type 2 can mediate multiple ligand and antigen interactions. *Journal of Molecular Biology* 362, 1132–1147.
- GILBERT, H. E., EATON, J. T., HANNAN, J. P., HOLERS, V. M. & PERKINS, S. J. (2005). Solution structure of the complex between CR2 SCR 1–2 and C3d of human complement: an X-ray scattering and sedimentation modeling study. *Journal of Molecular Biology* 346, 859–873.
- GILLMOR, S. A., VILLASENOR, A., FLETTERICK, R., SIGAL, E. & BROWNER, M. F. (1997). The structure of mammalian 15-lipoxygenase reveals similarity to the lipases and the determinants of substrate specificity. *Nature Stuctural Biology* 4, 1003–1009.
- GLATTER, O. (1977). A new method for the evaluation of small angle X-ray scattering data. *Journal of Applied Crystallography* **10**, 415–421.
- GLATTER, O. (1982). Data treatment. In Small Angle X-ray Scattering (ed. O. Glatter and O. Kratky), pp. 120–165. London: Academic Press.
- GOETTIG, P., BRANDSTETTER, H., GROLL, M., GOHRING, W., KONAREV, P. V., SVERGUN, D. I., HUBER, R. & KIM, J. S. (2005). X-ray snapshots of peptide processing in mutants of tricorn-interacting factor F1 from *Thermoplasma* acidophilum. Journal of Biological Chemistry 280, 33387– 33396.
- GRAILLE, M., ZHOU, C.-Z., RECEVEUR-BRECHOT, V., COLLINET, B., DECLERCK, N. & VAN TILEURGH, H. (2005). Activation of the LicT transcriptional antiterminator involves a domain swing/lock mechanism provoking massive structural changes. *Journal of Biological Chemistry* 280, 14780–14789.
- GRISHAEV, A., WU, J., TREWHELLA, J. & BAX, A. (2005). Refinement of multidomain protein structures by combination of solution small-angle X-ray scattering and NMR data. *Journal of the American Chemical Society* 127, 16621–16628.
- GROSSMANN, J. G., ABRAHAM, Z. H., ADMAN, E. T., NEU, M., EADY, R. R., SMITH, B. E. & HASNAIN, S. S. (1993). X-ray scattering using synchrotron radiation shows

nitrite reductase from *Achromobacter xylosoxidans* to be a trimer in solution. *Biochemistry* **32**, 7360–7366.

- GUARNE, A., RAMON-MAIQUES, S., WOLFF, E. M., GHIRLANDO, R., HU, X., MILLER, J. H. & YANG, W. (2004). Structure of the MutL C-terminal domain: a model of intact MutL and its roles in mismatch repair. *EMBO Journal* 23, 4134–4145.
- GUINIER, A. & FOURNET, F. (1955). Small Angle Scattering of X-rays. New York: Wiley Interscience
- GUTBERLET, T., HEINEMANN, U. & STEINER, M. (2001). Protein crystallography with neutrons – status and perspectives. *Acta Crystallographica* D57, 349–354.
- HABEL, J. E., OHREN, J. F. & BORGSTAHL, G. E. O. (2001). Dynamic light-scattering analysis of full-length human RPA14/32 dimer: purification, crystallization and selfassociation. *Acta Crystallographica* D57, 254–259.
- HAMADA, D., HIGURASHI, T., MAYANGI, K., MIYATA, T., FUKUI, T., IIDA, T., HONDA, T. & YANAGIHARA, I. (2007). Tetrameric structure of the thermostable direct hemolysin from *Vibrio parabaemolyticus* revealed by ultracentrifugation, small-angle X-ray scattering and electron microscopy. *Journal of Molecular Biology* 365, 187–195.
- HAMMEL, M., FIEROBE, H. P., CZJZEK, M., FINET, S. & RECEVEUR-BRECHOT, V. (2004a). Structural insights into the mechanism of formation of cellulosomes probed by small angle X-ray scattering. *Journal of Biological Chemistry* 279, 55985–55994.
- HAMMEL, M., FIEROBE, H. P., CZJZEK, M., KURKAL, V., SMITH, J. C., BAYER, E. A., FINET, S. & RECEVEUR-BRECHOT, V. (2005). Structural basis of cellulosome efficiency explored by small angle X-ray scattering. *Journal* of *Biological Chemistry* 280, 38562–38568.
- HAMMEL, M., KRIECHBAUM, M., GRIES, A., KOSTNER, G. M., LAGGNER, P. & PRASSL, R. (2002). Solution structure of human and bovine β(2)-glycoprotein I revealed by small-angle X-ray scattering. *Journal of Molecular Biology* 321, 85–97.
- HAMMEL, M., SFYROERA, G., RICKLIN, D., MAGOTTI, P., LAMBRIS, J. D. & GEISBRECHT, B. V. (2007). A structural basis for complement inhibition by *Staphylococcus aureus*. *Nature Immunology* 8, 430–437.
- HAMMEL, M., WALTHER, M., PRASSL, R. & KUHN, H. (2004b). Structural flexibility of the N-terminal betabarrel domain of 15-lipoxygenase-1 probed by small angle X-ray scattering. Functional consequences for activity regulation and membrane binding. *Journal of Molecular Biology* 343, 917–929.
- HANSMANN, U. H. E. & OKAMOTO, Y. (1999). New Monte Carlo algorithms for protein folding. *Current Opinions in Structural Biology* 9, 177–183.
- HAO, Q. (2001). Phasing from an envelope. Acta Crystallographica D57, 1410–1414.
- HAO, Q. (2006). Macromolecular envelope determination and envelope-based phasing. *Acta Crystallographica* D62, 909–914.

- HAO, Q., DODD, F. E., GROSSMANN, J. G. & HASNAIN, S. S. (1999). *Ab initio* phasing using molecular envelopes from solution X-ray scattering. *Acta Crystallographica* D55, 243–246.
- HEAD-GORDON, T. & HURA, G. (2002). Water structure from scattering experiments and simulation. *Chemical Reviews* **102**, 2651–2670.
- HEIFETZ, A. & EISENSTEIN, M. (2003). Effect of local shape modifications of molecular surfaces on rigid-body protein-protein docking. *Protein Engineering* 16, 179–85.
- HELLER, W. T. (2005). Influence of multiple well-defined conformations on small-angle scattering of proteins in solution. *Acta Crystallographica* D61, 33–44.
- HELLER, W. T. (2006). ELLSTAT: shape modeling for solution small-angle scattering of proteins and protein complexes with automated statistical characterization. *Journal of Applied Crystallography* **39**, 671–675.
- HELLER, W. T., KRUEGER, J. K. & TREWHELLA, J. (2003). Further insights into calmodulin-myosin light chain kinase interaction from solution scattering and shape restoration. *Biochemistry* **42**, 10579–10588.
- HELLIWELL, J. R. (1997). Overview of synchrotron radiation and macromolecular crystallography. *Methods in Enzymology* 276, 203–217.
- HENDRICKSON, W. A. & OGATA, C. M. (1997). Phase determination from multiwavelength anomalous diffraction measurements. *Methods in Enzymology* 276, 494–523.
- HESS, M. T., GUPTA, R. D. & KOLODNER, R. D. (2002). Dominant Saccharomyces cerevisiae msh6 mutations cause increased mispair binding and decreased disassociation from mispairs by Msh2-Msh6 in the absence of ATP. Journal of Biological Chemistry 277, 25545–25553.
- HINSEN, K. (1998). Analysis of domain motions by approximate normal mode calculations. *Proteins* 33, 417–429.
- HINSEN, K., REUTER, N., NAVAZA, J., STOKES, D. L. & LACAPERE, J. J. (2005). Normal mode-based fitting of atomic structure into electron density maps: application to sarcoplasmic reticulum Ca-ATPase. *Biophysical Journal* 88, 818–827.
- HOPE, H. (1990). Crystallography of biological macromolecules at ultra-low temperature. *Annual Reviews of Biophysics and Biophysical Chemistry* 19, 107–126.
- HOPFNER, K. P., KARCHER, A., CRAIG, L., WOO, T. T., CARNEY, J. P. & TAINER, J. A. (2001). Structural biochemistry and interaction architecture of the DNA double-strand break repair Mre11 nuclease and Rad50-ATPase. *Cell* **105**, 473–485.
- HOPFNER, K. P., KARCHER, A., SHIN, D. S., CRAIG, L., ARTHUR, L. M., CARNEY, J. P. & TAINER, J. A. (2000). Structural biology of Rad50 ATPase: ATP-driven conformational control in DNA double-strand break repair and the ABC-ATPase superfamily. *Cell* **101**, 789–800.
- HOWLIN, B., BUTLER, S. A., MOSS, D. S., HARRIS, G. W. & DRIESSEN, H. P. C. (1993). TLSANL: TLS parameteranalysis program for segmented anisotropic refinement

of macromolecular structures. *Journal of Applied Crystallography* **26**, 622.

- HUBBARD, S. R., HODGSON, K. O. & DONIACH, S. (1988). Small-angle X-ray scattering investigation of the solution structure of troponin C. *Journal of Biological Chemistry* 263, 4151–4158.
- HUNG, L.-W., WANG, I. X., NIKAIDO, K., LIU, P.-Q., AMES, G. F. & KIM, S.-H. (1998). Crystal structure of the ATPbinding subunit of an ABC transporter. *Nature* 396, 703–707.
- JANCARIK, J., PUFAN, R., HONG, C., KIM, S.-H. & KIM, R. (2004). Optimum solubility (OS) screening: an efficient method to optimize buffer conditions for homogeneity and crystallization of proteins. *Acta Crystallographica* D60, 1670–1673.
- JELSCH, C., TEETER, M. M., LAMZIN, V., PICHON-PESME, V., BLESSING, R. H. & LECOMTE, C. (2000). Accurate protein crystallography at ultrahigh resolution: valence electron distribution in crambin. *Proceedings of the National Academy* of Sciences USA 97, 3171–3176.
- JOHS, A., HAMMEL, M., WALDNER, I., MAY, R. P., LAGGNER, P. & PRASSL, R. (2006). Modular structure of solubilized human apolipoprotein B-100. Low resolution model revealed by small angle neutron scattering. *Journal of Biological Chemistry* 281, 19732–19739.
- JONES, P. M. & GEORGE, A. M. (1999). Subunit interactions in ABC transporters: toward a functional architecture. *FEMS Microbiology Letters* 179, 187–202.
- JONES, S. & THORNTON, J. M. (1996). Principles of proteinprotein interactions. *Proceedings of the National Academy of Sciences USA* 93, 13–20.
- KAMATA, K., MITSUYA, M., NISHIMURA, T., EIKI, J. & NAGATA, Y. (2004). Structural basis for allosteric regulation of the monomeric allosteric enzyme human glucokinase. *Structure* 12, 429–438.
- KARPLUS, M. & MCCAMMON, J. A. (2002). Molecular dynamics simulations of biomolecules. *Nature Stuctural Biology* 9, 646–652.
- KARPOWICH, N., MARTSINKEVICH, O., MILLEN, L., YUAN, Y.-R., DIA, P. L., MACVEY, K., THOMAS, P. J. & HUNT, J. F. (2001). Crystal structure of the MJ1267 ATP binding cassette reveal an induced-fit effect at the ATPase active site of an ABC transporter. *Structure* 9, 571–586.
- KE, H. (1997). Overview of isomorphous replacement phasing. *Methods in Enzymology* 276, 448–461.
- KESKIN, O., MA, B. & NUSSINOV, R. (2005). Hot regions in protein-protein interactions: the organization and contribution of structurally conserved hot spot residues. *Journal of Molecular Biology* 345, 1281–1294.
- KIDERA, A. & GO, N. (1990). Refinement of protein dynamic structure: normal mode refinement. *Proceedings of* the National Academy of Sciences USA 87, 3718–3722.
- KLEYWEGT, G. J. & JONES, T. A. (1997). Model building and refinement practice. *Methods in Enzymology* 277, 208–230.

- KNABLEIN, J., NEUEFEIND, T., SCHNEIDER, F., BERGNER, A., MESSERSCHMIDT, A., LOWE, J., STEIPE, B. & HUBER, R. (1997). Ta<sub>6</sub>Br<sub>12</sub><sup>(2+)</sup>, a tool for phase determination of large biological assemblies by X-ray crystallography. *Journal of Molecular Biology* **270**, 1–7.
- Ko, T. P., ROBINSON, H., GAO, Y. G., CHENG, C. H., DEVRIES, A. L. & WANG, A. H. (2003). The refined crystal structure of an eel type III antifreeze protein RD1 at 0.62 Å resolution reveals structural microheterogeneity of protein and solvation. *Biophysical Journal* 84, 1228–1237.
- KOCH, M. H., VACHETTE, P. & SVERGUN, D. I. (2003). Small-angle scattering: a view on the properties, structures and structural changes of biological macromolecules in solution. *Quarterly Reviews of Biophysics* 36, 147–227.
- KOLLMAN, J. M. & QUISPE, J. (2005). The 17 A structure of the 420 kDa lobster clottable protein by single particle reconstruction from cryoelectron micrographs. *Journal of Structural Biology* **151**, 306–314.
- KONAREV, P. V., PETOUKHOV, M. V. & SVERGUN, D. I. (2001). MASSHA – a graphics system for rigid-body modeling of macromolecular complexes against solution scattering data. *Journal of Applied Crystallography* 34, 527–532.
- KONAREV, P. V., VOLKOV, V. V., SOKOLOVA, A. V., KOCH, M. H. J. & SVERGUN, D. I. (2003). PRIMUS: a Windows PC-based system for small-angle scattering data analysis. *Journal of Applied Crystallography* 36, 1277–1282.
- KONG, Y., MING, D., WU, Y., STOOPS, J. K., ZHOU, Z. H. & MA, J. (2003). Conformational flexibility of pyruvate dehydrogenase complexes: a computational analysis by quantized elastic deformational model. *Journal of Molecular Biology* 330, 129–135.
- KONTOPIDIS, G., WU, S. Y., ZHELEVA, D. I., TAYLOR, P., MCINNES, C., LANE, D. P., FISCHER, P. M. & WALKINSHAW, M. D. (2005). Structural and biochemical studies of human proliferating cell nuclear antigen complexes provide a rationale for cyclin association and inhibitor design. *Proceedings of the National Academy of Sciences USA* **102**, 1871–1876.
- KORZHNEV, D. M., BILLETER, M., ARSENIEV, A. S. & OREKHOV, V. Y. (2001). NMR studies of Brownian tumbling and internal motions in proteins. *Progress Nuclear Magnetic Resonance Spectroscopy* 38, 197–266.
- KOSINSKI, J., STEINDORF, L., BUJNICKI, J. M., GIRON-MONZON, L. & FRIEDHOFF, P. (2005). Analysis of the quaternary structure of the MutL C-terminal domain. *Journal of Molecular Biology* 351, 895–909.
- KOZAK, M. (2005). Direct comparison of the crystal and solution structure of glucose/xylose isomerase from *Streptomyces rubiginosus. Protein Peptide Letters* **12**, 547–550.
- KOZIN, M. B. & SVERGUN, D. I. (2001). Automated matching of high- and low-resolution structural models. *Journal of Applied Crystallography* 34, 33–41.

- KOZIN, M. B., VOLKOV, V. V. & SVERGUN, D. I. (1997). ASSA, a program for three-dimensional rendering in solution scattering from biopolymers. *Journal of Applied Crystallography* **30**, 811–815.
- KRATKY, O. & POROD, G. (1949). Roetgenuntersuchung Geloester Fadenmolekuele. *Recueil des Travaux Chimiques* des Pays-Bas 68, 1106–1122.
- KREBS, A., DURCHSCHLAG, H. & ZIPPER, P. (2004). Small angle X-ray scattering studies and modeling of *Eudistylia* vancouverii chlorocruorin and *Macrobdella decora* hemoglobin. *Biophysical Journal* 87, 1173–1185.
- KREBS, W. G., ALEXANDROV, V., WILSON, C. A., ECHOLS, N., YU, H. & GERSTEIN, M. (2002). Normal mode analysis of macromolecular motions in a database framework: developing mode concentration as a useful classifying statistic. *Proteins* 48, 682–695.
- KUHN, L. A., SIANI, M. A., PIQUE, M. E., FISHER, C. L., GETZOFF, E. D. & TAINER, J. A. (1992). The interdependence of protein surface topography and bound water molecules revealed by surface accessibility and fractal density measures. *Journal of Molecular Biology* 228, 13–22.
- KUWAMOTO, S., AKIYAMA, S. & FUJISAWA, T. (2004). Radiation damage to a protein solution, detected by synchrotron X-ray small-angle scattering: dose-related considerations and suppression by cryoprotectants. *Journal of Syncbrotron Radiation* 11, 462–468.
- Kwok, L. W., SHCHERBAKOVA, I., LAMB, J. S., PARK, H. Y., ANDRESEN, K., SMITH, H., BRENOWITZ, M. & POLLACK, L. (2006). Concordant exploration of the kinetics of RNA folding from global and local perspectives. *Journal of Molecular Biology* 355, 282–293.
- LAMERS, M. H., GEORGESCU, R. E., LEE, S. G., O'DONNELL, M. & KURIYAN, J. (2006). Crystal structure of the catalytic alpha subunit of *E. coli* replicative DNA polymerase III. *Cell* **126**, 881–892.
- LAMERS, M. H., GEORGIJEVIC, D., LEBBINK, J. H., WINTERWERP, H. H. K., AGIANIAN, B., DE WIND, N. & SIXMA, T. K. (2004). ATP increases the affinity between MutS ATPase domains: Implications for ATP hydrolysis and conformational changes. *Journal of Biological Chemistry* 279, 43879–43885.
- LAMERS, M. H., PERRAKIS, A., ENZLIN, J. H., WINTERWERP, H. H., DE WIND, N. & SIXMA, T. K. (2000). The crystal structure of the DNA mismatch repair protein MutS binding to a G × T mismatch. *Nature* **407**, 711–717.
- LATTMAN, E. E. (1989). Rapid calculation of the solution scattering profile from a macromolecule of known structure. *Proteins* 5, 149–155.
- LEACH, A. R. (2001). Exploring conformational space using simulation methods. In *Molecular Modelling: Principles and Applications* (ed. P. Hall), pp. 457–508. Harlow: Pearson Education.
- LEE, K. K., GAN, L., TSURUTA, H., HENDRIX, R. W., DUDA, R. L. & JOHNSON, J. E. (2004). Evidence that a local

refolding event triggers maturation of HK97 bacteriophage capsid. *Journal of Molecular Biology* **340**, 419–433.

- LEI, M., ZAVODSZKY, M. I., KUHN, L. A. & THORPE, M. F. (2004). Sampling protein conformations and pathways. *Journal of Computational Chemistry* 25, 1133–1148.
- LEVITT, M., SANDER, C. & STERN, P. S. (1985). Protein normal-mode dynamics: trypsin inhibitor, crambin, ribonuclease, and lysozyme. *Journal of Molecular Biology* 181, 423–447.
- LEVY, Y. & BECKER, O. M. (2002). Conformational polymorphism of wild-type and mutant prion proteins: energy landscape analysis. *Proteins* **47**, 458–468.
- LI, L., CHEN, R. & WENG, Z. (2003). RDOCK: refinement of rigid-body protein docking predictions. *Proteins* 53, 693–707.
- LI, X., KESKIN, O., MA, B., NUSSINOV, R. & LIANG, J. (1985). Protein-protein interactions: hot spots and structurally conserved residues often locate in complemented pockets that preorganize in the unbound states: implications for docking. *Journal of Molecular Biology* 334, 781–795.
- LINDAHL, E. & DELARUE, M. (2005). Refinement of docked protein-ligand and protein-DNA structures using low frequency normal mode amplitude optimiziation. *Nucleic Acids Research* 33, 4496–4506.
- LIPFERT, J., COLUMBUS, L., CHU, V. B. & DONIACH, S. (2007a). Analysis of small-angle X-ray scattering data of protein-detergent complexes by singular value decomposition. *Journal of Applied Crystallography* **40**, 1–5.
- LIPFERT, J., DAS, R., CHU, V. B., KUDARAVALLI, M., BOYD, N., HERSCHLAG, D. & DONIACH, S. (2007b). Structural transitions and thermodynamics of a glycine-dependent riboswitch from *Vibrio cholerae*. *Journal of Molecular Biology* 365, 1393–1406.
- LIPFERT, J. & DONIACH, S. (2007). Small-angle X-ray scattering from RNA, proteins, and protein complexes. *Annual Review of Biophysics and Biomolecular Structure* 36, 307–327.
- LIU, Q., WEAVER, A. J., XIANG, T., THIEL, D. J. & HAO, Q. (2003). Low-resolution molecular replacement using a six-dimensional search. *Acta Crystallographica* D59, 1016–1019.
- LOCHER, K. P., LEE, A. T. & REES, D. C. (2002). The *E. coli* BtuCD structure: a framework for ABC transporter architecture and mechanism. *Science* 296, 1091–1098.
- LOSONCZI, J. A., ANDREC, M., FISCHER, M. W. & PRESTEGARD, J. H. (1999). Order matrix analysis of residual dipolar couplings using singular value decomposition. *Journal of Magnetic Resonance* 138, 334–342.
- LUCIC, V., FORSTER, F. & BAUMEISTER, W. (2005). Structural studies by electron tomography: from cells to molecules. *Annual Reviews in Biochemistry* 74, 833–865.
- LUNDE, C. S., ROUHANI, S., FACCIOTTI, M. T. & GLAESER, R. M. (2006). Membrane-protein stability in a phospholipid-based crystallization medium. *Journal of Structural Biology* **154**, 223–231.

- MA, B., ELKAYAM, T., WOLFSON, H. & NUSSINOV, R. (2003). Protein-protein interactions: structurally conserved residues distinguish between binding sites and exposed protein surfaces. *Proceedings of the National Academy of Sciences USA* **100**, 5772–5777.
- MACKERELL JR., A. D., BASHFORD, D., BELLOTT, M., DUNBRACK JR., R. L., EVANSECK, J. D., FIELD, M. J., FISCHER, S., GAO, J., GUO, H., HA, S., JOSEPH-MCCARTHY, D., KUCHNIR, L., KUCZERA, K., LAU, F. T. K., MATTOS, C., MISCHNICK, S., NGO, T., NGUYGEN, D. T., PRODHOM, B., REIHER III, W. E., ROUX, B., SCHLENKRICH, M., SMITH, J. C., STOTE, R., STAUB, J., WATANABE, M., WIORKIEWICZ-KUCZERA, J., YIN, D. & KARPLUS, M. (1998). All-atom empirical potential for molecular modeling and dynamics studies of proteins. Journal of Physical Chemistry B 102, 3586–3616.
- MAIOROV, V. & ABAGYAN, R. (1997). A new method for modeling large-scale rearrangements of protein domains. *Proteins* 27, 410–424.
- MANDELL, J. G., ROBERTS, V. A., PIQUE, M. E., KOTLOVYI, V., MITCHELL, J. C., NELSON, F., TSIGELNY, I. & TEN EYCK, L. F. (2001). Protein docking using continuum electrostatics and geometric fit. *Protein Engineering* 14, 105–113.
- MARINO, M., ZOU, P., SVERGUN, D., GARCIA, P., EDLICH, C., SIMON, B., WILMANNS, M., MUHLE-GOLL, C. & MAYANS, O. (2006). The Ig doublet Z1Z2: a model system for the hybrid analysis of conformational dynamics in Ig tandems from titin. *Structure* 14, 1437–1447.
- MATTHEWS, B. W. (1968). Solvent content of protein crystals. Journal of Molecular Biology 33, 491–497.
- MATTINEN, M. L., PAAKKONEN, K., IKONEN, T., CRAVEN, J., DRAKENBERG, T., SERIMAA, R., WALTHO, J. & ANNILA, A. (2002). Quaternary structure built from subunits combining NMR and small-angle X-ray scattering data. *Biophysical Journal* 83, 1177–1183.
- McMURRAY, C. T. & TAINER, J. A. (2003). Cancer, cadmium and genome integrity. *Nature Genetics* 34, 239–241.
- MEINHOLD, L. & SMITH, J. C. (2007). Protein dynamics from X-ray crystallography: anisotropic, global motion in diffuse scattering patterns. *Proteins Structure Function* and Bioinformatics 66, 941–953.
- MENDENHALL, M. D. & HODGE, A. F. (1998). Regulation of Cdc28 cyclin-dependent protein kinase activity during the cell cycle of the yeast *Saccharomyces cerevisiae*. *Microbial Molecular Biology Reviews* 62, 1191–1243.
- MENDILLO, M. L., MAZUR, D. J. & KOLODNER, R. D. (2005). Analysis of the interaction between the *Saccharomyces cerevisiae* MSH2-MSH6 and MLH1-PMS1 complexes with DNA using a reversible DNA endblocking system. *Journal of Biological Chemistry* 280, 22245–22257.
- MENDILLO, M. L., PUTNAM, C. D. & KOLODNER, R. D. (2007). Escherichia coli MutS tetramerization domain structure reveals that stable dimers but not tetramers are

essential for DNA mismatch repair *in vivo. Journal of Biological Chemistry* 282, 16345–16354.

- MERZEL, F. & SMITH, J. C. (2002a). Is the first hydration shell of lysozyme of higher density than bulk water? *Proceedings of the National Academy of Sciences USA* 99, 5378–5383.
- MERZEL, F. & SMITH, J. C. (2002b). SASSIM: a method for calculating small-angle X-ray and neutron scattering and the associated molecular envelope from explicit-atom models of solvated proteins. *Acta Crystallographica* D58, 242–249.
- METROPOLIS, N., ROSENBLUTH, A., ROSENBLUTH, M., TELLER, A. & TELLER, E. (1953). Equation of state calculations by fast computing machines. *Journal of Chemical Physics* 21, 1087–1092.
- MIAKE-LYE, R. C., DONIACH, S. & HODGSON, K. O. (1983). Anomalous x-ray scattering from terbium-labeled parvalbumin in solution. *Biophysical Journal* 41, 287–292.
- MITRA, K., SCHAFFITZEL, C., SHAIKH, T., TAMA, F., JENNI, S., BROOKS 3RD, C. L., BAN, N. & FRANK, J. (2005). Structure of the *E. coli* protein-conducting channel bound to a translating ribosome. *Nature* **438**, 318–324.
- MIZUGUCHI, K., KIDERA, A. & GO, N. (1994). Collective motions in proteins investigated by X-ray diffuse scattering. *Proteins* 18, 34–48.
- MONTANGE, R. K. & BATEY, R. T. (2006). Structure of the S-adenosylmethionine riboswitch regulatory mRNA element. *Nature* 441, 1172–1175.
- MOORE, P. B. (1980). Small-angle scattering. Information content and error analysis. *Journal of Applied Crystallography* 13, 168–175.
- MUELLER, J. J., HANSEN, S. & PUERSCHEL, H. V. (1996). The use of small-angle scattering and the maximum-entropy method for shape-model determination from distancedistribution functions. *Journal of Applied Crystallography* 29, 547–554.
- MULLER, K., LAGGNER, P., GLATTER, O. & KOSTNER, G. (1978). The structure of human-plasma low-density lipoprotein B. An X-ray small-angle scattering study. *European Journal of Biochemistry* 82, 73–90.
- MUSTARD, D. & RITCHIE, D. W. (2005). Docking essential dynamics eigenstructures. *Proteins* 60, 269–274.
- MYLONAS, F. & SVERGUN, D. I. (2007). Accuracy of molecular mass determination of proteins in solution by small-angle X-ray scattering. *Journal of Applied Crystallography* 40, s245–s249.
- NAGAR, B., HANTSCHEL, O., SEELIGER, M., DAVIES, J. M., WEIS, W. I., SUPERTI-FURGA, G. & KURIYAN, J. (2006). Organization of the SH3-SH2 unit in active and inactive forms of the c-Abl tyrosine kinase. *Molecular Cell* 21, 787–798.
- NAGAR, B., HANTSCHEL, O., YOUNG, M. A., SCHEFFZEK, K., VEACH, D., BORNMANN, W., CLARKSON, B., SUPERTI-FURGA, G. & KURIYAN, J. (2003). Structural basis of the autoinhibition of c-Abl tyrosine kinase. *Cell* **112**, 859–871.

- NATH, A., ATKINS, W. M. & SLIGAR, S. G. (2007). Applications of phospholipid bilayer nanodiscs in the study of membranes and membrane proteins. *Biochemistry* 46, 2059–2069.
- NAVAZA, J. (2001). Implementation of molecular replacement in AMoRe. Acta Crystallographica D57, 1367–1372.
- NAVAZA, J., LEPAULT, J., REY, F. A., ALVAREZ-RUA, C. & BORGE, J. (2002). On the fitting of model electron densities into EM reconstructions: a reciprocal-space formulation. *Acta Crystallographica* D58, 1820–1825.
- NEMETH-PONGRACZ, V., BARABAS, O., FUXREITER, M., SIMON, I., PICHOVA, I., RUMLOVA, M., ZABRANSKA, H., SVERGUN, D., PETOUKHOV, M., HARMAT, V., KLEMENT, E., HUNYADI-GULYAS, E., MEDZIHRADSZKY, K. F., KONYA, E. & VERTESSY, B. G. (2007). Flexible segments modulate co-folding of dUTPase and nucleocapsid proteins. *Nucleic Acids Research* 35, 495–505.
- NILSSON, L., CLORE, G. M., GRONENBOM, A. M., BRUNGER, A. T. & KARPLUS, M. (1986). Structural refinement of oligonucleotides by molecular dynamics with nuclear Overhauser effect interproton restrains: application to 5'd(C-G-T-A-C-G)<sub>2</sub>. *Journal of Molecular Biology* 188, 455–475.
- NOWAK, E., PANJIKAR, S., KONAREV, P., SVERGUN, D. I. & TUCKER, P. A. (2006a). The structural basis of signal transduction for the response regulator PrrA from *Mycobacterium tuberculosis. Journal of Biological Chemistry* 281, 9659–9666.
- NOWAK, E., PANJIKAR, S., MORTH, J. P., JORDANOVA, R., SVERGUN, D. I. & TUCKER, P. A. (2006b). Structural and functional aspects of the sensor histidine kinase PrrB from *Mycobacterium tuberculosis. Structure* 14, 275–285.
- OBMOLOVA, G., BAN, C., HSIEH, P. & YANG, W. (2000). Crystal structure of mismatch repair protein MutS and its complex with a substrate DNA. *Nature* **407**, 703–710.
- OCKWELL, D. M., HOUGH, M., GROSSMANN, J. G., HASNAIN, S. S. & HAO, Q. (2000). Implementation of cluster analysis for *ab initio* phasing using the molecular envelope from solution X-ray scattering. *Acta Crystallographica* D56, 1002–1006.
- PALMA, P. N., KRIPPAHL, L., WAMPLER, J. E. & MOURA, J. J. (2000). BiGGER: a new (soft) docking algorithm for predicting protein interactions. *Proteins* 39, 372–384.
- PARGE, H. E., ARVAI, A. S., MURTARI, D. J., REED, S. I. & TAINER, J. A. (1993). Human CksHs2 atomic structure: a role for its hexameric assembly in cell cycle control. *Science* 262, 387–395.
- PARSIEGLA, G., REVERBEL-LEROY, C., TARDIF, C., BELAICH, J. P., DRIGUEZ, H. & HASER, R. (2000). Crystal structures of the cellulase Cel48F in complex with inhibitors and substrates give insights into its processive action. *Biochemistry* 39, 11238–11246.
- PASCAL, J. M., O'BRIEN, P. J., TOMKINSON, A. E. & ELLENBERGER, T. (2004). Human DNA ligase I completely encircles and partially unwinds nicked DNA. *Nature* 432, 473–478.

- PASCAL, J. M., TSODIKOV, O. V., HURA, G. L., SONG, W., COTNER, E. A., CLASSEN, S., TOMKINSON, A. E., TAINER, J. A. & ELLENBERGER, T. (2006). A flexible interface between DNA ligase and PCNA supports conformational switching and efficient ligation of DNA. *Molecular Cell* 24, 279–291.
- PENNOCK, E., BUCKLEY, K. & LUNDBLAD, V. (2001). Cdc13 delivers separate complexes to the telomere for end protection and replication. *Cell* **104**, 387–396.
- PEREZ, J., VACHETTE, P., RUSSO, D., DESMADRIL, M. & DURAND, D. (2001). Heat-induced unfolding of neocarzinostatin, a small all-beta protein investigated by small-angle X-ray scattering. *Journal of Molecular Biology* 308, 721–743.
- PERKINS, S. J. (2001). X-ray and neutron scattering analyses of hydration shells: a molecular interpretation based on sequence predictions and modeling fits. *Biophysical Chemistry* 93, 129–139.
- PERRAKIS, A., MORRIS, R. M. & LAMZIN, V. S. (1999). Automated protein model building combined with iterative structure refinement. *Nature Stuctural Biology* 6, 458–463.
- PERRY, J. J., FAN, L. & TAINER, J. A. (2007). Developing master keys to brain pathology, cancer, and aging from the structural biology of proteins controlling reactive oxygen species and DNA repair. *Neuroscience* 145, 1280–1299.
- PERRY, J. J., YANNONE, S. M., HOLDEN, L. G., HITOMI, C., ASAITHAMBY, A., HAN, S., COOPER, P. K., CHEN, D. J. & TAINER, J. A. (2006). WRN exonuclease structure and molecular mechanism imply an editing role in DNA end processing. *Nature Structural and Molecular Biology* 13, 414–422.
- PERUTZ, M. F., BOLTON, W., DIAMOND, R., MUIRHEAD, H. & WATSON, H. C. (1964). Structure of haemoglobin. An X-ray examination of reduced horse haemoglobin. *Nature* 203, 687–690.
- PETOUKHOV, M. V., EADY, N. A., BROWN, K. A. & SVERGUN, D. I. (2002). Addition of missing loops and domains to protein models by X-ray solution scattering. *Biophysical Journal* 83, 3113–3125.
- PETOUKHOV, M. V., MONIE, T. P., ALLAIN, F. H., MATTHEWS, S., CURRY, S. & SVERGUN, D. I. (2006). Conformation of polypyrimidine tract binding protein in solution. *Structure* 14, 1021–1027.
- PETOUKHOV, M. V. & SVERGUN, D. I. (2005). Global rigid body modeling of macromolecular complexes against small-angle scattering data. *Biophysical Journal* 89, 1237–1250.
- PETOUKHOV, M. V. & SVERGUN, D. I. (2006). Joint use of small-angle X-ray and neutron scattering to study biological macromolecules in solution. *European Biophysical Journal* 35, 567–576.
- Petoukhov, M. V., Svergun, D. I., Konarev, P. V., Ravasio, S., van den Heuvel, R. H., Curti, B. & Vanoni, M. A. (2003). Quaternary structure of

Azospirillum brasilense NADPH-dependent glutamate synthase in solution as revealed by synchrotron radiation x-ray scattering. *Journal of Biological Chemistry* **278**, 29933–29939.

- PETROVA, T. & PODJARNY, A. (2004). Protein crystallography at subatomic resolution. *Reports on Progress in Physics* 67, 1565–1605.
- PILZ, I. (1982). Proteins. In Small Angle X-ray Scattering (ed. O. Glatter and O. Kratky), pp. 239–293. London: Academic Press.
- PIOLETTI, M., FINDEISEN, F., HURA, G. L. & MINOR, D. L. J. (2006). Three-dimensional structure of KChIP1-Kv4.3 T1 complex reveals a cross-shaped octamer. *Nature Strucutral and Molecular Biology* 13, 987–995.
- POROD, G. (1951). X-ray small angle scattering of close packed colloidal systems. *Kolloid Zeitschrift* 124, 83–114.
- POROD, G. (1982). General Theory. In Small Angle X-ray Scattering (ed. O. Glatter and O. Kratky), pp. 17–51. London: Academic Press.
- PRESS, W. H., TEUKOLSKY, S. A., VETTERLING, W. T. & FLANNERY, B. P. (1992). Numerical Recipes in C, Second Edition. New York, NY: Cambridge University Press
- PROVENCHER, S. W. & GLOCKNER, J. (1983). Analysis of the components present in kinetics (or titration) curves. *Journal of Biochemical and Biophysical Methods* 7, 331–334.
- PUTNAM, C. D., CLANCY, S. B., TSURUTA, H., GONZALEZ, S., WETMUR, J. G. & TAINER, J. A. (2001). Structure and mechanism of the RuvB Holliday junction branch migration motor. *Journal of Molecular Biology* **311**, 297–310.
- PUTNAM, C. D., SHROYER, M. J., LUNDQUIST, A. J., MOL, C. D., ARVAI, A. S., MOSBAUGH, D. W. & TAINER, J. A. (1999). Protein mimicry of DNA from crystal structures of the uracil-DNA glycosylase inhibitor protein and its complex with *Escherichia coli* uracil-DNA glycosylase. *Journal of Molecular Biology* 287, 331–346.
- PUTNAM, C. D. & TAINER, J. A. (2005). Protein mimicry of DNA and pathway regulation. *DNA Repair* 4, 1410–1420.
- QAZI, O., BOLGIANO, B., CRANE, D., SVERGUN, D. I., KONAREV, P. V., YAO, Z. P., ROBINSON, C. V., BROWN, K. A. & FAIRWEATHER, N. (2007). The HC fragment of tetanus toxin forms stable, concentration-dependent dimers via an intermolecular disulphide bond. *Journal of Molecular Biology* **365**, 123–134.
- RAJAMANI, D., THIEL, S., VAJDA, S. & CAMACHO, C. J. (2004). Anchor residues in protein-protein interactions. *Proceedings of the National Academy of Sciences USA* **101**, 11287–11292.
- RALLISON, J. M. & HARDING, S. E. (1985). Excluded volume for pairs of triaxial ellipsoids at dominant Brownian motion. *Journal of Colloid Interface Science* 103, 284–289.
- RAVELLI, R. B. & McSWEENEY, S. M. (2000). The 'fingerprint' that X-rays can leave on structures. *Structure Fold Design* 8, 315–328.

- REDECKE, L., BERGEN, M., CLOS, J., KONAREV, P. V., SVERGUN, D. I., FITTSCHEN, U. E., BROEKAERT, J. A., BRUNS, O., GEORGIEVA, D., MANDELKOW, E., GENOV, N. & BETZEL, C. (2007). Structural characterization of betasheeted oligomers formed on the pathway of oxidative prion protein aggregation *in vitro. Journal of Structural Biology* **157**, 308–320.
- REJTO, P. A. & FREER, S. T. (1996). Protein conformational substrates from X-ray crystallography. *Progress in Biophysical and Molecular Biology* 66, 167–196.
- RODIER, F., BAHADUR, R. P., CHAKRABARTI, P. & JANIN, J. (2005). Hydration of protein-protein interfaces. *Proteins* 60, 36–45.
- ROSEMAN, A. M. (2000). Docking structures of domains into maps from cryo-electron microscopy using local correlation. *Acta Crystallographica* **D56**, 1332–1340.
- ROSENBERG, O. S., DEINDL, S., SUNG, R. J., NAIRN, A. C. & KURIYAN, J. (2005). Structure of the autoinhibited kinase domain of CaMKII and SAXS analysis of the holoenzyme. *Cell* **123**, 849–860.
- ROSSMANN, M. G. (1995). *Ab initio* phase determination and phase extension using non-crystallographic symmetry. *Current Opinions in Structural Biology* 5, 650–655.
- ROSSMANN, M. G. (2000). Fitting atomic models into electron-microscopy maps. *Acta Crystallographica* D56, 1341–1349.
- ROUILLER, I., DELABARRE, B., MAY, A. P., WEIS, W. I., BRUNGER, A. T., MILLIGAN, R. A. & WILSON-KUBALEK, E. M. (2002). Conformational changes of the multifunctional p97 AAA ATPase during its ATPase cycle. *Nature Structural Biology* 9, 950–957.
- SAKURAI, S., KITANO, K., YAMAGUCHI, H., HAMADA, K., OKADA, K., FUKUDA, K., UCHIDA, M., OHTSUKA, E., MORIOKA, H. & HAKOSHIMA, T. (2005). Structural basis for recruitment of human flap endonuclease 1 to PCNA. *EMBO Journal* 34, 683–693.
- SANISHVILI, R., VOLZ, K. W., WESTBROOK, E. M. & MARGOLIASH, E. (1995). The low ionic strength crystal structure of horse cytochrome c at 2·1 Å resolution and comparison with its high ionic strength counterpart. *Structure* 3, 707–716.
- SCHERES, S. H. G., H. VALLE, M., HERMAN, G. T., EGGERMONT, P. P., FRANK, J. & CARAZO, J. M. (2007). Disentangling conformational states of macromolecules in 3D-EM through likelihood optimization. *Nature Methods* 4, 27–29.
- SCHLICK, T. (2001). Time-trimming tricks for dynamic simulations: splitting force updates to reduce computational work. *Structure* 9, R45–53.
- SCHNECKE, V., SWANSON, C. A., GETZOFF, E. D., TAINER, J. A. & KUHN, L. A. (1998). Screening a peptidyl database for potential ligands to proteins with side-chain flexibility. *Proteins* 33, 74–87.
- SCHOMAKER, V. & TRUEBLOOD, K. (1968). On rigid body motion of molecules in crystals. *Acta Crystallographica* B24, 63–76.

- SEMENYUK, A. V. & SVERGUN, D. I. (1991). GNOM a program package for small-angle scattering data processing. *Journal of Applied Crystallography* 24, 537–540.
- SHANNON, C. E. & MOORE, W. (1949). The Mathematical Theory of Communication. Urbana, IL: University of Illinois Press.
- SHELDRICK, G. M. & SCHNEIDER, T. R. (1997). SHELXL: high-resolution refinement. *Methods in Enzymology* 277, 319–343.
- SHELL, S. S., PUTNAM, C. D. & KOLODNER, R. D. (2007). The N terminus of *Saccharomyces cerevisiae* Msh6 is an unstructured tether to PCNA. *Molecular Cell* 26, 565– 578.
- SHIN, D. S., PELLEGRINI, L., DANIELS, D. S., YELENT, B., CRAIG, L., BATES, D., YU, D. S., SHIVJI, M. K., HITOMI, C., ARVAI, A. S., VOLKMANN, N., TSURUTA, H., BLUNDELL, T. L., VENKITARAMAN, A. R. & TAINER, J. A. (2003). Full-length archaeal Rad51 structure and mutants: mechanisms for RAD51 assembly and control by BRCA2. *EMBO Journal.* 22, 4566–4576.
- SIM, G. A. (1959). The distribution of phase angles for structrues containing heavy atoms. II. A modification of the normal heavy atom method for non-centrosymmetric structures. *Acta Crystallographica* 12, 813–815.
- SMITH, G. R., STERNBERG, M. J. & BATES, P. A. (2005). The relationship between the flexibility of proteins and their conformational states on forming protein-protein complexes with an application to protein-protein docking. *Journal of Molecular Biology* 347, 1077–1101.
- SMITH, K. F., HARRISON, R. A. & PERKINS, S. J. (1990). Structural comparisons of the native and reactivecentre-cleaved forms of alpha 1-antitrypsin by neutronand X-ray scattering in solution. *Biochemical Journal* 267, 203–212.
- SMITH, P. E., VAN SCHAIK, R. C., SZYPERSKI, T., WUTHRICH, K. & VAN GUNSTEREN, W. F. (1995). Internal mobility of the basic pancreatic trypsin inhibitor in solution: a comparison of NMR spin relaxation measurements and molecular dynamics simulations. *Journal of Molecular Biology* 246, 356–365.
- SOKOLOVA, A. V., VOLKOV, V. V. & SVERGUN, D. I. (2003a). Database for rapid protein classification based on small-angle X-ray scattering data. *Crystallography Reports* 48, 1027–1033.
- SOKOLOVA, A. V., VOLKOV, V. V. & SVERGUN, D. I. (2003b). Prototype of a database for rapid protein classification based on solution scattering data. *Journal of Applied Crystallography* **36**, 865–868.
- SONDERMANN, H., NAGAR, B., BAR-SAGI, D. & KURIYAN, J. (2005). Computational docking and solution x-ray scattering predict a membrane-interacting role for the histone domain of the Ras activator son of sevenless. *Proceedings of the National Academy of Sciences USA* **102**, 16632–16637.
- Sousa, M. C., Trame, C. B., Tsuruta, H., Wilbanks, S. M., Reddy, V. S. & McKay, D. B. (2000). Crystal and

solution structures of an HslUV protease-chaperone complex. *Cell* **103**, 633–643.

- STUHRMANN, H. B. (1973). Comparison of the three basic scattering functions of myoglobin in solution with those from the known structure in crystalline state. *Journal of Molecular Biology* 77, 363–369.
- STUHRMANN, H. B. (1981). Anomalous small angle scattering. Quarterly Reviews in Biophysics 14, 433–460.
- STUHRMANN, H. B. (1982). Contrast variation. In Small Angle X-ray Scattering (ed. O. Glatter and O. Kratky), pp. 197–213. London: Academic Press.
- STUHRMANN, H. B., HAAS, J., IBEL, K., WOLF, B., KOCH, M. H., PARFAIT, R. & CRICHTON, R. R. (1976). New low resolution model for 50S subunit of *Escherichia coli* ribosomes. *Proceedings of the National Academy of Sciences* USA 73, 2379–83.
- STUHRMANN, H. B., TARDIEU, A., MATEU, L., SARDET, C., LUZZATI, V., AGGERBECK, L. & SCANU, A. M. (1975). Neutron scattering study of human serum low density lipoprotein. *Proceedings of the National Academy of Sciences* USA 72, 2270–2273.
- SUHRE, K., NAVAZA, J. & SANEJOUAND, Y. H. (2006). NORMA: a tool for flexible fitting of high-resolution protein structures into low-resolution electronmicroscopy-derived density maps. *Acta Crystallographica* D62, 1098–1100.
- SUHRE, K. & SANEJOUAND, Y. H. (2004). ElNemo: a normal mode web server for protein movement analysis and the generation of templates for molecular replacement. *Nucleic Acids Research* 32, W610–614.
- SUN, Z., REID, K. B. & PERKINS, S. J. (2004). The dimeric and trimeric solution structures of the multidomain complement protein properdin by X-ray scattering, analytical ultracentrifugation and constrained modeling. *Journal of Molecular Biology* 343, 1327–1343.
- SVERGUN, D. (1992). Determination of the regularization parameter in indirect-transform methods using perceptual criteria. *Journal of Applied Crystallography* 25, 495–503.
- SVERGUN, D., BARABERATO, C. & KOCH, M. H. (1995). CRYSOL – a program to evaluate X-ray solution scattering of biological macromolecules from atomic coordinates. *Journal of Applied Crystallography* 28, 768–773.
- SVERGUN, D. I. (1999). Restoring low resolution structure of biological macromolecules from solution scattering using simulated annealing. *Biophysical Journal* 76, 2879– 2886.
- SVERGUN, D. I. (2007). Small-angle scattering studies of macromolecular solutions. *Journal of Applied Crystallogra*phy 40, s10–s17.
- SVERGUN, D. I., BURKHARDT, N., PEDERSEN, J. S., KOCH, M. H., VOLKOV, V. V., KOZIN, M. B., MEERWINK, W., STUHRMANN, H. B., DIEDRICH, G. & NIERHAUS, K. H. (1997a). Solution scattering structural analysis of the 70 S *Escherichia coli* ribosome by contrast variation. I. Invariants and validation of electron microscopy models. *Journal of Molecular Biology* 271, 588–601.

- SVERGUN, D. I., BURKHARDT, N., PEDERSEN, J. S., KOCH, M. H., VOLKOV, V. V., KOZIN, M. B., MEERWINK, W., STUHRMANN, H. B., DIEDRICH, G. & NIERHAUS, K. H. (1997b). Solution scattering structural analysis of the 70 S *Escherichia coli* ribosome by contrast variation. II. A model of the ribosome and its RNA at 3-5 nm resolution. *Journal of Molecular Biology* 271, 602–618.
- SVERGUN, D. I., PETOUKHOV, M. V. & KOCH, M. H. (2001a). Determination of domain structure of proteins from X-ray solution scattering. *Biophysical Journal* 80, 2946–2953.
- SVERGUN, D. I., PETOUKHOV, M. V. & KOCH, M. H. (2001b). Determination of domain structure of proteins from X-ray solution scattering. *Biophysical Journal* 76, 2879–2886.
- SVERGUN, D. I., PETOUKHOV, M. V., KOCH, M. H. J. & KOENIG, S. (2000). Crystal versus solution structures of thiamine diphosphate-dependent enzymes. *Journal of Biological Chemistry* 275, 297–302.
- SVERGUN, D. I. & STUHRMANN, H. B. (1991). New developments in direct shape determination from small-angle scattering. 1. Theory and model calculations. *Acta Crystallographica* **A47**, 736–744.
- SVERGUN, D. I., VOLKOV, V. V., KOZIN, M. B. & STUHRMANN, H. B. (1996). New developments in direct shape determination from small-angle scattering. 2. Uniqueness. *Acta Crystallographica* A52, 419–426.
- TAI, K. (2004). Conformational sampling for the impatient. *Biophysical Chemistry* 107, 213–220.
- TAKAHASHI, Y., NISHIKAWA, Y. & FUJISAWA, T. (2003). Evaluation of three algorithms for ab initio determination of three dimensional shapes from one dimensional solvent scattering profiles. *Journal of Applied Crystallography* 36, 546–550.
- TAMA, F., MIYASHITA, O. & BROOKS 3RD, C. L. (2004). Flexible multi-scale fitting of atomic structures into lowresolution electron density maps with elastic network normal mode analysis. *Journal of Molecular Biology* 337, 985–999.
- TAMA, F. & SANEJOUAND, Y. H. (2001). Conformational change of proteins arising from normal mode analysis. *Protein Engineering* 14, 1–6.
- TAMA, F., WRIGGERS, W. & BROOKS 3RD, C. L. (2002). Exploring global distortions of biological macromolecules and assemblies from low-resolution structural information and elastic network theory. *Journal of Molecular Biology* 321, 297–305.
- TAYLOR, A., DUNNE, M., BENNINGTON, S., ANSELL, S., GARDNER, I., NORREYS, P., BROOME, T., FINDLAY, D. & NELMES, R. (2007). A route to the brightest possible neutron source? *Science* 317, 1092–1095.
- TERWILLIGER, T. C. (2003). Automated main-chain model building by template matching and iterative fragment extension. *Acta Crystallographica* D59, 38–44.
- TIDOW, H., MELERO, R., MYLONAS, E., FREUND, S. M. V., GROSSMANN, J. G., CARAZO, J. M., SVERGUN, D. I.,

VALLE, M. & FERSHT, A. R. (2007). Quaternary structures of tumor suppressor p53 and a specific p53-DNA complex. *Proceedings of the National Academy of Sciences* USA **104**, 12324–12329.

- TIEDE, D. M., ZHANG, R. & SEIFERT, S. (2002). Protein conformations explored by difference high-angle solution X-ray scattering: oxidation state and temperature dependent changes in cytochrome C. *Biochemistry* 41, 6605–6614.
- TIEDE, D. M. & ZUO, X. (2007). X-ray scattering for biomolecule structure characterization. In *Biophysical Techniques in Photosynthesis II* (ed. T. J. Aartsma and J. Matysik). Springer.
- TIRION, M. M. (1996). Large amplitude elastic motions in proteins from a single-parameter atomic analysis. *Physical Review Letters* 77, 1905–1908.
- TJIOE, E. & HELLER, W. T. (2007). ORNL\_SAS: software for calculation of small-angle scattering intensities of proteins and protein complexes. *Journal of Applied Crystallography* 40, 782–785.
- TRIKHA, J., THEIL, E. C. & ALLEWELL, N. M. (1995). High resolution structure of amphibian red-cell L-ferritin: potential roles for structural plasticity and solvation in function. *Journal of Molecular Biology* 248, 949–967.
- TSUTAKAWA, S. E., HURA, G. L., FRANKEL, K. A., COOPER, P. K. & TAINER, J. A. (2007). Structural analysis of flexible proteins in solution by small angle X-ray scattering combined with crystallography. *Journal of Structural Biology* **158**, 214–223.
- TUNG, C. S., WALSH, D. A. & TREWHELLA, J. (2002). A structural model of the catalytic subunit-regulatory subunit dimeric complex of the cAMP-dependent protein kinase. *Journal of Biological Chemistry* 277, 12423– 12431.
- URZHUMTSEV, A. G. & PODJARNY, A. (1995). On the solution of the molecular-replacement problem at very low resolution: application to large complexes. *Acta Crystallographica* **D51**, 888–895.
- UZAWA, T., KIMURA, T., ISHIMORI, K., MORISHIMA, I., MATSUI, T., IKEDA-SAITO, M., TAKAHASHI, S., AKIYAMA, S. & FUJISAWA, T. (2006). Time-resolved small-angle X-ray scattering investigation of the folding dynamics of heme oxygenase: implication of the scaling relationship for the submillisecond intermediates of protein folding. *Journal of Molecular Biology* 357, 997–1008.
- VACHETTE, P. & SVERGUN, D. (2000). Small-angle X-ray scattering by solutions of biological macromolecules: neutron and synchrotron radiation for condensed studies. In *Structure and Dynamics of Biomolecules* (ed. E. Fanchon, E. Geissler, J. L. Hodeau, J. E. Regnard and P. A. Timmins), pp. 199–237. Oxford: Oxford University Press.
- VALDAR, W. S. & THORNTON, J. M. (2001a). Conservation helps to identify biologically relevant crystal contacts. *Journal of Molecular Biology* **313**, 399–416.

- VALDAR, W. S. & THORNTON, J. M. (2001b). Proteinprotein interfaces: analysis of amino acid conservation in homodimers. *Proteins* 42, 108–124.
- VAN AALTEN, D. M., CONN, D. A., DE GROOT, B. L., BERENDSEN, H. J., FINDLAY, J. B. & AMADEI, A. (1997). Protein dynamics derived from clusters of crystal structures. *Biophysical Journal* 73, 2891–2896.
- VELLIEUX, F. M. D. & READ, R. J. (1997). Non-crystallographic symmetry averaging in phase refinement and extension. *Methods in Enzymology* 277, 18–53.
- VESTERGAARD, B. & HANSEN, S. (2006). Application of Bayesian analysis to indirect Fourier transformation in small-angle scattering. *Journal of Applied Crystallography* 39, 797–804.
- VESTERGAARD, B., SANVAL, S., ROESSLE, M., MORA, L., BUCKINGHAM, R. H., KASTRUP, J. S., GAJHEDE, M., SVERGUN, D. I. & EHRENBERG, M. (2005). The SAXS solution structure of RF1 differs from its crystal structure and is similar to its ribosome bound cryo-EM structure. *Molecular Cell* **20**, 929–938.
- VOLKMANN, N. & HANEIN, D. (1999). Quantitative fitting of atomic models into observed densities derived by electron microscopy. *Journal of Structural Biology* 125, 176–184.
- VOLKMANN, N. & HANEIN, D. (2003). Docking of atomic models into reconstructions from electron microscopy. *Methods Enzymol* 374, 204–225.
- VOLKOV, V. V. & SVERGUN, D. I. (2003). Uniqueness of ab initio shape determination in small-angle scattering. Journal of Applied Crystallography 36, 860–864.
- VON OSSOWSKI, I., EATON, J. T., CZJZEK, M., PERKINS, S. J., FRANDSEN, T. P., SCHULEIN, M., PANINE, P., HENRISSAT, B. & RECEVEUR-BRECHOT, V. (2005). Protein disorder: conformational distribution of the flexible linker in a chimeric double cellulase. *Biophysical Journal* 88, 2823– 2832.
- VUCETIC, S., BROWN, C. J., DUNKER, A. K. & OBRADOVIC, Z. (2003). Flavors of protein disorder. *Proteins* 52, 573–584.
- WALL, M. E., CLARAGE, J. B. & PHILLIPS, G. N. (1997a). Motions of calmodulin characterized using both Bragg and diffuse X-ray scattering. *Structure* 5, 1599–1612.
- WALL, M. E., EALICK, S. E. & GRUNER, S. M. (1997b). Three-dimensional diffuse X-ray scattering from crystals of Staphylococcal nuclease. *Proceedings of the National Academy of Sciences USA* 94, 6180–6184.
- WALL, M. E., GALLAGHER, S. C. & TREWHELLA, J. (2000). Large-scale shape changes in proteins and macromolecular complexes. *Annual Review of Physical Chemistry* 51, 355–380.
- WALTHER, D., COHEN, F. E. & DONIACH, S. (2000). Reconstruction of low-resolution three-dimensional density maps from one-dimensional small-angle X-ray solution scattering data for biomolecules. *Journal of Applied Crystallography* 33, 350–363.

- WEBER, P. C. (1997). Overview of protein crystallization methods. *Methods in Enzymology* 276, 13–22.
- WEEKS, C. M., BLESSING, R. H., MILLER, R., MUNGEE, R., POTTER, S. A., RAPPLEYE, J., SMITH, G. D., XU, H. & FUREY, W. (2002). Toward automated protein structure determination: BnP, the SnB-PHASES interface. *Zeitschrift fur Kristallographie* **217**, 686–693.
- WILLIAMS, R. S., CHASMAN, D. I., HAU, D. D., HUI, B., LAU, A. Y. & GLOVER, J. N. (2003). Detection of protein folding defects caused by BRCA1-BRCT truncation and missense mutations. *Journal of Biological Chemistry* 278, 53007–53016.
- WILLIS, B. T. M. & PRYOR, A. W. (1975). Thermal Vibrations in Crystallography. London: Cambridge University Press.
- WILSON, A. J. C. (1942). Determination of absolute from relative X-ray intensity data. *Nature* 150, 152.
- WILSON, M. A. & BRUNGER, A. T. (2000). The 1·0 Å crystal structure of Ca<sup>2+</sup>-bound calmodulin: an analysis of disorder and implications for functionally relevant plasticity. *Journal of Molecular Biology* **301**, 1237–1256.
- WINN, M. D., ISUPOV, M. N. & MURSHUDOV, G. N. (2001). Use of TLS parameters to model anisotropic displacements in macromolecular refinement. *Acta Crystallographica* D57, 122–133.
- WINN, M. D., MURSHUDOV, G. N. & PAPIZ, M. Z. (2003). Macromolecular TLS refinement in REFMAC at moderate resolutions. *Methods in Enzymology* 374, 300–321.
- WRIGGERS, W. & CHACON, P. (2001). Modeling tricks and fitting techniques for multiresolution structures. *Structure* 9, 779–788.
- WRIGGERS, W., MILLIGAN, R. A. & MCCAMMON, J. A. (1999). Situs: a package for docking crystal structures into low-resolution maps from electron microscopy. *Journal of Structural Biology* **125**, 185–195.
- WU, L. & HICKSON, I. D. (2006). DNA helicases required for homologous recombination and repair of damaged replication forks. *Annual Review of Genetics* **40**, 279–306.
- YAMAGATA, A. & TAINER, J. A. (2007). Hexameric structures of the archaeal secretion ATPase GspE and implications for a universal secretion mechanism. *EMBO Journal* 26, 878–890.
- YEATS, T. O. (1997). Detecting and overcoming crystal twinning. *Methods in Enzymology* 276, 344–358.
- YUAN, Y.-R., BLECKER, S., MARTSINKEVICH, O., MILLEN, Y., THOMAS, P. J. & HUNT, J. F. (2001). The crystal structure

of the MJ0796 ATP-binding cassette: implications for the structural consequences of ATP hydrolysis in the active site of an ABC transporter. *Journal of Biological Chemistry* **276**, 32313–32321.

- YUZAWA, S., YOKOCHI, M., HATANAKA, H., OGURA, K., KATAOKA, M., MIURA, K., MANDIYAN, V., SCHLESSINGER, J. & INAGAKI, F. (2001). Solution structure of Grb2 reveals extensive flexibility necessary for target recognition. *Journal of Molecular Biology* **306**, 527–537.
- ZEGERMAN, P. & DIFFLEY, J. F. X. (2007). Phosphorylation of Sld2 and Sld3 by cyclin-dependent kinases promotes DNA replication in budding yeast. *Nature* 445, 281–285.
- ZHANG, K. Y. J., COWTAN, K. & MAIN, P. (1997). Combining constraints for electron density modification. *Methods in Enzymology* 277, 53–64.
- ZHANG, R., THIYAGARAJAN, P. & TIEDE, D. M. (2000a). Probing protein fine structures by wide angle solution X-ray scattering. *Journal of Applied Crystallography* 33, 565–568.
- ZHANG, X., SHAW, A., BATES, P. A., NEWMAN, R. H., BOWEN, B., ORLOVA, E., GORMAN, M. A., KONDO, H., DOKURNO, P. & LALLY, J. (2000b). Structure of the AAA ATPase p97. *Molecular Cell* 6, 1473–1484.
- ZHAO, J., HOYE, E., BOYLAN, S., WAISH, D. A. & TREWHELLA, J. (1998). Quaternary structure of a catalytic subunit-regulatory subunit dimeric complex and the holoenzyme of the cAMP-dependent protein kinase by neutron contrast variation. *Journal of Biological Chemistry* 273, 30448–30459.
- ZHENG, Y., DOERSCHUK, P. C. & JOHNSON, J. E. (1995). Determination of three-dimensional low-resolution viral structure from solution X-ray scattering. *Biophysical Journal* 69, 619–639.
- ZIPPER, P. & DURSCHSCHLAG, H. (2003). Modeling of protein solution structures. *Journal of Applied Crystallography* 36, 509–514.
- ZUKER, M., MATHEWS, D. H. & TURNER, D. H. (1999). Algorithms and Thermodynamics for RNA Secondary Structure Prediction: A Practical Guide in RNA Biochemistry and Biotechnology. Dordrecht: Kluwer Academic Publishers.
- ZULAUF, M. & D'ARCY, A. (1992). Light scattering of proteins as a criterion for crystallization. *Journal of Crystal Growth* **122**, 102–106.